

Friends or strangers? Attempts at reactivating buyer-supplier relationships

Downloaded from: https://research.chalmers.se, 2025-06-18 03:45 UTC

Citation for the original published paper (version of record):

Poblete, L., Bengtson, A. (2021). Friends or strangers? Attempts at reactivating buyer–supplier relationships. Journal of Business and Industrial Marketing, 36(2): 177-190. http://dx.doi.org/10.1108/JBIM-12-2019-0513

N.B. When citing this work, cite the original published paper.

research.chalmers.se offers the possibility of retrieving research publications produced at Chalmers University of Technology. It covers all kind of research output: articles, dissertations, conference papers, reports etc. since 2004. research.chalmers.se is administrated and maintained by Chalmers Library

TOPICAL REVIEW • OPEN ACCESS

Screening nature of the van der Waals density functional method: a review and analysis of the many-body physics foundation

To cite this article: Per Hyldgaard et al 2020 J. Phys.: Condens. Matter 32 393001

View the article online for updates and enhancements.

Recent citations

- Efficient yet accurate dispersion-corrected semilocal exchange-correlation functionals for non-covalent interactions Abhilash Patra *et al*
- Next-Generation Nonlocal van der Waals Density Functional
 D. Chakraborty *et al*
- <u>Rational Design of 2D h-BAs Monolayer as</u> <u>Advanced Sulfur Host for High Energy</u> <u>Density Li–S Batteries</u> Nabil Khossossi *et al*



IOP ebooks[™]

Bringing together innovative digital publishing with leading authors from the global scientific community.

Start exploring the collection-download the first chapter of every title for free.

IOP Publishing

J. Phys.: Condens. Matter **32** (2020) 393001 (52pp)

Topical Review

Screening nature of the van der Waals density functional method: a review and analysis of the many-body physics foundation

Per Hyldgaard¹, Yang Jiao¹ and Vivekanand Shukla¹

Department of Microtechnology and Nanoscience (MC2), Chalmers University of Technology, SE-41296 Gothenburg, Sweden

E-mail: hyldgaar@chalmers.se

Received 20 January 2020 Accepted for publication 23 March 2020 Published 23 June 2020



Abstract

We review the screening nature and many-body physics foundation of the van der Waals density functional (vdW-DF) method [Berland K et al 2015 Rep. Prog. Phys. 78 066501], a systematic approach to construct truly nonlocal exchange-correlation energy density functionals. To that end we define and focus on a class of consistent vdW-DF versions that adhere to the Lindhard screening logic of the full method formulation. The consistent-exchange vdW-DF-cx version [Berland K and Hyldgaard P 2014 Phys. Rev. B 89 035412] and its spin extension [Thonhauser T et al 2015 Phys. Rev. Lett. 115 136402] represent the first examples of this class; in general, consistent vdW-DFs reflect a concerted expansion of a formal recast of the adiabatic-connection formula [Hyldgaard P et al 2014 Phys. Rev. B 90 075148], an exponential summation of contributions to the local-field response, and the Dyson equation. We argue that the screening emphasis is essential because the exchange-correlation energy reflects an effective electrodynamics set by a long-range interaction. Two consequences are that (1) there are, in principle, no wiggle room in how one balances exchange and correlation, for example, in vdW-DF-cx, and that (2) consistent vdW-DFs have a formal structure that allows them to incorporate vertex-correction effects, at least in the case of levels that experience recoil-less interactions (for example, near the Fermi surface). We explore the extent to which the strictly nonempirical vdW-DF-cx formulation can serve as a systematic extension of the constraint-based semilocal functionals. For validation, we provide a complete survey of vdW-DF-cx performance for broad molecular processes, for the full set of 55 benchmarks in GMTKN55 [Goerigk L et al 2017 Phys. Chem. Chem. Phys. 19 32184] and comparing to the quantum-chemistry calculations that are summarized in that paper. We also provide new vdW-DF-cx results for metal surface energies and work functions that we compare to experiment. Finally, we use the screening insight to separate the vdW-DF nonlocal-correlation term into pure-vdW-interaction and local-field-susceptibility effects and present tools to compute and map the binding signatures.

¹ Author to whom any correspondence should be addressed.

Original content from this work may be used under the terms of the Creative Commons Attribution 4.0 licence.

Any further distribution of this work must maintain attribution to the author(s) and the title of the work, journal citation and DOI.

Keywords: density functional theory, van der Waals interactions, many-particle physics, Lindhard screening

(Some figures may appear in colour only in the online journal)

1. Introduction

The van der Waals (vdW) density functional (vdW-DF) method [1-16] is a systematic approach to construct robust, truly nonlocal approximations for the exchange-correlation (XC) energy functional $E_{xc}[n]$ in density functional theory (DFT). It provides a computationally efficient account [17, 18] of dispersion forces while keeping a seamless integration with the local (spin) density approximation [4, 10, 19-24](LDA) in the homogeneous electron-gas (HEG) limit. The aim is to extend the success of the constraint-based semilocal generalized-gradient approximations [25-27] (GGAs), such as PBE [28] and PBEsol [29], relying exclusively on the nonlocal variation in the electron density $n(\mathbf{r})$. Traditional (GGAbased) DFT works well for systems characterized by dense electron concentrations, e.g., individual (small) molecules and hard (bulk) materials [30, 31]. However, more is needed to accurately describe systems that are sparse [12, 32], that is, have important low-density regions across which vdW forces and other truly nonlocal-correlation effects contribute significantly to the cohesion, structure, and function. The vast classes of molecular crystals and complexes, layered materials, polymers and biochemical systems are important examples of such sparse matter.

The versions and variants [3–5, 8, 9, 11, 16, 40–44] of the vdW-DF method enable strictly parameter-free, nonempirical DFT calculations for general (sparse or dense) matter [12, 30, 32, 45].

The vdW-DF method is based on an exact reformulation of the adiabatic connection formula (ACF) [22, 25, 46], as discussed in references [10, 12, 47] and as further detailed within. We enforce current conservation on the electron-gas response description [10]. The ACF and current-conservation foundation has a number of advantages. The vdW-DF versions and variants can, for example, be subjected to a formal couplingconstant analysis [14, 15, 48–51] to isolate kinetic-energy, kinetic-correlation-energy, and electron–electron interaction binding contributions. This analysis provides a refined mapping [14, 52] of where, for example, the vdW interactions between molecules and at surfaces originate in space [10, 53–59]. The coupling-constant scaling analysis can also be used to discuss the choice of Fock-exchange mixing in vdW-DF-based hybrids [13, 15].

At the heart of all constraint- and ACF-based XC functional designs lie an analysis of the electron-gas response to an external potential, $\delta \Phi_{\text{ext}}^{\omega}$, oscillating at a specific frequency ω , and causing density changes $\delta n_{\lambda}^{\omega}$. The response function $\chi_{\lambda}(\omega) = \delta n_{\lambda}^{\omega} / \delta \Phi_{\text{ext}}^{\omega}$ is investigated as a function of an assumed reduction, $V_{\lambda} \equiv \lambda V$, of the electron–electron interaction V. The ACF then provides a formal determination of the XC energy as an average over the coupling constant λ ,

$$E_{\rm xc} = -\int_0^1 d\lambda \int_0^\infty \frac{du}{2\pi} \operatorname{Tr}\{V \chi_\lambda(\mathrm{i}u)\} - E_{\rm self}.$$
 (1)

The trace 'Tr' is often taken in a spatial representation of the response function, $\langle \mathbf{r} | \chi_{\lambda}(\omega) | \mathbf{r}' \rangle \equiv \chi_{\lambda}(\mathbf{r}, \mathbf{r}'; \omega)$, using the electron-electron interaction matrix element $V(\mathbf{r} - \mathbf{r}') = |\mathbf{r} - \mathbf{r}'|^{-1}$. For an illustration of how the ACF and how individual vdW-DF contributions appear when all spatial arguments are written out, we refer the readers to reference [60]. Here we stick with a compact notation, noting that traces are invariant under basis-set changes and that our discussion of the role of Lindhard screening [61–63] would otherwise become overly cumbersome. The last term of equation (1) is the electron selfenergy, given by $E_{\text{self}} = \text{Tr}\{\hat{n}\hat{V}\}/2$, where $\hat{n}(\mathbf{r})$ denotes the operator for the electron density at \mathbf{r} .

The vdW-DF method rests on the observation that the vdW interactions are an inherent part of the XC energy functional [1, 64, 65] and should be treated on the same footing as all other electron-electron interaction energy effects. Like the subsequent and related VV10 [66], rVV10 [67], and rVV10-corrected XC formulations [68-72], the vdW-DF versions and variants describe all forces directly on the foundation of the full electron-density variation [9, 10, 14, 45, 47]. We do not rely on an atom-centered vdW account, in contrast to the Grimme dispersion-correction schemes [33, 73–75], the Becke–Johnsson exchange-hole scheme [76–78], the Wannier-based vdW formulation [79], the Tkatchenko-Scheffler formulation [80], and the associated self-consistent screening extension [81]. We also avoid the need for an auxiliary determination of the atomic dipolar (or multipolar) susceptibilities and the choice of a damping function or a Coulomb-range-separation parameter that must otherwise be used to avoid double counting. The vdW-DF method simply helps us design explicit, truly nonlocal XC density functionals for standard, yet vdW-inclusive, ground-state DFT calculations.

For practical approximations, i.e., vdW-DF versions, we start with a nonlocal, double-pole model for the plasmon propagator, termed S_{xc} , references [4, 9, 12, 47]. This propagator adapts the HEG plasmon description [19, 20, 82–86], used in early LDA [21, 22, 83], to the logic of the gradient expansion [4, 47, 87, 88] of formal many-body perturbation theory (MBPT). The propagator reflects the local-energy variation of an internal (sometimes called inner) GGA-type semilocal functional [4, 8–10, 12, 89]. It therefore effectively reflects the local-field response $\tilde{\chi}(\omega) = \chi(\omega)/[1 + V\chi(\omega)]$ and screening in a weakly perturbed electron gas.



Figure 1. Comparison of rev-PBE-D3 (the DFT-D3 correction [33] of revPBE [34], light blue), of vdW-DF [4] (the original general-geometry version of the vdW-DF method, orange), and of vdW-DF-cx [9] (the 2014 consistent-exchange method update, red) performance for all of the (intra-plus inter-molecular) noncovalent-interaction benchmark sets defined in the GMTKN55 molecular-testing suite [35]. The bars contrast mean-absolute energy deviation (MAD) and root-mean-square deviation (RMSD) values in kcal mol⁻¹, although capped for the IDISP (actual vdW-DF MAD = 7.61) and WATER27 (actual vdW-DF MAD = 9.53) benchmark sets. We used the spin extension of the vdW-DF method [11], where relevant, and the QUANTUM ESPRESSO plane-wave code package [36], with a set of previously tested [13, 15, 37] norm-conserving pseudopotentials from the ABINIT package [38]. The Makov–Payne correction scheme [39] were used for charged-system calculations, affecting results for benchmark sets marked by an asterisk.

For explicit vdW-DF designs, we typically proceed to extract a nonlocal-correlation energy contribution [4, 8, 9, 12, 45],

$$E_{\rm c}^{\rm nl} \approx \int \frac{\mathrm{d}u}{4\pi} \, \mathrm{Tr} \{ S_{\rm xc}^2 - (\nabla S_{\rm xc} \cdot \nabla V/4\pi)^2 \}, \qquad (2)$$

by expanding to second order in terms of S_{xc} . The derivation of this truncated expression for E_c^{nl} has been documented several places, for example, in references [4, 7, 12, 17, 60]; it will also emerge naturally in our discussion of the full vdW-DF method and of consistent vdW-DF versions, below. For layered geometries we can also retain the full nonlocalcorrelation description subject to a slightly different approximation for the local-field response [2, 3, 6].

The vdW-DF approximations for the full XC energy, that is, vdW-DF versions and variants,

$$E_{\rm xc}^{\rm vdW-DF} = E_{\rm xc}^{\rm in} + \delta E_{\rm x}^{\rm 0} + E_{\rm c}^{\rm nl},\tag{3}$$

contain a cross-over exchange term [9] δE_x^0 , reflecting the choice of gradient-corrected exchange. The vdW-DF formulation equation (3) captures local and general nonlocal correlation effects in E_{xc}^{in} and E_c^{nl} , respectively. The exchange effects are captured in the semilocal functional $E_{xc}^0 \equiv E_{xc}^{in} + \delta E_x^0$.

This review paper highlights the nature and discusses potential advantages of focusing the vdW-DF framework on the class of consistent designs, i.e., designs that seek to implement the full screening logic of the ACF [9, 10, 45]. The key idea, developed within, is that approximations to exchange are independent of the coupling constant [14, 48, 90] and must completely reside in the approximations to $\tilde{\chi}(\omega)$. This implies that we can use the Dyson equation, reflecting Lindhard screening [61–63], to balance exchange and correlations in the vdW-DF designs. In practice, for consistent vdW-DF designs, there should be no binding-energy consequences of retaining a cross-over exchange component δE_x^0 in equation (3). This is effectively true in the consistent-exchange vdW-DF-cx version [9], and in the consistent spin-extension, svdW-DF-cx [11].

Figure 1 demonstrates that there are advantages in terms of accuracy and robustness in moving from the original vdW-DF version [4] and to consistent (spin) vdW-DF-cx. The figure compares the mean absolute deviation (MAD) and root-mean square deviation (RMSD) values obtained for the set of all (inter- and intra-) molecular noncovalent-interaction benchmark sets that are part of (and defined in) the GMTKN55 benchmark suite [35]. The computational details will be given in section 5. The performance on individual interaction problems can vary, but the vdW-DF-cx performance is significantly better than that of vdW-DF. The vdW-DF-cx has just a GGAlevel exchange and should, in the GMTKN55 survey [35] be compared with (one of) the best-performing dispersioncorrected GGA, namely revPBE-D3 [33, 34] (also shown). Again, the vdW-DF-cx is seen to fare better, statistically speaking. In addition, vdW-DF-cx is seen to perform as well as revPBE-D3 when we broaden the comparison to the full GMTKN55 benchmark suite in section 6.

This revPBE-D3/vdW-DF/vdW-DF-cx performance comparison highlights the importance of balancing exchange and correlation in XC functionals [9, 12, 22, 25, 45]. The first two functionals differ only in the choice of nonlocal correlation, whereas the last two differ only in the exchange, and hence in the exchange–correlation balance. The regions of smooth density variations (regions with low-to-medium values of the scaled density gradient, $s = |\nabla n|/(2\pi^2)^{1/3}/(2n^{4/3})$ generally dominate in the description of molecular interactions [9, 10]. We have documented that this also holds for noncovalent interactions [9, 10, 14, 53, 56]. This implies that the Langreth–Vosko (LV) MBPT analysis [91] underpins the dominant part of the vdW-DF-cx exchange description [9]—vet LV exchange has not previously impressed anybody for molecules [92]. Meanwhile, the revPBE exchange choice [34] largely follows the hole-based design logic [93] of PBE [28] but avoids one constraint that strengthens PBE and PBEsol relevance for bulk matter that has a dense electron distribution [29].

The revPBE exchange [34] is a good choice for molecules [35], but that is clearly not enough: figure 1 shows that vdW-DF performs significantly worse than revPBE-D3. Mean-while, picking a good nonlocal correlation term (that helps vdW-DF-cx perform slightly better than revPBE-D3 for non-covalent interactions, figure 1) is also not enough, for the same reason. Rather, it is the foundation in the Lindhard screening logic [10, 12, 61–63] that puts vdW-DF-cx ahead of both revPBE-D3 and vdW-DF, even for molecular problems.

So far the vdW-DF-cx version has proven itself useful and accurate for descriptions of a range of dense and sparse-matter problems. Successful applications include the study of traditional bulk-matter cohesion, structure, thermo-physical properties [9, 45, 94, 95], and vacancy formation [96], as well as of binding, structure, vibrations, polarization, elastic properties, phase transitions, defects, and dynamics in molecular, polymer, and layered materials [45, 97-112]. This is, for example, relevant for development of ionic liquids, batteries, thermo-electrics, and organic solar-cell materials [98, 113–118]. The vdW-DF-cx version tends to overestimate the binding energy in some layered systems [12, 119, 120], but remains accurate on structure [68, 121]. It works for both intraand for inter-molecular bonds [9, 11, 13–15, 44, 122–124], for hybrid organic-inorganic perovskites [125], and for analysis and design of porous-materials gas storage [11, 45, 126-128] and of surface pacification [129]. It also works for studies of the role of polycyclic aromatic hydrocarbons in interstellar H₂ formation [130], of electric-field driven reactions [131], and of mechanochemical cocrystal formation [132]. Use of vdW-DF-cx is also helping the understanding of the crossover from physisorption to weak chemisorption in molecular adhesion [11, 133–145], of catalysis [146–148], of hydrogen tunneling [149], of magnetic coupling between adsorbates [150], and of charge transfer and electron transport in and through molecular adsorbates [151-153]. It is therefore worth exploring why (spin) vdW-DF-cx works well.

However, the use of screening to balance exchange and correlation (as in vdW-DF-cx) may find use also in future nonlocal XC functional designs. The purpose of this review paper is therefore more than presenting a re-derivation of vdW-DF-cx itself. We focus on developing the concepts, and we explore and map the nature of binding contributions in both traditional bulk and in weaker vdW-interaction problems. The paper supplements reference [10] that began the work to interpret the vdW-DF method.

The paper has 8 major objectives. We emphasize that there exist (1) a formally exact vdW-DF framework that is closely tied to the screening nature of an effective electrodynamics, (2) powerful guiding principles for making consistent approximations, and (3) a way to simultaneously account for and balance vdW forces [1, 12, 64, 65, 154–158] and the screening [61–63, 159] & vertex-correction effects [86, 160–164] that all affect

the electron-gas response and zero-point energy (ZPE) dynamics in concert. Our approach is to (4) discuss what we need for crafting a robust nonlocal-correlation functional and (5) show that those needs can be accommodated, in an Occam approach, leading to so-called consistent vdW-DF versions, like vdW-DF-cx [9]. We find that current conservation [10] and compliance with the Lindhard–Dyson logic of screening [61–63] plays a critical role in balancing exchange and correlation effects in such consistent vdW-DF designs. For consistent vdW-DFs, the current conservation [10] implies an automatic charge conservation for the vdW-DF XC hole, i.e., the essential DFT criterion [10, 45] that the quasi-particles remain neutral [22, 25]. There has only been few prior discussions of the vdW-DF method foundation in a screening approximation [6, 7, 10, 45, 53, 165].

We furthermore (6) identify systems and problems that test the Occam solution strategy behind vdW-DF-cx and (7) summarize the results of such validation checks, whether previously reported or provided here. As such the review and analysis paper also contains new results in the form of (a) broad molecular benchmarking, (b) tests of performance for metal surface energies and work functions, and (c) mappings of the spatial variation in binding in bulk Si, Na, and W as well as in aC₆₀ dimer and a graphene bilayer. In addition, the paper contains (d) demonstrations of the non-additivity of binding among carbyne wires, carbon nanotubes, and C₆₀ fullerenes.

There has only been comparatively few prior numerical explorations of the spatial distribution of the vdW-DF binding contributions [10, 14, 53–59]; here we are therefore adding (8) a practical approach to separately track the variation in binding contributions arising from pure vdW interactions and from nonlocal vertex corrections (and related screening effects defined by an implicit cumulant expansion [12, 86]). This is done both for the carbyne-wire, the graphene-bilayer, and the Si, Na, & W bulk problems.

We note that to systematically extend the GGA success, we must deliver accuracy in concurrent descriptions of both sparse and dense matter [9, 32, 42, 45]. We hope to simultaneously succeed in predicting diverse properties with just one robust, transferable, and strictly-parameter-free functional. Consistent vdW-DF versions, like vdW-DF-cx [9] (together with its spin extension svdW-DF-cx [11],) aspire to work as such a general-purpose tool for all types of materials and their combination [45].

The paper is organized as follows. The next section 2 discusses the nature of nonlocal-correlation effects, including vdW interaction effects, but also noting the central role of vertex corrections. Section 3 provides a formal presentation of the design logic of constraint-based XC functionals, contrasting a direct path (leading to semilocal GGAs) and the indirect approach (leading to the vdW-DF method).

Section 4 discusses the design logic of the recent consistentexchange vdW-DF-cx version. Section 4 also interprets the nature of XC functional contributions in such consistent vdW-DF versions, noting that vdW-DF-cx can be sorted into a localfield and a Dyson-correction component; the former reflects vertex corrections (plus related screening effects [86, 163, 164]) and the latter reflects pure vdW interactions. Section 4 furthermore highlights the importance of key vdW-DF-cx features: (a) seamless integration [4, 166], (b) an underlying MBPT analysis of response in the electron gas [4, 7, 11, 91], (c) current conservation [10], and, in particular, (d) compliance with the Dyson/Lindhard screening logic [61–63].

Section 5 contains a summary of computational details while section 6 contains results and discussions that validate trust in the class of consistent vdW-DF designs. The section also contains an exploration of the role that nonlocal vertex corrections and related screening effects [85, 86, 163, 164] play in material binding.

Finally, section 7 contains a summary and conclusions and there are four appendices detailing: (A) the relation of an exponential re-summation and screening, (B) the vdW-DF double-pole plasmon-propagator model, the electrodynamicsresponse nature of terms arising in a full (non-truncated) implementation of the vdW-DF method, (C) a discussion of how to include higher-order expansion terms in E_c^{nl} , and (D) universal-kernel evaluations of both a pure-vdW contribution and of a cumulant contribution that describes nonlocal vertex-type screening effects in consistent vdW-DF versions.

2. Nature of nonlocal correlations

Figure 2 summarizes the fundamental idea of DFT and the need for developing truly nonlocal, vdW-inclusive functionals. The figure simultaneously highlights that there is something extremely right with the charge-conservation criterion that underpins the design of traditional semilocal XC functionals (i.e., GGAs) and that there are important conceptual reasons for correcting and extending the GGA logic [10, 64, 65, 91, 158].

2.1. Ubiquitous vdW interactions challenge DFT

The top left panel of figure 2 shows schematics of a physical systems in which electrons experience full electron–electron interactions *V*; every electron interacts with each and every other electron, as illustrated by arrows. This interacting problem defines the behavior of molecules and materials but, except in small systems, there is no realistic approach to obtain exact solutions under general conditions. The top right panel shows that DFT nevertheless succeeds with an, in principle, correct yet computationally tractable determination of the ground state density $n(\mathbf{r})$ and energy E_{tot} . It does so by considering an equivalent system of noninteracting quasi-particles (shown as hexagons) moving in some effective single-particle potential (represented by the change in background color). The effective potential is a functional derivative of the XC energy functional E_{xc} .

The top right panel of figure 2 also summarizes the core problem for practical use of DFT: deciding exactly how we should place the nut (electron) in its nutshell (a surrounding so-called XC hole). Together, the electron–XC-hole pairs form neutral composites. However, such quasi-particles have an internal structure with charged components (the electron and the associated GGA-type XC hole [10]) and thus an inherent electrodynamics response.



Figure 2. Nature of the DFT formulation (top right panel), of the electron-electron interaction problem (top left panel) and the DFT need for systematic inclusion of vdW interactions, for example, among disjunct material fragments (bottom panel). The top panels show schematics of the DFT mapping of the interacting system onto a quasi-particle problem with independent-particle dynamics defined by an effective potential (green background). Each electron is wrapped in a screening cloud forming neutral electron-XC-hole composites (shown as pentagons), which in local (LDA) or semilocal (GGA) DFT are fairly compact and cannot reflect the density beyond a low-density region between, for example, molecules [6, 32]. The bottom panel illustrates the inherent limitations of LDA/GGA. The panel shows a schematic of a vdW interaction problem: two density fragments, each with electrons (exemplified by blue dots) and associated XC holes (green shells), are separated by a characteristic length R (reflecting a typical intermolecular separation). We can ignore the electrostatic and other types of binding across the intermediate density void (represented by a vertical dashed line). However, the neutral electron-XC-hole composites do have an internal zero-point energy (ZPE) dynamics. They act as antennas, and cause mutual electrodynamics couplings that produce a vdW force between the fragments [1, 3, 4, 10, 12, 64, 65, 91, 158].

The XC hole describes the tendency of an electron at point **r** to inhibit presence of other electrons at point **r'**. The combination of each electron and of its associated XC hole forms neutral complexes that serve as noninteracting quasi-particles in practical DFT calculations. Screening and exchange effects strongly influence the electron-gas behavior and will wrap an electron (at position **r**) in a matching XC hole $n_{xc}(\mathbf{r}; \mathbf{u} = \mathbf{r'} - \mathbf{r})$. An important guideline is that this XC hole must be of exactly opposite charge,

$$\int \mathrm{d}\mathbf{u} \, n_{\mathrm{xc}}(\mathbf{r}; \mathbf{u}) = -1. \tag{4}$$

The details in how we approximate this hole are important: the XC energy is given by the Coulomb coupling inside the composite electron–XC-hole quasi-particles

$$E_{\rm xc} \equiv \frac{1}{2} \int_{\mathbf{r}} \int_{\mathbf{r}'} \frac{n(\mathbf{r}) n_{\rm xc}(\mathbf{r}; \mathbf{r}' - \mathbf{r})}{|\mathbf{r} - \mathbf{r}'|}.$$
 (5)

While equation (5) may appear as simple electrostatics, it also reflects the ZPE dynamics of the electrons relative to their

associated XC holes [10, 64, 65]. The collective excitations, i.e., plasmons identified by some general eigenvalue index η and frequency ω_{η} , typically dominate in the specification of the electron gas response. Accordingly, the XC energy approximations must contain a leading term reflecting the plasmon ZPE [10, 25, 158]:

$$E_{\rm xc} \approx E_{\rm pl}^{\rm ZPE} = \sum_{\eta} \frac{\omega_{\eta}}{2}.$$
 (6)

Since ω_{η} depends on the electron-density variation, the leading-term approximation, equation (6), allows an elegant representation of gradient effects on the XC energy equation (5) at surfaces [25]. Of course, most formulations of the LDA [19–23, 84] and the design of the popular constraint-based GGAs, such as PBE [28] and PBEsol [29], proceed by modeling fairly compact XC holes [93] $n_{xc}(\mathbf{r}; \mathbf{u})$; the details are set subject to the charge-conservation criterion equation (4). However, the plasmon ZPE picture, equation (6), and the XC hole picture, equation (5), can be reconciled by noting that we can link the energy density of a semi-local XC functional to an assumed plasmon dispersion, references [2–4, 7, 12, 25].

The bottom panel of figure 2 illustrates the conceptual problems for semilocal density functional approximations. The problems (for the design of vdW-inclusive functionals) are that the XC hole is typically seen as compact, having a modified Gaussian shape in LDA and GGA [19, 20, 84, 93, 167], and as static, without a ZPE dynamics [1, 10, 65, 77, 78]. The first assumption directly affects our description of the electrodynamics, as the XC hole also defines an approximate dielectric function, equation (20) below. Maggs, Rapcewicz, and Ashcroft addressed the second assumption, pointing out that a static view of the XC hole is incomplete. LDA and GGAs hide and sometimes ignore the electrodynamics coupling between the electron-XC hole systems [10, 64, 65]. The argument was originally cast in terms of diagrams [64, 65, 91], that is, insight from MBPT that goes beyond the input for specifying the details of the GGA-type descriptions [26, 64, 91, 168]. However, the argument can be summarized by evaluating changes in $E_{\rm pl}^{\rm ZPE}$ induced by the electrodynamics coupling among electron–XC-hole complexes [10, 65].

The bottom panel of figure 2 shows a simple onedimensional model with two disjunct density fragments (assumed to reside on either side of the dashed line). It focuses the discussion on the virtual dynamics of electrons (negative balls on either side) and highlights the central GGA problem: the electron–XC-hole pairs are seen as compact and, for example, entirely confined inside a given electron-density fragment.

For a simple illustration of GGA limitations, we assume that the GGA-type descriptions (for either fragment) can be summarized by just one characteristic plasmon frequency $\omega_{pl,0}$. The frequency $\omega_{pl,0}$ describes the electron–XC-hole dynamics given by relative displacement coordinates $x_{i=1,2}$; for simplicity, we also here assume that the characteristic plasmon dynamics can be represented as a quantum harmonic oscillator with an effective spring-constants k (illustrated in the lower panel) and some effective mass m_{eff} , chosen such that $\omega_{\text{pl},0} = \sqrt{k/m_{\text{eff}}}$. Adapting reference [25], we can represent the GGA-type XC energy for the combined but disjunct double-fragment system using

$$H_{\rm dis} = \sum_{i=1,2} \frac{m_{\rm eff}}{2} \left(\frac{{\rm d}x_i}{{\rm d}t}\right)^2 + \frac{1}{2}kx_i^2,$$
 (7)

as we also assume no density overlap of the fragments. However, the model, equation (7), is fundamentally flawed.

The point is that each GGA-type quasi-particle, i.e., the electron–XC-hole complex, is a rattler that is formed by charged components (although it is neutral overall). As such, they are antennas transmitting and receiving due to the ZPE dynamics [10, 64, 65, 158]. We cannot ignore the near-field electrodynamics coupling among GGA-type XC-hole/electron systems, even when considering disjunct fragments [10, 65]. Specifically, at electron displacements x_1 and x_2 and at mean (GGA-type quasi-particle) separation R_{12} , there will be a near-field Hamiltonian contribution

$$H_{\text{near-field}} = -\frac{2x_1 x_2}{R_{12}^3}.$$
 (8)

The systems with mutual electrodynamics coupling, equations (7) and (8), can be solved by a simple canonical transformation yielding new coupled-harmonic-oscillator frequencies [56]:

$$\omega_{\pm} = \sqrt{\omega_{\rm pl,\,0}^2 \pm \frac{2}{m_{\rm eff}R_{12}^3}}.$$
(9)

In this simple model, the net ZPE energy gain by the electrodynamics coupling is

$$\Delta E_{\rm xc} \approx \Delta E_{\rm pl}^{\rm ZPE} \sim \delta E_{\rm vdW},\tag{10}$$

$$\delta E_{\rm vdW} = \frac{1}{2} (\omega_+ + \omega_- - 2\omega_{\rm pl,0}) \propto -R_{12}^{-6}, \qquad (11)$$

and thus represents the asymptotic vdW attraction. The electrodynamics-coupling argument [1, 10, 56, 64, 65, 158], and insight from the underlying MBPT characterizations, cannot easily be cast directly as an explicit refinement of a semilocal XC hole, nor in a semilocal functional [91].

2.2. Inclusion of vdW forces in formal DFT

For an introduction on how the vdW-DF method includes such nonlocal-correlation effects, we need formal definitions of the electron-gas problem and for the MBPT solution approach. At any given coupling constant λ , we separate the Hamiltonian $H_{\lambda} = H_0 + V_{\lambda}$ into a single-particle and the electron-electron interaction part. We use $\hat{\psi}(\mathbf{r})$ to denote the operator for removing an electron at \mathbf{r} . The operators for the local density and for the kinetic energy are $\hat{n}(\mathbf{r}) = \hat{\psi}^+(\mathbf{r})\hat{\psi}(\mathbf{r})$ and $\hat{T} = -\frac{1}{2}\int_{\mathbf{r}}\hat{\psi}^+(\mathbf{r}) \nabla_{\mathbf{r}}^2\hat{\psi}(\mathbf{r})$, respectively. The noninteracting Hamiltonian part H_0 also contains a (single-electron) potential V_{ext} representing, for example, the electron-ion interacting. For deriving the ACF and the results behind our couplingconstant analysis, it is convenient to let H_0 include an extra single-particle potential term that keeps the solution density independent of λ [22, 25, 46]. The temporal arguments of operators, like $\delta \hat{n}(\mathbf{r}, t) = \hat{n}(\mathbf{r}, t) - \langle \hat{n}(\mathbf{r}) \rangle$, reflect the timepropagation under H_{λ} .

Also, we define time-ordered Green functions and density-density correlation functions

$$g_{\lambda}(\mathbf{r},\mathbf{r}';t-t') = -i\langle T\{\hat{\psi}(\mathbf{r},t),\hat{\psi}^{+}(\mathbf{r}',t')\}\rangle_{\lambda},\qquad(12)$$

$$\chi_{\lambda}(\mathbf{r},\mathbf{r}';t-t') = -i\langle T\{\delta\hat{n}(\mathbf{r},t),\delta\hat{n}(\mathbf{r}',t')\}\rangle_{\lambda}.$$
 (13)

Here 'T' denotes the operator of time ordering [86, 167, 169] and the subscripts ' λ ' identify the coupling constant for which we evaluate ground-state expectation values. The quasiparticle dynamics equation (12) is evaluated for a given spin and we assume that g be can treated as diagonal in the spin indices, for simplicity of our discussion. As noted in the introduction, we use a compressed operator-type notation for their temporal Fourier transforms, for example, using $g_{\lambda}(\omega)$ as a shorthand that represents the full coordinate variation

$$g_{\lambda}(\mathbf{r}, \mathbf{r}'; \omega) = \langle \mathbf{r} | g_{\lambda}(\omega) | \mathbf{r}' \rangle.$$
(14)

The Green function $g_0(\omega)$, describing the dynamics in the absence of the electron–electron interaction V_{λ} , is available from an explicit construction from the solutions of H_0 . The quasi-particle dynamics can then, in principle, be determined by a Dyson equation

$$g_{\lambda}(\omega) = g_{\lambda}^{0}(\omega) + g_{\lambda}^{0}(\omega)\sigma(\omega)g_{\lambda}(\omega), \qquad (15)$$

where $\sigma(\omega)$ denotes the self-energy representing the effects of the electron–election interaction V_{λ} on the quasi-particle dynamics [167, 169].

An external-potential fluctuation, $\delta \Phi_{\text{ext}}(\mathbf{r}', t) = \delta \Phi_{\text{ext}}^{\omega}(\mathbf{r}') e^{-i\omega t}$ will, at coupling constant λ , produce a density response

$$\delta n_{\lambda}^{\omega}(\mathbf{r}) = \int_{\mathbf{r}'} \langle \mathbf{r} | \chi_{\lambda}(\omega) | \mathbf{r}' \rangle \delta \Phi_{\text{ext}}^{\omega}(\mathbf{r}'), \qquad (16)$$

where $\omega > 0$ is assumed, without loss of generality.

ľ

The vdW-DF method succeeds in tracking the ZPEcoupling effects of the electron–XC-hole systems by relying on a new (not semi-local) framework for the design of XC energy functionals [2–4, 9, 10, 12, 45, 47]. Formally, the XC energy is given by the electron–electron correlation function $\chi(\omega)$ through the ACF, equation (1) or by the XC-hole formulation, equation (5), using the formal specification [10, 170]

$$h_{\rm xc}(\mathbf{r};\mathbf{r}'-\mathbf{r}) = -\delta(\mathbf{r}-\mathbf{r}') -\frac{2}{n(\mathbf{r})} \int_0^\infty \frac{\mathrm{d}u}{2\pi} \int_0^1 \mathrm{d}\lambda \,\langle \mathbf{r} |\chi_\lambda(\mathrm{i}u)|\mathbf{r}'\rangle.$$
(17)

Both the XC hole and the XC energy functional is therefore critically dependent on a correct inclusion of the ZPE dynamics of the collective excitations of the electron gas, i.e., plasmons, defined as poles of $\chi_{\lambda}(iu)$ or as nodes of the corresponding Lindhard dielectric function $\kappa_{\lambda}(iu)$ = $(1 + V_{\lambda}\chi_{\lambda})^{-1}$. The tracking of energy shifts by the coupling of the electron and plasmons ZPE dynamics is, for example, the defining step in the early LDA specifications [21, 22], based on a Green function calculation of electron quasi-particle energy shifts [83].

Effectively, in the vdW-DF method we employ an exact recast of the ACF and hence of the true XC energy

$$E_{\rm xc} = \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(\kappa_{\rm ACF}(iu))\} - E_{\rm self}.$$
 (18)

The recast is defined by an effective longitudinal dielectric function [10, 47]:

$$\kappa_{\rm ACF}(\omega) \equiv \exp\left(-\int_0^1 d\lambda \, V\chi_\lambda(\omega)\right).$$
(19)

The derivation fits in this sentence, inserting equations (19) into (18) leads immediately back to equation (1). Equations (18) and (19) provide a formulation of the exact XC energy as an electrodynamics problem. This is convenient for discussing dispersive interactions acting in concert with all other interaction effects in the electron gas.

We note in passing that the utility focus is the right way to think of the MBPT progress that we have made in the Chalmers–Rutgers vdW-DF team [1–4, 6–12, 45, 47, 56, 171]. In the vdW-DF method we seek to obtain implicit designs of nonlocal XC functionals, starting from a GGA-level description to model the plasmon dispersion [3, 4, 6, 8, 9, 11]. The vdW-DF team first saw equation (18) as an approximative mean-value evaluation of the coupling-constant integration in the ACF [2–4, 8]. We soon realized that discussing the derivation is moot: an explicit derivation from the traditional ACF, equation (1), would only be relevant if we actually intended to start from some explicit approximations for $\chi_{\lambda}(\omega)$. We do not.

The vdW-DF method is crafted in the electron-gas tradition [45]. The complexity of a direct MBPT evaluation has always prompted XC developers to instead look at the formal structure of diagrams and MBPT as a guide [9, 10, 21, 22, 25, 26, 46, 83, 89, 91, 172–174]. The vdW-DF method does not magically vanish the challenges of brute-force MBPT. It does, however, turn the table and make it easier to leverage insight from electrodynamics [1–3, 10, 47, 53, 59, 64, 65, 158, 165, 171, 175–182], conservation laws [4, 10, 45, 83, 183], and the theory of screening [6, 10, 61–63, 157–159, 179, 184–187].

We are free to take equations (18) and (19) as the vdW-DF framework [10] and focus on finding good physically motivated approximations for $\kappa_{ACF}(iu)$. Our framework is, after all, an exact rewrite of the formally exact ACF [10, 47]. The success of the overall vdW-DF program [1–4, 6, 8–10, 12, 32] shows that this is indeed possible. Emphasizing a current-conservation criterion [10] and the Lindhard screening logic [62, 63] leads to better accuracy in the consistent spin vdW-DF-cx version [9, 11, 45]. This is exciting, for it points to the soundness of the vdW-DF design strategy. We do not think it is a total coincidence.

In fact, taking equations (18) and (19) as the starting point for approximations is a direct reflection of some well-known MBPT successes. First, the dielectric function formulation equation (19) is inspired by a cluster expansion of electron-gas screening [86, 163]. Second, using equation (18) to describe internal-energy contributions, equation (18), permits a succinct MBPT discussion of the nature and magnitude of vdW and Casimir forces [158]. Both of these points are detailed further in this review.

The ACF recast as an effective electrodynamics, equations (18) and (19), also reflects an explicit construction of an associated effective (λ -averaged) density-density correlation function $\chi_{ACF}(\omega)$, developed and discussed in references [10, 12, 45, 47]. We first observe that working with an effective ACF electrodynamics function to represent the XC energy effects in DFT has a direct physical meaning to most DFT practitioners: the form of $\kappa_{ACF}(\omega)$ reflects the density correlations in the exact XC hole,

$$n_{\rm xc}(\mathbf{r};\mathbf{r}'-\mathbf{r}) = -\delta(\mathbf{r}-\mathbf{r}') -\frac{1}{2\pi n(\mathbf{r})} \int_0^\infty \frac{\mathrm{d}u}{2\pi} \nabla_{\mathbf{r}}^2 \langle \mathbf{r} | \ln(\kappa_{\rm ACF}(iu)) | \mathbf{r}' \rangle.$$
(20)

To see this, we write out the off-diagonal matrix elements of the ACF dielectric function

$$\langle \mathbf{r} | \ln(\kappa_{\text{ACF}}(iu)) | \mathbf{r}' \rangle = -\int_0^1 d\lambda \int_{\mathbf{r}''} V(\mathbf{r} - \mathbf{r}'') \langle \mathbf{r}'' | \chi_{\lambda}(iu) | \mathbf{r}' \rangle,$$
(21)

and use the Gauss-condition on the Coulomb interaction matrix element, $\nabla_{\mathbf{r}}^2 V(\mathbf{r} - \mathbf{r}') = -4\pi \delta(\mathbf{r} - \mathbf{r}')$. We find

$$\int_{0}^{1} d\lambda \langle \mathbf{r} | \chi_{\lambda}(iu) | \mathbf{r}' \rangle = \frac{1}{4\pi} \nabla_{\mathbf{r}}^{2} \langle \mathbf{r} | \ln(\kappa_{\text{ACF}}(iu)) | \mathbf{r}' \rangle, \quad (22)$$

which, in turn, proves equation (20).

We also observe that the traditional ACF contains a λ averaging of weighted $\chi_{\lambda}(\omega)$ contributions while we seek a representation of the effective response $\chi_{ACF}(\omega)$ to leverage the electrodynamics insight that underpins vdW interactions [1, 64, 65, 158]. The effective response $\chi_{ACF}(\omega)$ is given in terms of a set of λ -averaged poles of $\chi_{\lambda}(\omega)$, i.e., plasmons of the effective ACF electrodynamics. We are not aware of any explicit formulation of the coupling-constant variation of the collective excitations themselves (nor could we be sure that such an approach is tractable). In the vdW-DF method we overcome this challenge by instead setting the effective response form implicitly [10],

$$-V \chi_{\rm ACF}(\omega) \equiv 1 - \exp\left(-\int_0^1 d\lambda \, V \, \chi_{\lambda}(\omega)\right), \qquad (23)$$

so that it reflects a Lindhard-type relation [10, 62]

$$\kappa_{\rm ACF}(\omega) = (1 + V\chi_{\rm ACF}(\omega))^{-1}.$$
 (24)

This is motivated because there is physical meaning in the effective ACF electrodynamics and deviating from the screening logic is an uncontrolled approximation [63].

In fact, since the ACF recast, given by equations (20) and (18), should be seen as a proper electrodynamics problem, it must reflect an effective long-range interaction, denoted

 $V_{\rm ACF} \sim V_{\lambda_{\rm eff}}$. Working with an effective long-range interaction always makes it essential to pursue a consistent handling of screening effects [10, 61–63]. As indicated, the ACF electrodynamics ($V_{\lambda_{\rm eff}}$ and $\kappa_{\rm ACF}(\omega)$) is defined by an effective coupling constant value $0 < \lambda_{\rm eff} < 1$, as discussed in section 3. For a given external potential change $\delta \Phi_{\rm ext}^{\omega}$ (assumed to oscillate at frequency ω), the ACF recast leaves no wiggle room in how we balance the external field change $\delta E_{\rm ext}^{\omega}(\mathbf{r}) = -\nabla \delta \Phi_{\rm ext}^{\omega}$ and the resulting induced polarization field $\delta \mathbf{P}^{\omega}(\mathbf{r})$ (unless we make an uncontrolled approximation in the Dyson equation for the electron-gas response). The ACF dielectric function $\kappa_{\rm ACF}$ also allows us to define a local potential $\delta \Phi_{\rm loc}^{\omega} \equiv \kappa_{\rm ACF}^{-1}(\omega) \delta \Phi_{\rm ext}^{\omega}$ and an associated local electric field $\delta \mathbf{E}_{\rm loc}^{\omega}(\mathbf{r}) = -\nabla \delta \Phi_{\rm loc}^{\omega}$. Mutually consistent, local-field and external-field susceptibilities must be defined

$$\tilde{\boldsymbol{\alpha}}(\mathbf{r},\mathbf{r}';\omega) = \delta \mathbf{P}^{\omega}(\mathbf{r}) / \delta \mathbf{E}_{\text{loc}}^{\omega}(\mathbf{r}'), \qquad (25)$$

$$\boldsymbol{\alpha}_{\text{ext}}(\mathbf{r}, \mathbf{r}'; \omega) = \delta \mathbf{P}^{\omega}(\mathbf{r}) / \delta \mathbf{E}_{\text{ext}}^{\omega}(\mathbf{r}').$$
(26)

The local-field susceptibility $\tilde{\alpha}(\omega)$ corresponds to an effective polarization insertion [169] $\tilde{\chi}_{ACF}(\omega)$ that must also adhere to the Lindhard screening logic [10]:

$$\kappa_{\rm ACF}(\omega) = 1 - V \tilde{\chi}_{\rm ACF}(\omega). \tag{27}$$

Working with the effective susceptibilities, equations (25) and (26), also provides an intuitive picture. This picture can be used to simplify our discussion of nonlocal-correlation effects in the vdW-DF method and thus help guide approximations to equation (18).

We can compute (within an approximation to the ACF electrodynamics description) both the resulting density change δn^{ω} and the polarization field δP^{ω} produced by the induced current. However, current-conservation and the continuity equation stipulates that these results must be related [10, 47],

$$\delta n^{\omega} = \chi_{\rm ACF}(\omega) \delta \Phi^{\omega}_{\rm ext} = \nabla \cdot \boldsymbol{\alpha}_{\rm ext}(\omega) \cdot \nabla \delta \Phi^{\omega}_{\rm ext} \qquad (28)$$

$$= \tilde{\chi}_{\rm ACF}(\omega) \delta \Phi^{\omega}_{\rm loc}.$$
 (29)

The second line follows because equations (24) and (27) imply an effective Dyson-type equation [10], which can also be expressed

$$\chi_{\rm ACF} V = \tilde{\chi}_{\rm ACF} V + \tilde{\chi}_{\rm ACF} V \tilde{\chi}_{\rm ACF} V + \cdots .$$
(30)

Our approximation for the dielectric function must therefore contain explicit longitudinal projections [10]:

$$\chi_{\rm ACF}(\omega)V = \nabla \cdot \boldsymbol{\alpha}_{\rm ext}(\omega) \cdot \nabla V, \qquad (31)$$

$$\tilde{\chi}_{\rm ACF}(\omega)V = \nabla \cdot \tilde{\alpha}(\omega) \cdot \nabla V. \tag{32}$$

The longitudinal projection is, in turn, essential for making robust approximations for the associated dielectric function, $\kappa_{ACF}(\omega)$. Assume that $\epsilon(\omega) = 1 + 4\pi\tilde{\alpha}(\omega)$ denotes an approximation for a dielectric tensor function that aims to reflect the effective electrodynamics. To ensure current conservation [10], we must use the longitudinal projection

$$\kappa_{\text{long}}(\omega) \equiv \nabla \cdot \boldsymbol{\epsilon}(\omega) \cdot \nabla G, \tag{33}$$

where $G = -V/4\pi$ denotes the Coulomb Green function.

The need for a longitudinal projection has direct implications for the vdW-DF method, as is also further discussed in appendix **B**. For consistent vdW-DF approximations, like vdW-DF-cx [9], we focus on making ACF dielectric-function approximations that have this explicit longitudinally projection. That is, we focus on formulations where we effectively use

$$\kappa_{\rm ACF}(\omega) \to \kappa_{\rm long}(\omega) = \nabla \cdot \epsilon(\omega) \cdot \nabla G,$$
 (34)

inside formal vdW-DF functional specification, equation (18).

We observe that using the projected form equation (33) is consistent with the vdW-DF framework, equation (18). To show this we note that the exact XC functional specification, equation (18), can equally well be written

$$E_{\rm xc} = \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(1 - \tilde{\chi}_{\rm ACF}(iu)V)\} - E_{\rm self}.$$
 (35)

This follows simply by writing out a Taylor expansion of the logarithm and using the invariance of the trace over spatial coordinates. Given equation (35), it is clear that we are always in a position to explicitly enforce the longitudinal projection and to ensure current conservation via equation (33). The projection is systematically used in the evaluation of the nonlocal-correlation term, for example, in its truncated form equation (2), reference [12].

Mahan used the formal 'ln(κ)' structure of equation (18) to compute the interaction of disjunct fragments by an electrodynamics coupling [158]. Mahan's analysis was cast in model susceptibilities. In the vdW-DF method, equation (1), we inherit the Mahan 'ln(κ)' analysis [158], leading us not only to the above-summarized electrodynamics interpretation of the ACF, but also directly to a formal XC-energy evaluation [10, 158]:

$$\int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(\kappa_{\mathrm{ACF}}(\mathrm{i}u))\} = E_{\mathrm{pl}}^{\mathrm{ZPE}} + E_{\tilde{\alpha}}^{\mathrm{res}}.$$
 (36)

The terms are given by the residuals in the complex-frequency contour integral [22, 25, 63, 158], that is implied in the ACF, equation (1). The leading component is just the ZPE sum, equation (6), of plasmons, defined as the zeros of $\kappa_{ACF}(\omega)$, reference [10]. The term $E_{\tilde{\alpha}}$ is the sum of residuals produced at the poles of $\tilde{\alpha}(\omega)$. As such, this term is set by the poles of $\kappa_{ACF}(\omega)$, references [10, 158].

The vdW-DF framework and method are naturally set up to capture the Ashcroft ZPE-coupling mechanism for the vdW interactions because of the electrodynamics recasting. This follows because the ACF contour integral [22, 25] reflects the collective excitations, and the nature of such plasmons depends on the system-specific electrodynamics coupling among the set of electron–XC-hole composites [10, 158]. There is no need to make an assumption of a harmonic behavior for plasmons (as we did for illustration purposes above).

For example, for two disjunct fragments, the binding energy contribution from the XC energy binding reduces to the expected asymptotic form of the vdW interaction [10],

$$\Delta E_{\rm xc} \to \Delta E_{\rm vdW}^{\mathcal{AB}} = -\int \frac{\mathrm{d}u}{4\pi} \mathrm{Tr}_{\mathcal{A}} \{ \boldsymbol{\alpha}_{\rm ext}^{\mathcal{A}} \cdot T_{\mathcal{AB}}^{\rm dip} \cdot \boldsymbol{\alpha}_{\rm ext}^{\mathcal{B}} \cdot T_{\mathcal{AB}}^{\rm dip} \},$$
(37)

where $T^{\text{dip}}(\mathbf{r}, \mathbf{r}') \equiv -\nabla_{\mathbf{r}} \nabla_{\mathbf{r}'} V(\mathbf{r} - \mathbf{r}')$. In equation (37), we have used subscripts on the dipole–dipole coupling, T_{AB}^{dip} , to emphasize that we consider components connecting points in the spatially disjunct regions. The result equation (37) follows simply by expanding the logarithm in the formal recast, equation (18) to second order in $\chi_{ACF}(iu)$ using equation (34) and the nature of an assumed electrodynamics approximation, $\epsilon(\omega) = 1 + 4\pi\tilde{\alpha}(\omega)$.

The result, equation (37), corresponds exactly to the starting point of the Zaremba–Kohn description of physisorption [184, 185] and holds as long as we can ignore interactioninduced changes in the density, reference [10]. At the same time, equation (18) and vdW-DF are set up to track such vdW attractions (and other nonlocal-correlation effects) also when they occur within materials fragments or between fragments with a finite density overlap [2–4, 10, 12, 45].

2.3. Inclusion of GGA-type screening effects

In the following sections, supported by appendix A, we explain how the vdW-DF method adapts the second susceptibility term $E_{\bar{\alpha}}^{\text{res}}$ of equation (36) to secure a consistent description both of vdW interactions and of general screening effects, including vertex corrections.

We recall that the electron-gas is a many-particle system and that the bare Coulomb form, $|\mathbf{r}_1 - \mathbf{r}_2|^{-1}$, of the interaction matrix element can only directly describe the situation with exactly two electrons at play. Vertex corrections quantify the extent that an actual interaction event (for example, defining the quasi-particle dynamics) differs from the bare Coulomb interaction [167, 169]. In MBPT, we handle a large part of all the interaction consequences by collecting these raw-Coulomb modifications into an effective form, described by an interaction vertex Γ , while keeping track of double counting. The approach is exact when consistently implemented [89, 168, 169, 188, 189]. As we shall summarize in section 3, the extent of vertex corrections also directly controls the formal differences between constraint-based GGAs and LDA, when starting from an RPA-like model analysis [25, 91, 174, 190]. The top panel of figure 3 shows a Dyson expansion for the electron Green function where the fourth term is a simple example of a contribution containing an interaction vertex correction.

The handling of vertex corrections in the vdW-DF method works both in the HEG limit and in the presence of density gradients. For example, the recent (s)vdW-DF-cx version [9, 11] competes favorably with constraint-based GGAs, even in materials with a dense electron distribution. A high accuracy for (s)vdW-DF-cx (and for the related variants, vdW-DF-C09 and vdW-DF-optB86b [41, 42]), is documented both for oxide ferroelectrics [45, 191], for the structural and thermo-physical properties of metals [42, 94, 95], and for many other types of problems, as mentioned in the introduction.

The term E_{α}^{res} is formally the sum of residuals at the poles of $\kappa_{\text{ACF}}(\omega)$, but that observation does not reflect how we put



Figure 3. Feynman-diagram solutions for the quasi-particle dynamics of a level having recoil-less interactions with a surrounding itinerant electron gas, described by the ZPE dynamics of plasmons ω_{η} (shown as wiggly lines). The level can be a core level or an extended state near the Fermi surface [85, 86, 163, 164]. The (core-)level Green function \mathcal{G}^0 is shown by arrow lines. Panel (a) illustrates a traditional Dyson expansion for the interacting or screened Green function \mathcal{G} , with the fourth term illustrating the nature of vertex corrections. Panel (b) illustrates the elegance and nature of the linked-cluster expansion or cumulant solution [86, 159–162] for $\mathcal{G}(t - t')$; in the time domain it is sufficient to consider a single connected diagram as the exponential re-summation provides an automatic inclusion of, for example, all vertex contributions (in the absence of recoil [86, 163, 164]). Panel (c) illustrates that the $\mathcal{G}^0 W$ approximation for the electron self energy underpins the corresponding frequency-domain formulation [164] of this cumulant expansion, equation (67).

the material susceptibility to work in the vdW-DF method. Within a single-particle picture, such as the Hartree–Fock (HF) approximation, Mahan derived the simple form [158],

$$E_{\tilde{\alpha}}^{\text{res,HF}} = -\sum_{\xi} \frac{\omega_{\xi}^{\text{sp}}}{2},$$
(38)

where ω_{ξ}^{sp} denotes the single-particle excitation energies. Mahan's characterization of vdW interactions [158] provides an XC-energy determination, via equation (36), that is consistent with the random phase approximation (RPA) for spatially inhomogeneous systems [10]. The RPA is a special limit of the vdW-DF method, reference [10], but the vdW-DF method aims to go beyond RPA in keeping vertex corrections.¹

In the vdW-DF method, we use a bootstrap approach to nucleate the nonlocal-correlation functionals around GGA ideas [4, 8, 10]. This is done instead of the RPA reliance of a lowest-order approximation for the local-field response function. DFT wisdom suggests keeping local exchange and correlation together [25] as done, e.g., in the screened-exchange description for local correlations in LDA [21, 22]. Vertex corrections are of central relevance for the success of LDA [19, 20, 25, 84]. However, vertex corrections are also essential for an accurate description of screening and electron correlations in cases with a pronounced density gradient. In a discussion of the quasi-particle dynamics, they can be succinctly described in cases with recoil-less interactions [85, 86, 163, 164], and we seek to port that insight.

Specifically, we rely on an effective model component, $E_{\text{xc},\alpha} \neq E_{\tilde{\alpha}}^{\text{res},\text{HF}}$, that reflects the non-vdW parts of a GGAlike XC functional. It does that in terms of a simple, scalar model susceptibility form $\alpha(\omega)$. We note that a local-field susceptibility, like $\tilde{\alpha}$ or α , naturally contains nonlocalcorrelation effects also when they arise within a density fragment. Such fragments can be molecules or traditional bulk matter, cases where a GGA can be expected to work. In practice, we nucleate this vdW-DF start around a so-called internal functional E_{xc}^{in} that retains all of the vertex correction effects entering the LDA specification. The vdW-DF method then tries to capture additional gradient-corrected vertex corrections through an exponential re-summation [86, 163, 164] in $E_{xc,\alpha}$.

Collective excitations are expected to contribute significantly also in the GGA-type description $E_{xc,\alpha}$. A proper GGA definition already exhausts most of the plasmon term E_{pl}^{ZPE} in the implicit XC functional specification, equation (36). We can assume that most of the screening effects, including vertex corrections [85, 86, 163, 164], are already included in the GGA. It is primarily the pure vdW interaction term, denoted $E_{c, vdW}^{nl}$ (and given by the logic described in subsections 2.1 and 2.2) that is missing from the GGA account when the GGA is used to describe sparse-matter systems. The formal structure of equation (36) is highly useful since the contour integral tracks all screening effects in the electron gas [10, 158]. It allows us to isolate the pure vdW-type interactions in a term with an explicit longitudinal projection, as discussed in section 2.2.

In this paper we focus on the class of consistent vdW-DF versions (like the vdW-DF-cx version [9]), that arise with an indirect strategy for XC-functional design (as defined and detailed below) and that also adhere to the Dyson/Lindhard screening logic [10, 12]. The design approach can be seen as simply trusting the ACF guiding principle for the vdW-DF method [4, 7, 10, 12, 14, 45]. What consistent vdW-DF versions do, in practice, is to renormalize the physics content of the equation (36) contributions:

$$E_{\tilde{\alpha}}^{\text{res}} - E_{\text{self}} \to E_{\text{xc},\alpha},$$
 (39)

¹ We note that both RPA and the vdW-DF method evaluate the XC energy as a formal contour-integral counting of ZPE shifts [10, 158]. The formal structures of the XC specification in RPA and the vdW-DF method are identical, just given by different models of the longitudinal dielectric functions [10].

$$E_{\rm pl}^{\rm ZPE} \to E_{\rm c, \, vdW}^{\rm nl}.$$
 (40)

This leads to a revised functional specification

$$E_{\rm xc}^{\rm DF} = E_{\rm xc,\alpha} + E_{\rm c, vdW}^{\rm nl}.$$
(41)

What we have effectively implemented in the recent consistent-exchange (spin-)vdW-DF-cx version [9–11, 45], is thus a pure vdW term $E_{c, vdW}^{nl}$ which aims to supplement the non-vdW parts of a GGA-type description (in $E_{xc,\alpha}$).

The reorganization, equations (36), (39), and (40), reflects the interpretation of the vdW-DF method as a mutual electrodynamics coupling of GGA-type XC holes [10]. The same GGA-like screening logic [10] (defined by formal input E_{xc}^{in}) enters in both terms of equation (41). The form of $E_{c, vdW}^{nl}$, expressed in terms of the physics content of E_{xc}^{in} , is another key vdW-DF step [4, 7, 11, 12, 45], that will be summarized later.

3. Screening in formal DFT

Here we present a more systematic discussion of the role of screening in formulations of local, semi-local, and nonlocal XC energy functionals. We motivate the design logic of the vdW-DF method in general. By stressing the role of an effective Dyson equation and a current-conservation criterion, we also prepare for our subsequent presentation (in section 4) of the consistent-exchange vdW-DF-cx version and for an analysis of the screening nature of the vdW-DF-cx components.

3.1. Response in the electron gas

The exact XC energy functional for DFT calculations is given by formal response theory through the ACF [22, 25, 46], equation (1). To detail the discussion, it is convenient to introduce the longitudinal Lindhard dielectric function [61-63]:

$$\kappa_{\lambda}(\omega) = (1 + V_{\lambda}\chi_{\lambda}(\omega))^{-1}.$$
(42)

Here $\chi_{\lambda}(\omega)$ denotes the Fourier transform of equation (13). It is also convenient to define a screened or effective many-body interaction

$$W_{\lambda}(\omega) = \kappa_{\lambda}(\omega)^{-1} V_{\lambda}.$$
(43)

given by the integral equation

$$W_{\lambda}(\omega) = V_{\lambda} + W_{\lambda}(\omega)\tilde{\chi}_{\lambda}(\omega)V_{\lambda}.$$
(44)

This screened interaction is independent of the spin of scattering electrons [169].

The Lindhard analysis of screening gives

$$\tilde{\chi}_{\lambda}(\omega) = \chi_{\lambda}(\omega)\kappa_{\lambda}(\omega), \qquad (45)$$

$$\kappa_{\lambda}(\omega) = 1 - V_{\lambda} \tilde{\chi}_{\lambda}(\omega), \qquad (46)$$

reflecting a Dyson equation for the density-density correlation function

$$\chi_{\lambda}(\omega) = \tilde{\chi}_{\lambda}(\omega) + \tilde{\chi}_{\lambda}(\omega)V_{\lambda}\tilde{\chi}_{\lambda}(\omega) + \cdots$$
$$= \tilde{\chi}_{\lambda}(\omega) + \tilde{\chi}_{\lambda}(\omega)V_{\lambda}\chi_{\lambda}(\omega).$$
(47)

The interaction-kernel function $\tilde{\chi}_{\lambda}(\omega)$ of equation (47) describes the local-field density response [169, 188]. The density change $\delta n_{\lambda}^{\omega}$, equation (16), induced by an external potential $\delta \Phi_{\text{ext}}^{\omega}$, can equivalently be expressed

$$\delta n_{\lambda}^{\omega} = \tilde{\chi}_{\lambda}(\omega) \delta \Phi_{\rm loc}^{\omega}, \tag{48}$$

with $\Phi_{\rm loc}^{\omega} \equiv \kappa_{\lambda}^{-1}(\omega) \delta \Phi_{\rm ext}^{\omega}$.

Physics insight on the local-field response $\tilde{\chi}_{\lambda}$ (or equivalently on screening) leads to guidelines for the design of density functional specifications. For example, a Hubbard-type analysis of the HEG density response [19, 20, 25, 84, 190, 192] to a potential perturbation $\delta \Phi_{\text{loc}}^{\omega}(\mathbf{q})$ (defined by frequency ω and wavevector \mathbf{q}) modifies the RPA by inclusion of a vertex-correction function $\gamma(q = |\mathbf{q}|)$:

$$\tilde{\chi}_{\lambda}(\mathbf{q},\omega) = \frac{\tilde{\chi}_{\lambda=0}(\mathbf{q},\omega)}{1 + V(q)\gamma(q)\tilde{\chi}_{\lambda=0}}.$$
(49)

Here V(q) denotes the Fourier transform of the electron–electron interaction matrix element

$$V(q = |\mathbf{q}|) \equiv \int d\mathbf{r} \frac{e^{i\mathbf{q}\cdot(\mathbf{r}-\mathbf{r}')}}{|\mathbf{r}-\mathbf{r}'|} = \frac{4\pi}{q^2}.$$
 (50)

The function $\gamma(q)$ approximates the interaction vertex Γ at conditions that holds for the HEG. The Hubbard-type response approximation corresponds to the density–density correlation function [25]:

$$\chi_{\lambda}(\mathbf{q},\omega) = \frac{\tilde{\chi}_{\lambda=0}(\mathbf{q},\omega)}{1-\lambda b(\mathbf{q},\omega)},\tag{51}$$

$$b(\mathbf{q},\omega) = V(q)[1-\gamma(q)]\tilde{\chi}_{\lambda=0}(\mathbf{q},\omega).$$
 (52)

The wavevector arguments reflect a Fourier transform in $(\mathbf{r} - \mathbf{r}')$ of, for example, $\langle \mathbf{r} | \chi_{\lambda}(\omega) | \mathbf{r}' \rangle$. The Hubbard approximation for the HEG response, equations (51) and (52), makes it possible to provide an analytical evaluation of the coupling-constant integral. The result is a formal, Hubbard-based specification of the HEG XC energy [25]:

$$E_{\rm xc}^{\rm Hub} = \int \frac{\mathrm{d}u}{2\pi} \frac{\mathrm{d}\mathbf{q}}{(2\pi)^3} \frac{\ln(a(\mathbf{q},\omega=\mathrm{i}u))}{[1-\gamma(q)]} - E_{\rm self},\quad(53)$$

$$a(\mathbf{q},\omega) \equiv [1 - b(\mathbf{q},\omega)]. \tag{54}$$

The form equation (53) reduces to the RPA result for the homogeneous gas in the limit $\gamma(|\mathbf{q}|) \rightarrow 0$. The longitudinal projections, equations (32) and (34), are implicit in the Hubbard-type specification of the HEG XC energy, equation (53). This follows because the translational invariance ensures that $a(\mathbf{q}, \omega)$ is diagonal in q space.

The success of DFT for descriptions of traditional, dense material can be seen as a consequence of including vertex corrections, that is, taking $\gamma(|\mathbf{q}|) \neq 0$. The RPA description fails but an MBPT characterization of response in the near-HEG limit, and of the role of the $V(q)\gamma(|\mathbf{q}|)$ variation at $\mathbf{q} \rightarrow 0$ in particular, formally guides the design of the LDA and GGA XC energy functionals [25, 26, 89, 91, 168, 189]. In practice, the effects of the important $V(q = 0)\gamma(|\mathbf{q}| = 0)$ value is absorbed

1

into the modern QMC-based specification [23] of the LDA XC energy functional E_{xc}^{LDA} , [25, 91].

Going beyond LDA, the initial analysis work concentrated on setting a quadratic nonlocal form [1, 65, 89, 173, 193]:

$$E_{\rm xc} \approx E_{\rm xc}^{\rm LDA} - \frac{1}{2} \int_{\mathbf{r}} \int_{\mathbf{r}'} K_{\rm xc}[n](\mathbf{r}, \mathbf{r}') \left(n(\mathbf{r}) - n(\mathbf{r}') \right)^2.$$
(55)

For DFT calculations, XC functional specifications can be obtained by considering the global density variation $n(\mathbf{r})$ and enforcing XC hole conservation, as done in the averaged/weighted density approximations (ADA/WDA) [173]. They can also be obtained in a perturbation analysis, considering small density fluctuations $n(\mathbf{r}) = n_0 + \delta n(\mathbf{r})$ around a constant (average) density n_0 . In the near-HEG limit, the kernel K_{xc} is translational invariant, set by the average electron density, and given by the Fourier components

$$K_{\rm xc}(\mathbf{q}) \equiv \int d\mathbf{r} \, \mathrm{e}^{\mathrm{i}\mathbf{q}\cdot(\mathbf{r}-\mathbf{r}')} \, K_{\rm xc}(\mathbf{r}-\mathbf{r}'). \tag{56}$$

For density perturbations, $\delta n_{\mathbf{q}}(\mathbf{r}) = \delta n_{\mathbf{q}} e^{i\mathbf{q}\cdot\mathbf{r}}$ of a given wavevector $\mathbf{q} \neq 0$, the near-HEG results equation (55) becomes [91]:

$$E_{\rm xc} \approx E_{\rm xc}^{\rm LDA} + K_{\rm xc}[n](\mathbf{q}) \, |\delta n_{\rm q}|^2. \tag{57}$$

Using the Hubbard response description the MBPT result for the near-HEG kernel is [91]

$$K_{\rm xc}(\mathbf{q}) = -2 \operatorname{V}(q) \gamma(q = |\mathbf{q}|), \tag{58}$$

and thus formally set by the **q** variation of the Hubbard-vertex function $\gamma(\mathbf{q})$. The kernel momentum dependence is expressed [47, 91, 174]:

$$K_{\rm xc}(q = |\mathbf{q}|) = K_{\rm xc}(0) + \frac{\pi}{8k_{\rm F}^4}Z(q)q^2.$$
 (59)

The $\mathbf{q} = 0$ contribution can be ignored since it is already part of the LDA specification [91]. The corrections for semi- and nonlocal XC functionals are reflected in the dimensionless quantity Z(q).

The nature of the local-field response, and hence screening can, in principle, be computed from a diagram analysis [169]; in practice, such analysis is made primarily to extract design guidelines, for example, references [89, 91, 168, 174, 189]. The local-field response can be further separated into spin components of the interaction kernel [188],

$$\tilde{\chi}_{\lambda}(\omega) \equiv \sum_{\mu,\nu} P_{\lambda}^{\mu,\nu}(\omega).$$
(60)

Here $P_{\lambda}^{\mu,\nu}(\mathbf{r}_1, \mathbf{r}_2; \omega)$ describes how the μ -spin density at position \mathbf{r}_1 is affected by a local potential acting on the ν -spin density at position \mathbf{r}_2 . The vdW-type nonlocal-correlation diagram [64, 65, 91]—figure 1c in reference [189] and figure 4c in reference [91]—has diagonal as well as off-diagonal components.

The semilocal (or GGA) XC functionals can, in general, be expressed by the specification of a local energy-per-particle variations $\varepsilon_{xc}^{0}(\mathbf{r})$:

$$E_{\rm xc}^0 = \int_{\mathbf{r}} n(\mathbf{r}) \, \varepsilon_{\rm xc}^0(n(\mathbf{r}), s(\mathbf{r})). \tag{61}$$

A closely related representation is also useful for a discussion of vdW-DF-cx and other nonlocal-correlation XC functional versions [14, 55, 57],

$$E_{\rm xc} = \int_{\mathbf{r}} n(\mathbf{r}) \,\varepsilon_{\rm xc}[n](\mathbf{r}). \tag{62}$$

This latter form merely reflects an expression of the energy density for the XC functional; the energy-per-particle variation is then, as indicated, a functional of the electron density variation. We shall mostly use the energy-per-particle concept for local and semilocal functionals, as indicated by the superscript '0' in equation (61).

In the LDA, the $\varepsilon_{x/c}^{\text{LDA}}$ variation is entirely specified by the local value of the Fermi vector $k_F(\mathbf{r}) = (3\pi^2 n(\mathbf{r}))^{1/3}$. In the near-HEG limit, relevant for the GGA design, the value of ε_{xc}^0 also depends on the scaled form of the local density gradient [4, 7, 25],

$$s(\mathbf{r}) = \frac{|\nabla n(\mathbf{r})|}{[2n(\mathbf{r})k_{\rm F}(\mathbf{r})]}.$$
(63)

For small periodic density variations $\delta n_{\mathbf{q}}(\mathbf{r}) = \delta n_{\mathbf{q}} \mathbf{e}^{\mathbf{i}\mathbf{q}\cdot\mathbf{r}}$, the length of the density gradient is itself proportional to $q = |\mathbf{q}|$, and the same applies to the value of the scaled gradient *s*, equation (63). Following references [7, 47, 91, 174] and using equation (55), we extract the perturbation limit

$$\varepsilon_{\rm xc}^0 - \varepsilon_{\rm xc}^{\rm LDA} \to -\varepsilon_{\rm x}^{\rm LDA} \frac{Z_{\rm xc}}{9} s^2.$$
 (64)

Here $Z_{xc} = Z(q \rightarrow 0)$ is asserted from exchange-like $P^{\mu=\nu}(\omega)$ diagram contributions [7, 89, 91, 174]. Since Z_{xc} is given by the $\gamma(q \rightarrow 0)$ limit, vertex corrections also guide the GGA designs [91, 174].

For a motivation of the design choices made in the vdW-DF method, we will, below, consider a relation between the quasiparticle dynamics, expressed for spin ν , and $\sum_{\mu} W_{\lambda}(\omega) P_{\lambda}^{\mu,\nu}(\omega)$. The effective interaction kernel is here

$$P_{\lambda}(\omega) = \sum_{\mu} P_{\lambda}^{\mu,\nu}(\omega) \tag{65}$$

and we suppress the spin dependence for simplicity in the discussion; in the standard class of spin-balanced problems, we have $P_{\lambda}(\omega) = \tilde{\chi}_{\lambda}(\omega)/2$.

3.2. Screening and exponential re-summation

Exponential re-summation [86, 159, 160, 167] and canonical transformation are related solution strategies for the interacting electron gas problem. The central idea is that the many-particle interactions are effectively screened. For example, the use of a canonical transformation [159, 194–196] $H_{\lambda} \rightarrow \tilde{H}_{\lambda} = \hat{U}H_{\lambda}\hat{U}^{\dagger}$, can produce a significant cancellation of the leading many-particle interaction terms (in \tilde{H}_{λ}) with a suitable choice

of the unitary operator U [159]. The corresponding transformation of the many-particle wavefunction is often expressed

$$\Psi(\mathbf{r}_1,\ldots,\,\mathbf{r}_N)=\exp(J)\Psi(\mathbf{r}_1,\ldots,\,\mathbf{r}_N),\qquad(66)$$

in terms of a so-called Jastrow factor [194, 195] \tilde{J} . This factor is chosen to partly reflect electron correlation and suppress $\tilde{\Psi}$ values whenever $|\mathbf{r}_i - \mathbf{r}_j|$ values are small [194]. The implied screening simplifies the numerical evaluation of the interaction effects, as used, for example, in quantum Monte Carlo calculations [195, 196].

To set the stage for our discussion of screening, we consider the Green function description of the dynamics of a quasi-particle

$$g_{\lambda}(\omega) = g_{0}(\omega) \left[1 + \sigma_{\lambda}(\omega)g_{\lambda}(\omega)\right]$$
$$\equiv g_{0}(\omega) \exp(J_{\lambda}(\omega)), \tag{67}$$

in the presence of an electron–electron interaction and screening. The Dyson correction factor, in the square brackets, is here cast as a cluster or cumulant formulation [86, 163, 164], defined by an exponential factor $J_{\lambda}(\omega)$. This factor is seen as analogous to the Jastrow factor, although in the present MBPT context. The Green function $g(\omega)$ reflects the single-electron excitations to vacuum [169], and the spatial variation defines corresponding orbitals of the quasi-particle dynamics.

Figure 3 shows related Feynman-diagram solutions for the so-called plasmaron model [85, 86, 163, 164], assuming recoil-less interactions. The dynamics of each quasi-particle level, or orbital, is described by an orbital-specific Green function \mathcal{G} . In a single-particle description it is described by \mathcal{G}_0 , represented by arrows; the coupling to other quasi-particle levels is ignored. The wiggly lines represent approximations to the screened interactions, for example, given by the so-called plasmon propagator

$$S_{\lambda}(\omega) \equiv -V_{\lambda}\chi_{\lambda}(\omega), \qquad (68)$$

that tracks $W_{\lambda}(\omega) - V_{\lambda}$, references [21, 22, 83, 85, 86]; the full Green function solution \mathcal{G} then captures the characteristic screening (that affects given quasi-particle level) by the surrounding electron gas, as described by the ZPE dynamics of plasmons [85, 86, 163]. The same formal expansion can also be used to describe the dynamics of a level interacting with virtual vibrational excitations [164]. The top panel of figure 3 shows the form of a traditional Dyson expansion subject to assumption of recoil-less interactions, making it clear that vertex corrections play an important role in setting the quasi-particle dynamics.

The bottom left panel of figure 3 shows the elegance of the cumulant approach for solving the time evolution $\mathcal{G}(t)$ of a specific orbital, again under the assumption of recoil-less interactions. This cumulant approach adapts the ideas of the linked-cluster expansion [167] to the description of the quasi-particle dynamics [86, 160, 161]. The point is that the linked-cluster expansion reduces to just a single connected diagram for which there exists a complete analytical solution [86, 163, 164].

Importantly, the cumulant expansion provides an automatic inclusion of all screening and vertex correction effects, relative to a stated approximation for the plasmon (or phonon) propagator. The underlying assumption of recoil-less interactions holds, for example, for a description of core levels [85, 86] but also for quasi-particle states near the Fermi level [163]. More generally, the exponential-re-summation idea can be used to extract potentially highly accurate solutions from MBPT, focusing on simple diagrams [86, 163, 164].

The bottom-right panel of figure 3 shows the cumulantexpansion idea as represented instead in the frequency domain [164], highlighting a formal similarity with the GW approximation [163, 188]. In frequency space, the expansion of the quasi-particle dynamics in the electron-plasmon coupling yields [164]:

$$\mathcal{G}_{0}(\omega)e^{J_{\lambda}(\omega)} = \mathcal{G}_{0}(\omega)[1 + J_{\lambda}(\omega) + \cdots]$$
$$= \mathcal{G}_{0}(\omega)[1 + \sigma^{(1)}_{\mathcal{G}W_{\lambda}}(\omega)\mathcal{G}^{0}(\omega) + \cdots], \quad (69)$$

$$\sigma_{\mathcal{G}W_{\lambda}}^{(1)}(\omega) = \mathbf{i} \int_{\omega'} \mathcal{G}_0(\omega - \omega') W_{\lambda}(\omega'), \tag{70}$$

again under assumption of recoil-less interactions. Setting the exponential re-summation factor,

$$J_{\lambda}(\omega) = \sigma_{\mathcal{G}W_{\lambda}}^{(1)}(\omega)\mathcal{G}_{0}(\omega), \qquad (71)$$

allows us to capture vertex corrections implicitly. The use of this exponential re-summation factor allows a $\mathcal{G}W^{(1)}$ -based account to be meaningful for metals, at least near the Fermi level [163, 164].

The vdW-DF method aims to capture the vdW forces without loosing the response and screening logic that underpins the GGA success. The electron-gas response provides a formal definition of the XC hole, equation (17), and, in turn, the ACF functional specification, equation (5). The screening, including vertex-correction effects, enters in the specifications of the local-field response [6, 7, 25, 89, 91, 174, 189] $\tilde{\chi}_{\lambda}(\omega)$ as well as in the Dyson-equation specification equation (47) of $\chi_{\lambda}(\omega)$, and hence in $\kappa_{\lambda}^{-1}(\omega)$.

The cumulant expansion [86, 163], analyzed in frequency space [164], is a guide for understanding the vdW-DF design logic. Screening in vdW-DF is described by constructing a scalar, but fully nonlocal, model for a local-field susceptibility $\alpha(\omega)$ that reflects a GGA-type response behavior.

Figure 4 shows, in panels (a) through (c), a compact formulation of the complete Feynman-diagram evaluation of the electron thermo-dynamical potential [167], the screening WP of the effective interaction [188], and the quasi-particle dynamics [169], respectively. The triangle represents the general electron–electron interaction vertex form, denoted Γ , while arrows here depict the fully interacting electron Green functions g. The wiggly lines represent the screened interaction W. All internal coordinates (at interaction vertices) are integrated out.

Importantly, the linked-cluster result (panel a), also defines full specifications of both the interaction screening $W_{\lambda}(\omega)P_{\lambda}(\omega)$ (panel b) and of $\sigma_{\lambda}(\omega)g_{\lambda}(\omega) \approx J_{\lambda}(\omega)$ (panel c);



Figure 4. Formal linked-cluster result for the total energy in the interacting electron gas (panel a). We suppress spin indices on electron Green functions *g* and on the screened electron–electron interaction *W*; the triangle denotes the full vertex function. The same vertex function also enters the formal specifications of the spin-resolved local-field response function $P(\omega)$ and of the electron self-energy $\sigma(\omega)$. The cluster expansion result leads to a relation between the factor $W(\omega)P(\omega)$ (panel b) and the Dyson-correction factor $\sigma(\omega)g(\omega)$ (panel c). The former screens and thus defines the effective electron–electron interaction [188], while the latter guides a cumulant-expansion approximation for the quasi-particle dynamics [86, 163, 164].

they arise by pulling out W and g, respectively. The commonorigin argument motivates a mutual relation

$$\sigma_{\lambda}(\omega)g_{\lambda}(\omega) \sim -W_{\lambda}(\omega)P_{\lambda}(\omega)$$
$$= -V_{\lambda}\chi_{\lambda}(\omega)/2, \qquad (72)$$

where we focus on the quasi-particle dynamics for a given spin, and where the last line holds when $P_{\lambda}(\omega) = \tilde{\chi}_{\lambda}(\omega)/2$. The Hedin equations [188] prove equation (72) when expressed in a form with full frequency integration, appendix A.

We see equation (72), and the usefulness [86, 163, 164] of the cumulant expansion idea, equations (69)–(71), as a guide to construct $\tilde{\chi}_{ACF}$ approximations, for use in the vdW-DF method. By relying on an exponential re-summation, we can build from a simple model local-field susceptibility $\alpha(\omega)$, as long as we also enforce the projection equation (32). A GGAlike internal functional is a good place to nucleate this response modeling [4, 7, 8, 10].

The vdW-DF design strategy for truly nonlocal functionals thus consists of a sequence of steps. First we use

$$\kappa_{\text{int}} = 1 - \chi_{\text{int}}^{2} V$$
$$\sim 1 + 4\pi \alpha(\omega) \equiv \epsilon(\omega), \tag{73}$$

to implicitly define an internal (or lower-level) dielectricfunction approximation $\epsilon(\omega)$; this follows by partial integration since $\nabla^2 V = -4\pi$. Second, we use

$$\epsilon(\omega) = \exp(S_{\rm xc}(\omega))$$
$$= 1 + S_{\rm xc}(\omega) + \cdots, \qquad (74)$$

to cast the associated scalar-model susceptibility $\alpha(\omega)$ —and hence $\tilde{\chi}_{ACF}(\omega)$ —through an exponential re-summation,

$$4\pi\alpha(\omega) = S_{\rm xc}(\omega) + \sum_{j=2}^{\infty} \frac{1}{j!} (S_{\rm xc}(\omega))^j, \tag{75}$$

in an effective plasmon propagator S_{xc} . As indicated by the subscript, this effective plasmon propagator is set so that it is consistent with the energy-per-particle variation of an internal semilocal XC functional [8, 10, 12], as summarized in appendix B. Finally, we use equation (32) together with an effective Dyson equation (30) to complete the specification of the integrand $\kappa_{ACF}(\omega) \approx \kappa_{long}(\omega)$ in the ACF recast, equation (18). These steps describe a full implementation of the vdW-DF method; for the design of the popular general-geometry vdW-DF versions, the resulting description is also expanded in S_{xc} , references [4, 7, 12].

We note that the implied combination of an exponential re-summation, equation (74) and a Dyson expansion, equation (30), captures higher-order and truly nonlocal correlation effects [1, 12, 65, 91]. This internal semilocal XC functional has no correlations beyond those reflected in LDA [4, 7]. In other words, all gradient-corrected and truly nonlocalcorrelation effects must emerge from the third of the abovelisted steps. Still, the summation in $S_{\rm xc}(\omega)$ provides the proper framework for capturing screening and vertex effects in $\alpha(\omega)$. Meanwhile, the combination of equations (30) and (32) ensures a longitudinal projection in $\kappa_{\rm ACF}^{-1}(\omega) = 1 + V\chi_{\rm ACF}(\omega)$. This projection is, as discussed in section 2.2, sufficient to ensure that equation (18) also captures the asymptotic vdW interactions.

For a more detailed motivation of the vdW-DF design strategy, we consider an exponential re-summation for the Lindhard screening function

$$\kappa_{\lambda}^{-1}(\omega) \equiv \exp(-F_{\lambda}(\omega))$$
 (76)

$$= 1 + V_{\lambda}\chi_{\lambda}(\omega) = 1 - S_{\lambda}(\omega). \tag{77}$$

For a particular coupling constant λ , we have a direct specification of the screening

$$F_{\lambda}(\omega) \equiv \ln (\kappa_{\lambda}(\omega))$$

= $(1 - \kappa_{\lambda}^{-1}) + \frac{1}{2}(1 - \kappa_{\lambda}^{-1})^{2} + \cdots$
= $(-V_{\lambda}\chi_{\lambda}) + \frac{1}{2}(-V_{\lambda}\chi_{\lambda})^{2} + \cdots$
= $S_{\lambda}(\omega) + \frac{1}{2}S_{\lambda}(\omega)^{2} + \cdots$ (78)

The re-summation can be expected to converge fast, $F_{\lambda} \approx S_{\lambda}$ as it describes the electrodynamics directly in terms of the screened response $\chi_{\lambda}(\omega)$. We also note that

ŀ

$$\begin{aligned} \kappa_{\lambda}^{-1}(\omega) &= 1 + V_{\lambda}\chi_{\lambda}(\omega) \\ &= 1 + W_{\lambda}\tilde{\chi}_{\lambda}(\omega) \\ &\sim 1 - 2\sigma_{\lambda}(\omega)g_{\lambda}(\omega) \end{aligned}$$
(79)

follows from equation (72). Again, the factor of 2 arises from summation over spin contributions in the quasi-particle description. equation (79) is consistent with a classic MBPT

text-book demonstration [169] that the electron–electron interaction energy can be computed both from knowledge of the external-field response $\chi(\omega)$ and from the $\sigma(\omega)g(\omega)$ product.

Taking equations (78) and (79) together suggests that we can set the exponential factor in the dielectric function

$$F_{\lambda}(\omega) \sim 2\sigma_{\lambda}(\omega)g_{\lambda}(\omega).$$
 (80)

This identification should be compared with the re-summation factor $J_{\lambda}(\omega)$ that characterizes the quasi-particle dynamics, equation (71), and the Jastrow factor \tilde{J} in equation (66).

We observe that the vdW-DF reliance on exponential resummations of dielectric functions, equations (19), (76), and (80), provides the correct formal structure for capturing a range of important vertex effects in the electron-gas response. We recall that the cumulant formulation, equation (76), for $\kappa_{\lambda}^{-1}(\omega)$ was, in fact, originally set up to capture the range of vertex-correction and hence screening effects for a core electron interacting with the itinerant electron gas [86]. This was expressed through the quasi-particle dynamics in the plasmaron model [85, 86]. Hedin as well as Gunnarsson and co-workers noted and demonstrated that the model ability to capture the range of vertex corrections for recoil-less interactions has broader use. It works, for example, for descriptions of quasi-particles near the Fermi surface [163, 164]. Finally, we find that the Hedin equations, appendix A, suggests the connection equation (72) between the quasi-particle dynamics and the cumulant or exponential re-summation, equation (76), for $\kappa_{\rm ACF}(\omega).$

Of course, for the ACF specification equations (18) and (19) we seek a λ -averaged evaluation of the response. By construction [10, 47] we have

$$-\int_{0}^{1} \mathrm{d}\lambda \, V\chi_{\lambda}(\omega) = \ln(\kappa_{\mathrm{ACF}}(\omega)), \tag{81}$$

while equation (78) gives

$$\int_{0}^{1} \frac{d\lambda}{\lambda} \ln(\kappa_{\lambda}(\omega)) \approx -\int_{0}^{1} \frac{d\lambda}{\lambda} V_{\lambda}\chi_{\lambda}(\omega)$$
$$= -\int_{0}^{1} d\lambda V\chi_{\lambda}(\omega). \tag{82}$$

That is, we can see $\ln(\kappa_{ACF}(\omega))$ as a mean-value evaluation given a characteristic exponent

$$F_{\rm ACF}(\omega) \equiv \ln(\kappa_{\rm ACF}(\omega)) \sim F_{\lambda_{\rm eff}(\omega)}.$$
 (83)

We finally note that treating $\kappa_{ACF}(\omega)$ like an actual Lindhard dielectric function $\kappa_{\lambda_{eff}}(\omega)$ is fully consistent with the logic of the coupling-constant analysis of XC functionals [14, 48–51, 90, 197]. For example, based on the coupling-constant analysis of the consistent-exchange vdW-DF-cx version [14, 15], we find that the actual XC functional E_{xc}^{DF} should be seen as a suitable average of the $\lambda = 0$ (all-exchange) limit and $\lambda \rightarrow 1$ (strongly correlated) limit [15]. This observation reflects the behavior in the associated XC holes [14]. The corresponding response description differs from that of the physical system but it is still defined by a long-range particle interaction, $V_{\rm ACF} \sim V_{\lambda_{\rm eff}}$.

Overall, we are led to trust a lower-level internal response description for a characterization of $F_{int}(\omega) \approx \ln(\kappa_{int}(\omega)) - S_{xc}(\omega)$, precisely because we rely on exponential re-summation. We note that a direct use of equations (69)–(71) leads to an explicit specification,

$$F_{\lambda_{\rm eff}}(\omega) \sim 2\sigma_{\rm GW_{\lambda_{\rm eff}}}^{(1)}(\omega)g_0(\omega)?$$
(84)

However, this would be an uncontrolled approximation because we do not know the extent that we have thus respected the longitudinal projection. We also do not know the value of λ_{eff} . Instead, in the vdW-DF method, we seek to retain the elegance of the cumulant-expansion approach [86, 163, 164] with an implicit construction of α from S_{xc} . We do that by setting $S_{\text{xc}} = \ln(\epsilon)$, where $\epsilon \approx \kappa_{\text{int}}$ denotes an approximation set by an internal functional.

3.3. Electrodynamics nature of XC functionals

We make three observations to further motivate and detail the logic of the vdW-DF method and the formulation of the consistent vdW-DF versions. The exact vdW-DF framework contains the constraint-based semilocal functionals as a limit [4, 7, 10, 12]. This makes it possible to also discuss the connection between the semilocal and the truly nonlocal functionals in some detail.

First, the identification of the ACF recast, equations (18) and (19), with the screening at some effective coupling constant ($0 < \lambda_{eff} < 1$), equation (83), implies that we should leverage insight from the theory of light–matter interactions and screening [63] in the vdW-DF designs.

It is, for example, interesting to use equation (79) to discuss optical excitation of a quasi-particle orbital of energy ϵ_r . We assume that we are at a frequency ω far from collective excitations and that there is only a limited amount of actual transitions, meaning $\Im \sigma(\omega) \to 0$. The imaginary part of the right-hand side of equation (79) is then, for $\omega \approx \epsilon_r$, set by the spectral density, i.e., set by the relevant inelastic excitation from ϵ_r up to the vacuum level [169]. Meanwhile, the imaginary part of the left-hand side of equation (79) is given by $\Im \chi(\omega)$, and effectively set as an evaluation of Fermi's golden rule for inelastic transitions [63]. The transition rate must, at $\omega \approx \epsilon_r$, have significant contributions from such optical excitations to the vacuum level. It is gratifying that the real part of the model dielectric function $\kappa_{\lambda}(\omega \approx \epsilon_r)$ define the screening of such excitations.

Moreover, since we have build both $\tilde{\chi}_{ACF}(\omega)$ and $g_{\lambda_{eff}}(\omega)$ from exponential re-summations, we expect the two expansions to be related. We note that the cumulant approach for the quasi-particle dynamics, given by equation (72), has accuracy near the Fermi level (and for localized orbitals). This suggests that our exponential re-summation for $\epsilon(\omega)$ has the formal structure to be accurate for screening at corresponding frequencies.

For a practical design framework, we set the exponential re-summation [8, 12, 47]

$$\epsilon(\omega) = \mathrm{e}^{S_{\mathrm{xc}}(\omega)}.\tag{85}$$

from an internal semi-local XC functional

$$E_{\rm xc}^{\rm in} = \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(\epsilon(\mathrm{i}u))\} - E_{\rm self}$$
$$= \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{S_{\rm xc}(\mathrm{i}u)\} - E_{\rm self}.$$
(86)

The formal structure is the same as that used for a direct design of GGA functionals, equation (89), as discussed below. This allows us to use equation (86) to set the details of $S_{xc}(\omega)$ approximation, as summarized in appendix B and elsewhere [4, 7, 9, 12].

Second, approximations for the ACF electrodynamics and for $\kappa_{ACF}(\omega)$ should be built in compliance with constraints and with the Lindhard screening logic [63].

We should seek approximations that adhere to the Dyson equation [61, 62], equation (30) for the effective ACF electrodynamics response. This follows because we should think of the ACF recast as reflecting an actual electrodynamics, given by a long-ranged interaction $V_{\text{ACF}} \sim V_{\lambda_{\text{eff}}}$.

Also, approximation for this ACF electrodynamics must be made subject to a current-conservation criterion [10], that is, $\kappa_{ACF}(\omega) \approx \kappa_{long}(\omega)$. We are constructing a ground-state XC functional that reflects an electrodynamics coupling of the ZPE dynamics of XC holes, equation (36). However, zero-point vibrations, like plasmons, involve electron currents and we should view vdW-inclusive XC functionals (for ground state DFT) as a limit [198, 199] of time-dependent DFT [200] or of current-density functional theory [201, 202].

The longitudinal projection, equation (33), is hardwired into the resulting vdW-DF functional description of nonlocalcorrelation effects, equation (2), as explained below. As detailed in section 4, the consistent-exchange vdW-DF-cx version seeks to enforce both the Dyson criterion, equation (30), and the longitudinal projection in the full nonlocal-functional specification.

Third, the physics of the ACF electrodynamics and dielectric function $\kappa_{ACF}(\omega) \approx \kappa_{long}(\omega)$ is at the same time partly known and incompletely explored. Equations (21) and (22) provide a direct link between the formally exact XC hole, equation (17), and $\kappa_{ACF}(\omega)$. On the one hand, we know a lot about this ACF dielectric function when it is used to describe semilocal functionals via equation (22). On the other hand, the Mahan and Ashcroft analysis of the nature of vdW forces [64, 65, 158], as well as the Anderson–Langreth–Lundqvist–Dobson launching of the vdW-DFs [1, 2, 166, 203–205], suggest that we must seek truly nonlocal formulations of n_{xc} and $\kappa_{ACF}(\omega)$.

In the vdW-DF method, we seek to port the GGA experience in the former to enhance the quality of the latter.

3.4. Nonlocal and semilocal XC functionals

Interestingly, our analysis of the screening nature of functionals leaves us with two suggestions for actual designs, a direct and an indirect formulation. They differ in how we connect the physics content of approximations like equation (34) to the underlying formally exact (λ -averaged) response description contained in equation (1). *The direct approach* is using equation (77) to just recoup the standard ACF

$$E_{\rm xc} \equiv \int_0^\infty \frac{du}{2\pi} \operatorname{Tr}\{\langle (1 - \kappa_\lambda^{-1}(iu)) \rangle_\lambda\} - E_{\rm self} \\ = \int_0^\infty \frac{du}{2\pi} \operatorname{Tr}\{\langle S_\lambda(iu) \rangle_\lambda\} - E_{\rm self},$$
(87)

where we have used

$$\langle A_{\lambda} \rangle_{\lambda} \equiv \int_{0}^{1} \frac{d\lambda}{\lambda} A_{\lambda}.$$
 (88)

to define the λ averaging.

The direct-design approach can also formally be seen as a result of both truncating the exponential re-summation equation (19) and the logarithm in equation (18) to lowest relevant order in $V\chi_{\lambda}(\omega)$. Since equation (87) is, in principle, exact by itself, it is clear that there are important cancellations in ACF recasting and in the exponential summation. Use of a direct approach (relying directly on the original ACF formulation), equation (87) is a highly motivated design strategy, as long as we are actually able to directly assert a good, robust approximation for $\langle (1 - \kappa_{\lambda}^{-1}(iu)) \rangle_{\lambda} = -\int_{0}^{1} d\lambda V\chi_{\lambda}(iu)$.

In practice, the direct design strategy rests on making systematic approximations that are based on formal MBPT for the HEG and for the weakly-perturbed electron gas [4, 7, 10, 11, 13–15, 21, 23, 25, 26, 28, 29, 48, 50, 64, 65, 89, 91, 168, 174]. The electron response is dominated by plasmons, i.e., collective excitations defined as zeros of the dielectric function of the system. We thus expect the semilocal (GGA) functional descriptions to be fairly approximated by a simple model for a plasmon propagator S'_{xc} . We can, for example, use the formulation inspired by a formal MBPT gradient expansion [47, 87],

$$E_{\rm xc} \approx E_{\rm xc}^0 \equiv \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{S_{\rm xc}'(\mathrm{i}u)\} - E_{\rm self},\qquad(89)$$

appendix B. The specification (89) is equivalent to LDA/GGA formulations given the semilocal XC hole

$$n_{\rm xc}^{0}(\mathbf{r};\mathbf{r}'-\mathbf{r}) = -\delta(\mathbf{r}-\mathbf{r}') - \frac{1}{2\pi n(\mathbf{r})} \int_{0}^{\infty} \frac{\mathrm{d}u}{2\pi} \nabla_{\mathbf{r}}^{2} \langle \mathbf{r} | S_{\rm xc}'(iu) \rangle | \mathbf{r}' \rangle$$
(90)

It is a problem that use of even a double-pole plasmon approximation for S_{xc}' leads to a GGA-type functional in the direct-design approach. As summarized in appendix B and in references [4, 7, 10, 12], the local energy-per-particle variation $\varepsilon_{xc}^{in}(\mathbf{r})$ of the semi-local internal function E_{xc}^{in} specifies the local plasmon dispersion [4, 7, 25], and invariably reduces the XC functional specification equation (90) to that of a semilocal form. The semilocal XC functional form has no clear mechanism for reflecting truly nonlocal correlation effects, including pure vdW attraction [64, 65, 91], section 2.

The indirect, or vdW-DF, approach, is to instead exploit equations (18) and (19) for higher-order expansions in the

plasmon propagator, building from equation (85). The exponential re-summation $\epsilon(\omega) = \exp(S_{xc}(\omega))$ is, by itself, exclusively used to reflect a GGA behavior minus pure vdW forces.

However, the indirect design provides a balanced account of general interactions by also explicitly enforcing a longitudinal projection [4, 10], that is, in the vdW-DF method, we use equation (34), with $\epsilon(\omega)$ inserted for $\epsilon(\omega)$, to define the XC functional

$$E_{\rm xc} \approx E_{\rm xc}^{\rm DF} \equiv \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(\kappa_{\rm long}(\mathrm{i}u))\} - E_{\rm self}.$$
 (91)

The implicit ACF approximation, $\kappa_{ACF} \approx \kappa_{long}$, secures a strict enforcement of current conservation in the description of the electrodynamics response [10].

The vdW-DF method is, to our knowledge, the first example of this indirect (Dyson-based) functional design approach. There is seamless integration with the underlying GGA formulation, given by $\epsilon(\omega) = \exp(S_{\rm xc}(\omega))$, as long as we properly balance exchange and correlation terms, as discussed in section 4.

To define specific versions in this vdW-DF method we assert the local field response, given by $\tilde{\chi}_{ACF}(\omega)V$, from a trusted MBPT analysis [4, 7, 91]. It is sufficient to formulate the input in terms of an effective scalar local-field susceptibility, given by $\alpha = \alpha I$ and equation (75), where I is the unit matrix in spatial coordinates; the associated (internalfunctional) approximation for the scalar dielectric function is given $\epsilon = \epsilon I$ with $\epsilon(\omega) = 1 + 4\pi\alpha(\omega)$. We use the MBPT analysis to set the gradient-corrected exchange of the internal semilocal XC functional, equation (86), and thus the details of the approximation for the vdW-DF plasmon propagator $S_{xc}(\omega)$, appendix B. The description of $\tilde{\chi}_{ACF}(\omega)V$ then follows by the exponential re-summation, equation (75).

The lowest-order contribution in equation (94) is just the internal functional itself. The vdW-DF method has a leading term given as

$$E_{\text{xc},n=1}^{\text{DF}} = \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\nabla S_{\text{xc}}(\mathrm{i}u) \cdot \nabla G\} - E_{\text{self}}$$
$$= E_{\text{xc}}^{\text{in}}, \tag{92}$$

where $G = -V/4\pi$. The last line follows since $\nabla^2 G = 1$.

The vdW-DF method yields a nonlocal-correlation energy approximation [2-4],

$$E_{\rm c}^{\rm nl} = \int \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{\ln(\nabla\epsilon(\mathrm{i}u)\cdot\nabla G) - \ln(\epsilon(\mathrm{i}u))\},\tag{93}$$

that is set entirely by the density variation [1, 6, 32, 53, 59, 165, 180, 205–211] and hence by occupied orbitals [31].

The vdW-DF version specifications [4, 8, 9, 11, 12] are given by equations (91), (34), (86) and (93). They are, in principle, strictly nonempirical [4, 10, 12, 30, 45] since the evaluation of equation (93) is defined by a systematic expansion in a known approximation for the plasmon propagator S_{xc} . In practice, the resulting functional specification, equation (3), normally includes a cross-over term, δE_x^0 .

Moreover, in the popular general-geometry vdW-DF versions [4, 5, 7–9, 12, 40–43, 212] the formal nonlocal-correlation energy specification equation (93) is expanded to second order in $S_{\rm xc}$, using $G = -V/4\pi$ and

$$\chi_{\rm ACF} V \approx -\nabla \left[S_{\rm xc} + \frac{1}{2} S_{\rm xc}^2 \right] \cdot \nabla G + (\nabla S_{\rm xc} \cdot \nabla G)^2.$$
(94)

The two terms grouped in the square bracket in equation (94) are part of an expansion for $\tilde{\chi}_{ACF}$, while the third term is the lowest-order contribution to the screening implied in equation (30). The combination of the second and third terms in equation (92) defines equation (2). However, we may also retain more or all steps in the expansion, equation (94), as done in the immediate precursor, namely the layered-geometry vdW-DF0 version [2, 3, 6].

Importantly, when expanding to order n = 2 (or more), we arrive at a truly nonlocal form for the description of nonlocal correlations. The vdW-DF method comes with a universal-kernel evaluation [4, 5] of the nonlocal-correlation term

$$E_{\rm c}^{\rm nl} = \frac{1}{2} \int_{\mathbf{r}} \int_{\mathbf{r}'} n(\mathbf{r}) \,\phi(n(\mathbf{r}), n(\mathbf{r}'), s(\mathbf{r}), s(\mathbf{r}')) \,n(\mathbf{r}'). \tag{95}$$

The universal-kernel evaluation, the definition of associated effective potential, and its use in efficient DFT calculations are detailed elsewhere [7, 12, 17, 18, 47, 60, 171].

Also, the nonlocal-correlation evaluation equation (95) has seamless integration [4, 5] with LDA and it is consistent with equation (55). This follows both because it nominally involves a double spatial integral over the density distribution and because equation (2) is easily shown to vanish in the HEG limit [4]. At the same time, neither equation (95) nor equation (55) should be seen as just a two-density-point summation, for example, as discussed in reference [10]. It captures the mutual electrodynamics coupling of two electron–XC-hole systems (that are centered at positions **r** and **r**').

It should be noted that the input $\epsilon(\omega)$ to any such vdW-DF version or variant, reflects the screening that exists in the GGA dielectric function $\epsilon(\omega) = \exp(S_{xc}(\omega))$. As such, the vdW-DF method provides a systematic extension that captures vdW forces as already screened by, for example, itinerant electrons [10, 64, 65].

Finally, we mention that care must be taken when building the full vdW-DF functional designs from an understanding of screening contained in a GGA functional. We nucleate the description around an internal semi-local functional E_{xc}^{in} . Appendix B documents that it is therefore sufficient to approximate $\ln(\epsilon(\omega)) = S_{xc}(\omega)$ by a double-plasmon-pole approximation $S_{xc}(\omega)$ that reflects the logic of a gradient expansion [4, 47, 87]. However, the nonlocal-correlation effects cannot be limited to the vdW binding mechanism discussed in section 2, references [10, 64, 65]. The second term of equation (36) must also reflect the exponential re-summation implied in asserting, $S_{xc}(\omega) = \ln(\epsilon)$.

The second term $E_{\text{xc},\alpha}$ in equation (41) reflects the poles of $\alpha(\omega) = (\epsilon(\omega) - 1)/4\pi$. We keep local XC effects together in $E_{\text{xc}}^{\text{in}}$, and hence in $S_{\text{xc}}(\omega)$, so clearly $E_{\text{xc},\alpha} \neq E_{\alpha}^{\text{res},\text{HF}} - E_{\text{self}}$. By the cumulant expansion [86, 163, 164], the electron dynamics (in $g_0(\omega)^{-1}g_{\lambda_{\text{eff}}}(\omega)$) also reflects terms that are quadratic

in $S_{\rm xc}$ and thus truly nonlocal (as we detail in appendices **B** through **D**). If the inner functional $E_{\rm xc}^{\rm in}$ already reflected gradient-corrected correlation, we could be double counting.

The vdW-DF solution strategy is that of the Occam's razor, that is, to simply set the semilocal inner functional as LDA plus gradient-corrected exchange. The idea is to let the vertex correction effects in GGA correlation emerge in $E_{xc,\alpha}$, rather than dealing with a potential double counting. However, this Occam solution strategy is only partly motivated by the analysis in references [86, 163, 164]. Important semilocal-correlation effects, for example, reflecting vertex corrections might be missing and we must validate that vdW-DF designs also works for traditional dense-matter challenges, for example, references [45, 95, 191].

4. Screening effects in consistent vdW-DF versions

This section summarizes the rationale for the consistentexchange vdW-DF-cx version. It does so in terms of illustrations of accuracy in computing the density variation and based on formal MBPT arguments. The section furthermore presents a detailed analysis of the underlying electrodynamics description. This section therefore discusses a different splitting of vdW-DF-cx terms than what is presented in equation (3). Appendix C explains how the electrodynamics nature of the vdW-DF method suggests an ordering of general nonlocalcorrelation-energy terms, going beyond the truncated expansion of the vdW-DF nonlocal-correlation energy.

Finally, to make it possible to compute and thus explore the electrodynamics analysis in practice, this section presents universal-kernel calculations of the individual pure-vdW and cumulant components. They describe the functional contributions to quadratic order in S_{xc} and appendix D provides details of the universal-kernel derivation and evaluation.

4.1. Look how good the densities are: molecular libration modes and hard-matter thermo-physics

The key DFT concept is the electron density variation. There are concerns that while many of the recent XC density functional developments may improve the description of energies, they tend to destroy the description of the electron densities [220]. We share the opinion that the real test of a functional usefulness and robustness lies in its ability to accurately predict the electron density variation in general systems. Of course, it is also relevant to test the accuracy in descriptions of reaction and process energies, for example, as in figure 1. However, to be useful as a general-purpose tool [45], DFT with the vdW-DF-cx [9] approximation must be accurate on the electron density from soft molecular matter [98, 99], where there are important sparse regions [32], and to hard traditional matter [45, 95], where the electron distribution is dense.

Good discriminators for quality in the density description are highly accurate predictions of atomic structure, of atomic vibrations and phonons, as well as of surface-specific work functions. The relevance of the first discriminator follows from the Hohenberg–Kohn theorem [221]: the structure-optimized nuclei positions \mathbf{R}_i must uniquely reflect the ground-state density variation $n(\mathbf{r})$. The relevance of the third follows since the work function reflects the variation (across the surface) in the full electrostatic potential $\Phi(\mathbf{r})$ that acts on the electrons; the work functions and work-function shifts (computed for specific surfaces) is an established indicator of quality of XC functionals, for example, as discussed and used in references [25, 89, 222–226].

The relevance of finding accurate vibrations follows since we work in the standard Born–Oppenheimer or adiabatic formulation of DFT. We do not treat the nuclei dynamics directly, since the problem only depends parametrically on the nuclei positions, denoted \mathbf{R}_i . Instead we compute DFT results for another (related) electrostatic potential $\Phi_{\mathbf{R}_i}(\mathbf{r})$, that influences a given nucleus *i*. It contains the Hartree potential from the electron density as well as the electrostatic-potential contributions that are produced at the rest of the nuclear positions, $\mathbf{R}_{j\neq i}$. For an infinitesimal displacement, the restoring force on nucleus *i* must be given exclusively by $\Phi_{\mathbf{R}_i}(\mathbf{r})$, as discussed, for example, in reference [7]. These restoring forces are a direct reflection of the electron-density variation, $n(\mathbf{r})$, and they define the phonons, and vibrations in general [227].

Figure 5 highlights the vdW-DF-cx performance on density quality. The figure summarizes two sets of vdW-DF-cx studies of the oligoacenes molecular crystals and of thermo-physical properties in the full set of nonmagnetic transition metals [95, 98, 99]. Computational details are given in the cited papers. Both panels assess the density-quality performance in term of the first (structure) and second (vibration) discriminators. We shall return to a discussion of a quality assessment of vdW-DF-cx in terms of the surface work function discriminator in section 6.

The top panel documents that the vdW-DF-cx results for the weak libration and rocking modes of the naphthalene crystals are in exceptional agreement with low-temperature measurements [99, 213]. It is important to note that this vdW-DF-cx phonon prediction is computed free of all experimental input, on structure or otherwise. The phonon dispersion is computed for the lattice constants and structure angles that result with full stress and atomic relaxations in vdW-DFcx. The vdW-DF-cx structure predictions are, in turn, in close agreement with low-temperature measurements across the set of oligoacene structures [98]. The vdW-DF-cx is found able to predict the molecular structure and vibrational response ahead of synthesis.

We also demonstrated that vdW-DF-cx structure predictions are accurate enough to, in principle, motivate (computationally costly) GW and Bethe–Salpeter calculations of the molecular-crystal optical response ahead of experimental input [98]. The ability to genuinely predict, for example, solar-cell function before actual synthesis has potential for accelerating, for example, molecular-based green-technology developments. Here, we take the set of oligoacene results [98, 99] as a validation that consistent vdW-DF-cx functional is accurate on the electron-density variation in sparse, molecular matter.



Figure 5. Top panel: phonon dispersion for the set of intermolecular (libration and rocking) modes in naphthalene, contrasting neutron measurements [213] (black) and vdW-DF-cx calculations (red) obtained with full stress and atomic-position relaxations [99]. The phonons are computed at the vdW-DF-cx result for the lattice constants and structure angles. Bottom panel: comparison of vdW-DF-cx and PBE, PBEsol, and AM05 [214] performance for nonmagnetic transition metals. The panel summarizes calculations in ground-state DFT but corrected using the quasi-harmonic approximation [215] that is evaluated from full phonon calculations [216] in each of the XC functionals. These calculations were performed in the plane-wave code VASP [217, 218] using normal or hard PAW setups (as noted by the legend). The atomic reference energies are, however, corrected by a study of the atomic spin polarization, as obtained using the proper spin vdW-DF-cx implementation that is available in QUANTUM ESPRESSO [11, 36, 95, 219]. The panel compares, among the functionals, the percentage errors (bars) as well as standard deviation (vertical lines) for the set of thermo-physical properties in a comparison with room-temperature measurements. The panels are adapted with permission from references [99] and [95], respectively. Copyright (2016 and 2017) by the American Physical Society.

The bottom panel of figure 5 illustrates what vdW-DFcx can do for characterizing transition-metal structure and thermo-physical properties at room temperature. As discussed in references [95, 113, 121], the vdW-DF-cx not only works well at describing the structure at a Born–Oppenheimer level, in raw DFT, but it has a robust description of the phonons that emerge with variations of both cubic and hcp lattices, for bulk and layered structures. At least for some of the cubic nonmagnetic transition-metal cases, we could have just compared our raw-DFT vdW-DF-cx structure characterization with measurements of structure and cohesive energy that are also back-corrected for ZPE and thermal expansion effects [228–230]. However, the back correction introduces a further dependence on experimental input. Instead we chose to combine the robust phonon description of vdW-DF-cx (and of highly successful GGA descriptions: PBE [28], PBEsol [29], and AM05 [214]) with the quasi-harmonic approximation [215, 216] for a full (or native) DFT determination of ZPE and thermal-expansion effects as a function of temperature [95].

The panel shows a comparison of performance for the set of thermo-physical properties: volume, main lattice constant (often denoted a, extracted for both cubic and hcc lattice structures), cohesive energy, bulk modulus and thermal-expansion coefficient. These are computed at room temperature and can be compared directly with room-temperature measurements. Also, since the data set of nonmagnetic transition metal is sufficiently large, we can extract meaningful statistical averages of deviations (denoted 'errors' in reference [95] and in the panel) for this important class of systems with a dense electron distribution. In the vdW-DF-cx case, we provide this hard-matter benchmark not only for the normal PAW set up of VASP [217, 218] but also (as noted by a subscript) with the 'hard' PAW setup to test convergence of the E_c^{nl} evaluation [18]. Also, for the study of cohesive energies, we relied on the QUANTUM ESPRESSO implementation of spin vdW-DF-cx [11] to describe the atom reference energy, finding a systematic improvement [95].

The bottom panel figure 5 suggests that vdW-DF-cx is highly accurate on structure and vibrations and therefore on the electron-density description in dense matter. The transitionmetal benchmarking result supplements other findings of good vdW-DF-cx performance on structure (and vibrations). Besides low-temperature comparisons with oligoacene and polyethylene measurements [98, 99, 103, 107], we mention that the vdW-DF-cx is found accurate on describing polarization of PbTiO₃ and vdW-bound layered materials [9, 45, 68, 113, 121, 230].

4.2. Charge and current conservation in vdW-DF

We believe that the vdW-DF-cx robustness advantages follow from the emphasis on conservation laws in the full vdW-DF method. We explicitly enforce current conservation in the response descriptions, equation (34), and this also ensures an automatic compliance with charge conservation for the vdW-DF XC hole [10].

Adapting equation (20), the full vdW-DF XC hole is

$$n_{\rm xc}^{\rm DF}(\mathbf{r};\mathbf{r}'-\mathbf{r}) = -\delta(\mathbf{r}-\mathbf{r}') - \frac{1}{2\pi n(\mathbf{r})} \int_0^\infty \frac{\mathrm{d}u}{2\pi} \nabla_{\mathbf{r}}^2 \langle \mathbf{r} | \ln(\kappa_{\rm long}(iu)) | \mathbf{r}' \rangle.$$
(96)

The Fourier transform of this XC hole is

$$n_{\rm xc}^{\rm DF}(\mathbf{r};\mathbf{q}') = \int_{\mathbf{u}} e^{i\mathbf{q}'\cdot\mathbf{u}} n_{\rm xc}^{\rm DF}(\mathbf{r};\mathbf{u}=\mathbf{r}'-\mathbf{r}). \tag{97}$$

Overall XC hole conservation, equation (4), mandates that $n_{xc}^{DF}(\mathbf{r}; \mathbf{q}' \to 0) = -1$. This is equivalent to demanding $\Delta n_{xc}^{DF}(\mathbf{r}; \mathbf{q}' \to 0) = 0$ where Δn_{xc}^{DF} is used to identify all XC hole contributions that originate from the ln(κ_{long}) part of the vdW-DF specification.

For our discussion it is convenient to explore the formal connections to a characterization of light–matter interactions [47]. Because we use a scalar representation of the model susceptibility, the current-conservation criterion, equation (32), reduces to

$$\tilde{\chi}_{\rm ACF}(\omega)V = \nabla\alpha(\omega) \cdot \nabla V. \tag{98}$$

However, the external-field susceptibility, given by equation (31), remains a tensor given by the analysis behind equation (28). To make the identification we simply expand α_{ext} in terms of $\tilde{\chi}$ (and thus α), sorting the contributions according to the number *m* of times that $\tilde{\chi}$ enters in the associated Dyson equation (30):

$$\alpha_{\text{ext}} = \sum_{m=1}^{\infty} \alpha_m. \tag{99}$$

We note that

$$\chi_{\rm ACF} V = \sum_{m=1}^{\infty} \left(\nabla \boldsymbol{\alpha} \cdot \nabla V \right)^m, \tag{100}$$

and it follows that $\alpha_1 = \alpha \mathbf{I}$ while

$$\boldsymbol{\alpha}_{m \ge 2} = \alpha (\nabla V) (\nabla \alpha \cdot \nabla V)^{m-2} (\nabla \alpha)$$
(101)

are matrix contributions defined by the outer product of the first ∇V and the last $\nabla \alpha$.

The evaluation of the vdW-DF method relies on expanding every instance of $\alpha(iu)$ (entering in equation (99)), in terms of S_{xc} to some order given j_k , using equation (75). The implied effects on the optical-response description can be sorted

$$\boldsymbol{\alpha}_{\text{ext}} = \sum_{n=1}^{\infty} \sum_{m \ge n}^{\infty} \sum_{i} \boldsymbol{\alpha}_{m,n}^{(i)}, \qquad (102)$$

according to both *m* and to the total number $n = j_1 + j_2 + \cdots + j_m$ of S_{xc} factors. An index *i* is used to keep track of the different ways that a multiple expansion complies with the *m* and *n*'s. Those details are not written out here as they have no relevance for the present discussion.

Each of the susceptibility terms of equation (102) corresponds to the XC hole contribution

$$\Delta n_{\mathrm{xc},m,n}^{\mathrm{DF},(i)} = \nabla_{\mathbf{r}'}^2 \int_{\mathbf{r}''} \nabla_{\mathbf{r}} \cdot \mathbf{A}_{m,n}^{(i)}(\mathbf{r},\mathbf{r}'') \cdot \nabla_{\mathbf{r}''} V(\mathbf{r}''-\mathbf{r}'), \quad (103)$$

where

$$\mathbf{A}_{m,n}^{(i)}(\mathbf{r},\mathbf{r}'') = \frac{1}{2\pi n(\mathbf{r})m} \int_0^\infty \frac{\mathrm{d}u}{2\pi} \,\boldsymbol{\alpha}_{m,n}^{(i)}(\mathbf{r},\mathbf{r}'';\mathrm{i}u). \tag{104}$$

Evaluating the Laplacian and performing a partial integration yields

$$\Delta n_{\mathrm{xc},m,n}^{\mathrm{DF},(i)} = 4\pi \nabla_{\mathbf{r}} \cdot \nabla_{\mathbf{r}'} \mathbf{A}_m(\mathbf{r},\mathbf{r}'), \qquad (105)$$

where

$$\nabla_{\mathbf{r}} \cdot \nabla_{\mathbf{r}'} \equiv \left\{ \frac{\partial}{\partial x} \frac{\partial}{\partial x'} + \frac{\partial}{\partial y} \frac{\partial}{\partial y'} + \frac{\partial}{\partial z} \frac{\partial}{\partial z'} \right\}.$$
 (106)

The longitudinal projection leading to the form equation (105) is thus sufficient to ensure explicit charge conservation for each partial XC-hole contribution, $\Delta n_{\text{xc},m,n}^{\text{DF},(i)}(\mathbf{r}; \mathbf{q}' \to 0) = 0.$

4.3. vdW-DF versions and variants

The original general-geometry vdW-DF1 version [4] did not fully rely on the logic of the vdW-DF method (as summarized above); instead it relied on equations (2) and (3), that is, using a formulation in which the semilocal functional component E_{xc}^0 is not firmly linked to the internal functional E_{xc}^{in} . In effect, there is a presence of a cross-over component $\delta E_x^0 = E_{xc}^0 - E_{xc}^{in}$. The same is, in principle, true for all other present general-geometry vdW-DF versions and variants [8, 9, 16, 40–43]. It is also true for the precursor, the layered geometry vdW-DF0 version [3, 6].

The reason for originally tolerating a cross-over term δE_x^0 is a build-in frustration in the present vdW-DF designs. On the one hand, the exchange content of E_{xc}^{in} must help curb nonlocal correlation contributions from low-density regions [1, 6, 9, 10, 12, 65, 171]. This motivated picking an LV form [7, 91, 174] for the exchange enhancement factor $F_x(s) = 1 + \mu_{LV}s^2$ in the internal functional. On the other hand, the LV exchange [91] is not, in itself, a good exchange description and leads to poor descriptions of, for example, atomization energies of molecules.

In the original layered- and general-geometry vdW-DF versions [3, 4], we picked revPBE as the functional exchange choice, entering in E_{xc}^0 . The idea was simply to minimize potential for double counting [6] and the argument was given by demonstrating that this original exchange choice for vdW-DF essentially eliminates all binding from exchange in the case of layered materials and noble gas atoms [3, 4, 6].

The emphasis on plasmon consistency in (spin) vdW-DFcx means that the actual vdW-DF-cx exchange (in E_x^0) is also given by the LV analysis [7, 91, 174], at scaled gradients s < 2-3. At small *s*, this LV behavior is similar to the PBEsol gradient expansion [29]. However vdW-DF-cx (PBEsol) builds from the LV handling of a screened-exchange (from a pureexchange) analysis of the $\mathbf{q} \rightarrow 0$ response limit [7, 174]. Since the PBEsol exchange enhancement (at small *s*) is in turn similar to that used in vdW-DF-C09 [41] and vdW-DF-optB86b [42], these variants are thus related to the consistent-exchange vdW-DF-cx [9].

The exchange choice in other vdW-DF variants and in vdW-DF2 [8] is picked from other considerations.

The vdW-DF family of XC functionals comes with a highly effective Roman–Soler algorithm [17] for evaluating E_c^{nl} . It also comes with an open-source library, termed LIB-VDWXC, for massively-parallel computations of this scheme [18]. While based on (parallel) fast-Fourier transforms, the

scheme (and the library) can easily be adapted to all-electron calculations on radial grids [231]. Generally, the E_c^{nl} evaluation has excellent scaling [18] and it is never a speed hindrance compared to, for example, a GGA in a plane-wave code, for a given choice of the wavefunction energy cut off.

A full, first-principle nonlocal-DFT characterization of truly large organic and even biological systems is today possible. The biophysics potential is illustrated by a recent subsystem and linear-scaling but first-principle DFT study of structure and molecular dynamics in a tobacco mosaic virus in an explicit aqueous solution [232].

The virus study used the related rVV10 nonlocalcorrelation functional [66, 67] as implemented in CP2K [233] and a modern TIER0 supercomputer [232]. However, in computational terms, rVV10 and vdW-DF differ essentially by looking up different universal-kernel files [4, 7, 67]; in QUAN-TUM ESPRESSO [36, 219], the rVV10 implementation is, in part, adapted from the subroutines for vdW-DF calculations [11, 36]. The rVV10 and vdW-DF share the Roman–Soler [17] evaluation scheme which seems to present no computational bottleneck up to at least a million atoms [232].

4.4. Rationale for consistent-exchange vdW-DF-cx

A simple argument makes it clear that the Dyson equation, equation (30), uniquely specifies the correct balance between vdW-DF exchange and correlation terms. The exchange exclusively resides in $\tilde{\chi}_{ACF}$ (as it must be independent of λ) while the rest of the Dyson-equation terms can exclusively contribute to the nonlocal-correlation term. However, as the ACF effectively reflects an actual long-range interaction, V_{ACF} , the Dyson equation (and Lindhard screening logic) leaves no wiggle room between local- and external-field response components [62, 63].

The recent consistent-exchange vdW-DF-cx version [9, 45] seeks to restore compliance with this electrodynamics guide for the vdW-DF method. Effectively, this means that we must use E_{xc}^{in} to specify the exchange component. A complete elimination of the cross-over term δE_x^0 is not possible in present vdW-DF versions, given by equation (3). However, the exchange component of vdW-DF-cx is chosen to effectively eliminate the adverse effects of $\delta E_x^0 \neq 0$ in the description of binding.

Part of the motivation for taking the consistent-exchange vdW-DF-cx path is a concern about conservation laws. The effect of using a traditional or direct GGA-type specification for the semilocal functional component $E_{xc}^0 = E_{xc}^{in} + \delta E_x^0$ can be seen as shifting to a direct-design approach, $S_{xc} \rightarrow S_{xc}'$, exclusively in the leading equation (94). This corresponds to an adjusted vdW-DF method framework:

$$E_{\rm xc}^{\rm DF'} \equiv \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr} \left\{ \ln \left(\kappa_{\rm long}(iu) + \delta S(iu) \right) \right\},$$

$$= E_{\rm xc}^{\rm vdW-DF}$$

$$+ \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr} \left\{ \ln \left(1 + \kappa_{\rm long}^{-1}(iu) \delta S(iu) \right) \right\}, \quad (107)$$

where $\delta S \equiv S_{xc}' - \nabla S \cdot \nabla G$. It is not clear when the second term of equation (107) complies with the current-conservation logic of the vdW-DF method [9]. Also, if we assume some other approach to enforce charge conservation in E_{xc}^{0} , it is not clear how that input can easily be merged with the conservation constraints in the present vdW-DF versions. The Occam razor approach of consistent vdW-DF implementations is to try to avoid such potential complications.

Over and above that, we recommend the vdW-DF-cx for general-purpose use because it aims to be true to the underlying MBPT logic. It does not break the Dyson/Lindhard logic of screening, and avoids an uncontrolled approximation [63]. Also, it exclusively relies on analysis of quantum Monte Carlo (QMC) data (in the LDA correlation [23] part of E_{xc}^{in}) and of MBPT (in the screened-exchange [7, 91, 174] part of E_{xc}^{in}). Of course, the vdW-DF-cx alignment with the full vdW-DF method idea is only perfect when the materials binding arises in regions with small to moderate values of the scaled gradients s < 2-3, as discussed elsewhere [9, 10]. That criterion is, however, expected to hold in bulk problems and for covalent binding in molecules. It often holds also for intermolecular binding cases [10].

Finally, sticking to the vdW-DF-cx logic holds another advantage: since all functional components are set directly in terms of S_{xc} , it makes it easier to seek systematic improvements in nonlocal-correlation functionals [9, 10, 12].

4.5. Spin vdW-DF-cx: a systematic extension

Spin effects enter the vdW-DF description because spin polarization adjusts the plasmon dispersion [22], and hence the plasmon propagation [11]. The vdW-DF propagator form S_{xc} is set by a semilocal functional E_{xc}^{in} and it is thus entirely given by LDA correlation and by gradient-corrected exchange, appendix B. Spin effects on the latter are exactly specified by an exact spin-scaling result [27]. Spin effects on the former are believed to be well described by the analysis in references [23, 24]. In the vdW-DF method, with the present choice of plasmon-propagator model, there is only one possible specification of spin effects on S_{xc} .

The implication is that there exists a proper spin extension of the vdW-DF method, reference [11]. The extension is uniquely defined for any given choice of the internal functional if we strictly follow the Occam design strategy: excluding gradient-corrected correlation in $E_{\rm xc}^{\rm in}$ and sticking with the simple, gradient-expanded $S_{\rm xc}$, appendix B and references [4, 12]. The natural spin extension (of a given vdW-DF version) results by simply continuing to use $S_{\rm xc}$ to determine all terms of the full vdW-DF method specification, equation (91). While the standard vdW-DF versions, given by equation (3), also contain an exchange cross-over term, $\delta E_{\rm x}^0$, the spin extention is still uniquely determined since the exact spin-scaling criterion [27] also defines spin-polarization effects on $\delta E_{\rm x}^0$.

Figure 6 summarizes a study of the weak chemisorption of graphene. We use a 6 layer unit-cell slab and a lateral unit-cell choice that is indicated by the white frame in the top panel. It was chosen to also permit a comparison with RPA calculations [235, 236]. The formulation of spin vdW-DF [11] permitted us to complete the bottom-panel comparison of adsorption of



Figure 6. Weak chemisorption of graphene on Ni (top panel) as described using proper spin extensions of the vdW-DF method (bottom panel). Blue and gray spheres identify Ni and C atoms, respectively. The vertical dashed line in the bottom panel shows the experimental results for the adsorption minimum distance 2.11 ± 0.07 Å [234]. The panel contrasts the results for the adsorption energy variation as obtained in the spin extension of the original vdW-DF1 (black circles), of vdW-DF2 (purple diamonds), and of vdW-DF-cx (green squares). For completeness the panels also shows the results for PBE and VV10 [66, 67]. The panels are reproduced, with permission, from the supplementary materials of reference [11]. Copyright (2015) by the American Physical Society.

graphene on Ni(111). Here we focus on a comparison with an experimental observation of the optimal adsorption distance [234] (indicated by the vertical dashed line).

Contrasting the graphene binding-energy curves we find that vdW-DF-cx differs from vdW-DF1 [4], from vdW-DF2 [8], from rVV10 [66, 67], and from PBE [28] in predicting both a physisorption minima and a weak-chemisorption minima at a shorter graphene-surface separation. The RPA results yield no clear preference for a chemisorption configuration at a 2.2 Å binding separation [235, 236]. Meanwhile, vdW-DF-cx correctly predicts the stability of the weak chemisorption minima, occuring at a graphene-surface separation 2.1 Å that aligns with the measured Ni(111) value 2.11 ± 0.07 Å, [234].

In molecular problems [35], and especially for atomization energies [13, 15, 237–242], a correct handling of spin effects is important. The same was also found to be true for transition-metal cohesive energies in cases where the atom is in a high-spin state [95]. It is necessary to eventually seek a non-collinear spin implementation of vdW-DF-cx [44] (and of other vdW-DF versions and variants) to address some triangular (or other frustrated) molecular problems [243] as well as for some so-called vdW crystals [44]. However, such a formulation is not yet publicly available.

For a general treatment of magnetic effects and molecular spin it is also possible that the vdW-DF ability to reflect spin effects may be improved by permitting more flexible formulations, that merge the vdW-DF-cx logic with the spin handling that is used in PBE [244, 245]. However, we are presently sticking with the above-summarized Occam design strategy for vdW-DF-cx [11]. This is because simplicity in the design focus (using just E_{xc}^{in} as the nucleation point in spin vdW-DF-cx) makes it easier to check the impact of fundamental design choices, for example, regarding the plasmon dispersion [9, 45].

4.6. Electrodynamics interpretation of consistent vdW-DF designs

Below, we discuss the formal nature of general consistent vdW-DF designs. That is, we consider the class of vdW-DF versions that comply with the screening laws of the ACF and of the vdW-DF framework, and that aim to implement the full logic of the vdW-DF method for XC functional designs.

In the consistent-exchange vdW-DF designs, the leading term must effectively be set by E_{xc}^{in} . It is thus restricted to GGA exchange and LDA correlation. The nonlocal-correlation energy of the vdW-DF method can formally be written

$$E_{\rm c}^{\rm nl} = E_{\rm c,\alpha}^{\rm nl} + E_{\rm c,\,vdW}^{\rm nl}.$$
 (108)

The splitting is here made depending on whether the nonlocalcorrelation effects originate from $\tilde{\chi}_{ACF}$ or from the $m \ge 2$ components of the Dyson expansion, equation (99). Taken together, the terms of equation (108) must serve as the consistent-vdW-DF replacement for the GGA formulation of gradient-corrected correlation. Appendices C and D discuss the details in a general such GGA extension, here we summarize the analysis in an electrodynamics context.

Electrodynamics-response terms with m = 1 define one important subset of contributions

$$E_{\mathrm{xc},\alpha} \equiv \int_0^\infty \frac{\mathrm{d}u}{2\pi} \operatorname{Tr}\{4\pi\alpha(\mathrm{i}u)\}$$
$$= E_{\mathrm{xc}}^{\mathrm{in}} + E_{\mathrm{c},\alpha}^{\mathrm{nl}}.$$
(109)

These are naturally ordered by the number n of S_{xc} factors:

$$E_{\mathrm{c},\alpha}^{\mathrm{nl}} = \sum_{n=2}^{\infty} E_{\mathrm{c},\alpha}^{n},\tag{110}$$

$$E_{c,\alpha}^{n} = (n!)^{-1} \int_{0}^{\infty} \frac{du}{2\pi} \operatorname{Tr}\{\nabla(S_{xc})^{n} \cdot \nabla G\}$$
$$= (n!)^{-1} \int_{0}^{\infty} \frac{du}{2\pi} \operatorname{Tr}\{(S_{xc})^{n}\}.$$
(111)

Here (as in appendix C) we retain the notation introduced in equation (102), using *n* to track the number of S_{xc} factors.

The m = n = 1 component is just the internal functional, as implied in equation (109).

The exponential-re-summation term of E_{xc}^{DF} , that is, equation (110), is set up to track general screening, including vertex corrections, when used together with the leading E_{xc}^{in} term. This follows by the discussion in section 3.2 and appendix B. However, in the context of describing vdW interactions, 'screening' is often taken to imply a moderation of the dispersion forces [81, 202]. To avoid confusion, we therefore choose to denote equation (110) instead as a 'cumulant' term, in our discussions below.

We note that the cumulant term captures vertex corrections even if the expansion in equation (111) is truncated at second order, at n = 2. The n = 1 term, i.e., the internal functional $E_{\rm xc}^{\rm in}$, is designed to carry the vertex corrections that are in LDA correlation directly [4, 7]. When truncating the expansion of $E_{\rm c,\alpha}^{\rm nl}$ at n = 2, as done in popular versions defined by equation (94), we cannot fully leverage the exponential resummation that underpins the vdW-DF method. However, we retain a re-summation character since we always keep at least two expansion terms in $E_{\rm c,\alpha}^{\rm nl}$.

The remaining nonlocal-correlation part

$$E_{\rm c, vdW}^{\rm nl} = -\sum_{m=2}^{\infty} \int_0^\infty \frac{\mathrm{d}u}{2\pi m} \operatorname{Tr}\{(\nabla \alpha \cdot \nabla V)^m\},\qquad(112)$$

$$E_{c, vdW}^{nl} \equiv \sum_{n=2}^{\infty} \sum_{m \ge n} \sum_{i} E_{xc,m,n}^{DF,(i)},$$
(113)

primarily reflects pure or asymptotic vdW effects. The association is directly motivated for the second-order expansion form

$$E_{c, vdW}^{nl, m=n=2} = -\int \frac{du}{4\pi} \operatorname{Tr}\{\nabla \cdot \boldsymbol{\alpha}_{1}(iu) \cdot \nabla V \nabla \cdot \boldsymbol{\alpha}_{1}(iu) \cdot \nabla V\}$$
(114)

This is always describing an attraction, appendix D, and it reflects the form for the vdW interaction that holds for two disjunct fragments, section 4 and reference [10].

More generally, the full specification equation (112) contains factors that can always be rewritten

$$\frac{1}{m} (\nabla \alpha \cdot \nabla V)^m = \frac{1}{m(m-1)} \sum_{\nu=1}^{m-1} (\nabla \alpha \cdot \nabla V)^{m-\nu} (\nabla \alpha \cdot \nabla V)^{\nu}.$$
(115)

For partly separated fragments, the interaction is still dominated by terms where the implied repeated spatial integrations will only involve two Coulomb factors $V(\mathbf{r} - \mathbf{r}')$ connecting these fragments. Apart from the weighting in equation (115), we have again terms that reflect the asymptotic vdW form, section 2.

We need the full vdW-DF machinery, given by the interplay of the vdW term equation (112) and of the cumulant term equation (111), in general. The full account sorts out the weighting between nonlocal-correlation effects in the more interesting general cases, when density fragments can no longer be considered disjunct.

Interpretation of terms in the HEG limit merits a separate discussion. We note that, as we approach the HEG, we will

no longer have system fragments and the original premise of interpreting equation (112) as pure vdW interactions eventually breaks down; similarly, we should explain the nature of equation (110) in the HEG limit [4].

In the HEG, there cannot be any actual vdW forces nor can there be any beyond-LDA vertex corrections. A vdW-DF-cx calculation is fully consistent with those observations: we explicitly comply with the criteria $E_c^{nl} \equiv 0$ and $E_{xc}^{in} \rightarrow E_{xc}^{LDA}$ in the HEG limit [4, 12]. The question is only one of *interpretation*.

We find it motivated to always view equation (112) as reflecting pure vdW interactions, i.e., forces arising from the Ashcroft-ZPE coupling mechanism [1, 10, 64, 65, 158]. This picture, figure 2, is valid in the HEG limit (even if it has no consequences on material binding then). Yes, these ZPE electron-correlation effects are already baked in into the LDA description of the HEG [25]. However, it is still instructive to view the relevance of the pure vdW interactions in isolation, as an evaluation of equation (112) allows us to do.

We view equation (110) as representing cumulant (i.e., vertex and other GGA-type screening) effects in general, again even in the HEG limit. Since equation (112) represents the Ashcroft-ZPE mechanism, there is also a need (even in the HEG) to explicitly treat a compensating effect, namely as provided by the cumulant term. This term must therefore be extracted from the LDA correlation that we have originally inserted in E_{xc}^{in} . Equation (110) specifies that subtraction exactly, ensuring that the vdW-DF method has seamless integration with LDA in the HEG limit [4].

Electrodynamics interpretation of terms in the standard, truncated-expansion vdW-DF versions. The leading (m = n = 2) components of equation (108) define the physics content of the (spin) vdW-DF-cx version [9, 11]. It is worth exploring in detail.

Appendix D shows that those terms can be formulated in terms of a universal kernel evaluation

$$E_{c,\alpha}^{nl,n=2} = \frac{1}{2} \int d^3 \mathbf{r} \int d^3 \mathbf{r}' n(\mathbf{r}) \phi^{\alpha}(d,d') n(\mathbf{r}'), \quad (116)$$
$$E_{c,vdW}^{nl,n=2} = \frac{1}{2} \int d^3 \mathbf{r} \int d^3 \mathbf{r}' n(\mathbf{r}) \phi^{vdW}(d,d') n(\mathbf{r}'), \quad (117)$$

where d and d' denote suitable ways to scale the distance between **r** and **r'**.

Figure 7 summarizes our numerical evaluation of the (universal-)kernel components, ϕ_{α} and ϕ_{vdW} . The variation is given in terms of scaled differences, as detailed in appendix D and elsewhere [4, 5, 60]. The sum

$$\phi(d, d') = \phi^{\alpha}(d, d') + \phi^{\text{vdW}}(d, d'), \quad (118)$$

defines the full E_c^{nl} kernel, and both of these kernels reflect the general two-density-point structure of equation (55). However, the universal-kernel forms, $\Phi_0^{\alpha}(d, d')$ and $\Phi_0^{vdW}(d, d')$, are found to always remain positive and negative, respectively.

In the vdW-DF-cx version, we have an expansion given by equations (41) and (109). The beyond- E_{xc}^{in} are terms given by



Figure 7. Universal nonlocal-correlation kernels for the full E_c^{nl} behavior (black curves) and for the vdW and cumulant components (red and blue curves respectively). The kernels reflect nonlocal-energy contributions from an electrodynamics coupling among GGA-type XC holes, centered at two different positions **r** and **r'**, reference [10]. The kernel behavior can, however, be fully represented by an effective distance *D* and a parameter δ which reflects how different these internal GGA-type XC holes are at positions **r** and **r'**. It is exclusively the set of $\delta = 0$ curves that are relevant for the homogeneous electron gas (HEG) limit; seamless integration with LDA follows in the HEG because the solid-black curve integrates to zero [4, 5].

equations (108), (116) and (117) and we make the following observations:

The first nonlocal-correlation part, equation (116), is, in itself, of direct interest for understanding the physics content of our electrodynamics modeling. This follows because it can be combined with $E_{\rm xc}^{\rm in}$ to yield a mapping of the model susceptibility $\alpha(\omega)$. The associated local energy contribution

$$e_{\mathbf{c},\alpha}^{\mathrm{nl}}(\mathbf{r}) = n(\mathbf{r}) \int_{\mathbf{r}'} \phi^{\alpha}(d(\mathbf{r},\mathbf{r}'),d'(\mathbf{r},\mathbf{r}')) n(\mathbf{r}'), \qquad (119)$$

provides a mapping of where we can expect cumulant effects (including nonlocal vertex corrections), in equation (109), to play a larger role in material binding. Being based on an exponential re-summation, it has a formal structure that allows a low-level approximation to be accurate on the quasi-particle dynamics, at least for core levels and near the Fermi surface [86, 163, 164]. We are motivated to extract this mapping also broadly, since we are documenting a strong general

performance for vdW-DF-cx, references [9–11, 15, 45, 95] and section 6.

The second nonlocal part of equation (108) represents pure vdW interactions [1, 10, 64, 65, 158], as they emerge in the presence of screening by the surrounding electron gas [10, 12, 64]. The associated local energy contribution

$$e_{\mathrm{c,vdW}}^{\mathrm{nl}}(\mathbf{r}) = n(\mathbf{r}) \int_{\mathbf{r}'} \phi^{\mathrm{vdW}}(d(\mathbf{r},\mathbf{r}'),d'(\mathbf{r},\mathbf{r}')) n(\mathbf{r}'), \qquad (120)$$

provides a mapping of where pure vdW binding contributes significantly to materials binding.

Seamless integration requires treating the cumulant and the pure-vdW terms together. There are in general compensations between the $e_{c,vdW}^{nl}$ and $e_{c,vdW}^{nl}$ contributions, while $e_{c,vdW}^{nl}$ dominates at larger separations. However, both parts are needed to secure seamless integration of vdW-DF-cx with LDA, i.e., compliance with equation (55) in the near-HEG limit. Inserting a constant density, n_0 , yields $E_{c,\alpha}^{nl} < 0$. One needs the



Figure 8. Nonlocal correlation contributions to binding in bilayer graphene. The left panel shows the spatial variation in the total nonlocal correlations computed in vdW-DF-cx. The middle and right panels depict the cumulant (or vertex-correction) and the pure vdW interaction components, respectively.

combined nonlocal-correlation effects, reflected in equations (95) and (118), to ensure seamless integration, as discussed elsewhere [4, 12].

The universal-kernel descriptions simplify an analysis of the nature of binding. Figure 8 illustrates how having efficient determinations of nonlocal-correlation contributions equations (119) and (120) can provide a spatial mapping of the nature of binding, here explored for the case of a graphene bilayer. The left panel shows a mapping of the full E_c^{nl} binding contribution; such mapping was introduced and discussed in references [55, 57, 58] and can be supplemented by a couplingconstant scaling analysis that isolates the kinetic-correlation energy binding component [14].

The middle and right panels of figure 8 shows a hereexplored separation of E_c^{nl} binding contributions into cumulant effects (nonlocal-vertex corrections) and pure vdW attraction, respectively. We see that the vdW attraction is larger and even more widely distributed in the trough region *between* the layers than what is apparent from the E_c^{nl} description [9, 10, 14, 59]. However, this pure vdW binding is also offset by a repulsion in $E_{c,a}^{nl}$, in the central region.

Figure 9 summarizes a vdW-DF-cx calculation and our pure-vdW/cumulant-resolved analysis of the binding between two carbyne wires. That is, we consider the attraction between two parallel one-dimensional wires of pure carbon as a function of their separation d, as illustrated in the top left panel.

This is a system where vdW-DF-cx has problems at describing the mutual binding energy (that is, the energy difference between the combined system and those of the components) per unit length at large wire distances. Depending of the intra-wire carbon-carbon separation l_{C-C} , this problem allows a simple tuning between an insulating nature (with dimerization) and a conducting nature (with no dimerization) within a tight-binding DFT model, by simply varying the intra-wire carbon-carbon separation [246]. This leads in turn to an elegant analysis of the impact of 1D conduction on the long-range component of the vdW attraction [105, 202, 246-254]. Reference [246] provides the analysis within a TS-based many-body-dispersion (MBD) method [80, 81], denoted TS-MBD. It tracks the power-law behavior of the mutual vdW interaction for variations of the wire separations d. The TS-MBD method is found to correctly produce a d^{-5} asymptotic scaling for insulating wires and an expected significant difference in the exponent for metallic wires [202, 247, 250].



Figure 9. Scaling in the vdW-DF-cx results for binding between a pair of carbyne wires. The left panel shows the geometry, identifying also a focus on tracking the mutual attraction per unit length as a function of the changes in wire distance $d = d_0 + \Delta$, where $d_0 = 3.81$ Å is the vdW-DF-cx results for the equilibrium wire separation. The middle right panel shows the computed variation in the total wire binding energy (per carbon atom in a wire), in the full nonlocal-correlation binding contribution (black curve) and in the binding contributions from pure vdW interactions (red curve) and cumulant corrections (blue curve). The top right and bottom panels show the corresponding scaling of an interpretation of the nonlocal-correlation binding in terms of an atom-based-vdW-pair interaction (with effective atom-centered-vdW C₆ coefficients) and in terms of a wire-based-vdW interaction form (with effective wire-vdW B₅ coefficients), respectively. The vertical line in the bottom panel positions the optimal separation d_0 relative to the $B_5(d)$ enhancements.

For vdW-DF method implementations that rely on the universal-kernel formulation (as vdW-DF1 and vdW-DF-cx do,) the interaction enhancement can stretch out to about a nanometer due to multipole or image effects. This is documented, for example, in a previous study of parallel semiconducting nanotubes [53]. The enhancement in vdW-DF-cx attraction cannot stretch as far as in TS-MBD because the kernel contribution from an electron–XC-hole pair, at \mathbf{r}_1 and \mathbf{r}_2 , *eventually* approaches a form decaying like $|\mathbf{r}_1 - \mathbf{r}_2|^{-6}$, references [4, 105]. The consequence is that vdW-DF-cx and vdW-DF always have an asymptotic vdW interaction energy that decays like d^{-5} for parallel wires. The vdW-DF-cx predicts an absence of dimerization and a conducting nature of the wire, but it fails to predict the associated enhancement and

non-integer power-law interaction behavior that should therefore result at large separations [246, 250].

However, the nature of vdW attraction at and near binding must necessarily be markedly different from the asymptotic form [105]. The vdW interaction among nanostructures is indeed also documented [53] to be different than what is in the asymptotics [10, 69, 105, 246, 253, 254]. We and others have therefore argued that even non-MBD XC functionals can still be useful and accurate for characterizations on the mutual attraction at binding separations [10, 69, 105].

The middle-right panel of figure 9 reports our vdW-DF-cx calculation of the mutual binding (per carbon atom in a wire) described by differences in the total energy (in green, with $E_{\rm cx}$ label) and in the nonlocal-correlation energy (in black, with $E_{\rm c}^{\rm nl}$ label). We have furthermore resolved the latter in pure-vdW (red) and cumulant (blue) contributions. The results are obtained for the fully relaxed intra-wire carbon–carbon separation $l_{\rm C-C} = 1.28$ Å. The interaction results are shown as a function of $\Delta d = d - d_0$, where $d_0 = 3.81$ Å denotes the equilibrium vdW-DF-cx value for the wire separation. Computational details are given in section 5.

The bottom panel of figure 9 also shows that pure-vdW term dominates the total nonlocal-correlation term as the distance increases, but the repulsive $E_{c,\alpha}^{nl}$ -based contributions significantly compensate this attraction at and near binding separations. There is also a saturation directly in $E_{c,vdW}^{nl}$ (not shown in this panel). The sum of these nonlocal-correlation contributions scales away in the high-density region, to provide a seamless integration with LDA [4, 5, 7, 10]. The sum of these contributions eventually vanishes when the distance *d* decreases.

Differences between the pure vdW attraction $(E_{c, vdW}^{nl})$ and the binding arising from the sum of nonlocal correlations (in E_c^{nl}) are certainly the correct behavior in weak-chemisorption binding [59] where competition among forces are responsible for producing density changes [12, 45, 255, 256]. It is also, at least qualitatively, the correct behavior for pure physisorption [175, 176, 181, 182] since both nonlocal-correlation components are needed to reflect the vdW-multipole [10] as well as other screening effects [10, 105]. The screening produces, for example, large image-plane corrections in vdW binding at surfaces [157, 179, 182, 184, 185] and among extended molecules [53]. The image-plane effects enhance the interaction description at close-to-intermediate separations from the asymptotic d^{-5} interaction behavior [53].

We shall return to a more detailed discussion of this interaction. Here we note that in vdW-DF-cx the description at and near materials binding separations is guided by the Lindhard/Dyson screening logic and have a formal reason for being robust toward system-specific changes in the electron-density variation [45, 56, 59].

4.7. Open questions for consistent vdW-DF

The presentation of the vdW-DF method and of consistentexchange vdW-DF versions (like vdW-DF-cx and svdW-DFcx) leave us at this stage with a set of open questions: *GGA-level vertex corrections?* Consistent vdW-DF relies on an exponential re-summation to capture effects similar to nonlocal vertex corrections. However, it is not clear if the resummation in $\alpha(\omega)$ is enough, if it can, in practice, benefit from the vertex advantages that exist for cumulant expansion for the quasi-particle dynamics [86, 163, 164]. The constraintbased semilocal functionals, like PBE and PBEsol, set a high standard. Transition metals and their surfaces are often considered a challenge for XC functionals [25, 56, 89, 95, 133, 181, 222, 223, 225, 226, 257]. A particular questions is then: does the indirect design logic of consistent vdW-DF retain enough semilocal correlation to work even here?

Non-additivity in vdW interactions? In the generalgeometry vdW-DF formulation [4, 12], we truncate the expansion of nonlocal-correlation effects in the plasmon-propagator S_{xc} to second order. This is not enough to capture so-called many-body dispersive interactions in the asymptotic limit [10, 64, 81, 105, 202, 252]. However, two questions remain: (1) does the reliance of the vdW-DF method on a GGA-type screening [10] provide non-additivity at binding separations? (2) Since not all fragments of a molecular crystal reside at the optimal mutual separation, do we get the right description of molecular crystals at binding?

All-round performance? We want nonlocal-correlation XC functionals that work for concurrent descriptions of both dense and sparse matter components. In consistent vdW-DF, we leverage both Dyson/Lindhard screening logic and current conservation. An important question is therefore: does the general-purpose capability [45] survive in the resulting vdW-DF-cx version?

Role nonlocal vertex effects? The GGAs and meta-GGAs are expected to reflect a mixture of nonlocal vertex effects and short-to-immediate-range vdW interactions [69], but offer no detailed characterization of the balance. This review explains that vdW-DF-cx calculations can be used to track such screening effects in isolation. It is therefore natural to inquire, where do we find important cumulant (vertex-correction) effects in general materials binding?

5. Computational details

Some of the open questions for consistent vdW-DF can be settled by analyzing published results, for example, vdW-DF-cx studies for molecules as well as for bulk and layered matter. For computational details of those studies we refer directly to the cited papers.

Other questions for consistent vdW-DF can be addressed by validation studies that we also provide in this paper. The results of some of these test have already been partly presented in figures 1, 5, 6, 8, and 9. However, the discussion will generally be expanded in the following validation section and we have collected the computational details in this section.

For a discussion of molecular and bulk binding processes (like binding and reactions) we generally investigate a compound system 'AB' as well as its relevant constitutes, for example, 'A' and 'B'. We compute in vdW-DF-cx, or in a general functional 'DF', both the total energy (in this general discussion denoted $E_{\rm DF}$) and XC functional terms, like $E_{\rm cl}^{\rm al}$.

These are sometimes computed with full atomic relaxations and sometimes at so-called reference geometries, when given in benchmark suites like the GMTKN55 [35]. The energy values of interest are the energy differences for the investigated process 'p', like

$$E_{\rm p} \equiv \Delta E_{\rm DF} = E_{\rm DF}^{\rm A} + E_{\rm DF}^{\rm B} - E_{\rm DF}^{\rm AB}, \qquad (121)$$

$$E_{\rm p,\,c}^{\rm nl} \equiv \Delta E_{\rm c}^{\rm nl} = E_{\rm c}^{\rm nl,A} + E_{\rm c}^{\rm nl,B} - E_{\rm c}^{\rm nl,AB}.$$
 (122)

The process energies may at times reflect a binding or a bulk cohesion, and they are then denoted E_b or E_{coh} to have better consistency with papers and surveys that are cited. Similar structures of energy differences, equations (121) and (122), hold when the properties 'p' of interest are isomerization energies and molecular unfolding energies, reaction barriers, or charge transfers (including ionization or electron-affinity processes).

We systematically use the plane-wave QUANTUM ESPRESSO DFT code package [36, 219]. Like our c-based library LIBVDWXC [18], this code suite has highly efficient, native implementations of both the fast-Fourier-transform (universal-kernel) evaluation [17] of the E_c^{nl} . It has the vdW-DF family of functions, including the consistent-exchange vdW-DF-cx version [9], and it has the proper vdW-DF spin extension [11]. It also permits stress relaxations of nonlocal-correlation functionals. Core electrons are represented via pseudopotentials since we use QUANTUM ESPRESSO [36, 219].

5.1. Molecular-system benchmarking

To obtain statistics on functional performance, we complete a full functional survey of molecular properties as tested in the extensive GMTKN55 suite of benchmarks sets [35]. The GMTKN55 not only has a wide range of benchmark sets for intra- and inter-molecular noncovalent-interaction energies, as already discussed in figure 1. However, the GMTKN55 suite also has groups of benchmark sets probing the performance for small- and large-molecular properties, and a group of benchmarks probing barriers for molecular reactions [35]. Taken together, the results on performance give a fair impression of the robustness and transferability of semi- and truly nonlocal density functionals, when used in the molecular realm.

We perform and report this comprehensive survey for both the consistent-exchange vdW-DF-cx version [9] and for the original vdW-DF1 [4, 5, 7].

For the molecular benchmarking we use Troullier–Martins norm-conserving pseudo potentials of the ABINIT package [38], with a 80 Ry wavefunction cutoff. We and collaborators have previously used this pseudopotential and cut off choice for benchmarking performance on molecular interactions, in connection with launching hybrid vdW-DF(-cx) versions [13, 15] and while testing the dependence on unit-cell sizes for plane-wave studies of molecular systems [37].

We use the gamma point *k*-point for our periodic unitcell representation of the molecular problems. Within each of the 55 different molecular benchmark sets, we identified the largest systems and then added 10 Å of vacuum to have a standard unit-cell size for all comparisons (among the individual processes and among functionals). This approach yields a setup that for the S22 benchmark set corresponds to the convergence recommendations made in reference [37]. Several benchmark sets involve comparing system of different charge states. For these we used the Makov–Payne correction [39] to compensate for spurious electrostatic interactions

molecules. The relevant comparison of (spin) vdW-DF-cx accuracy should be made on the so-called second (GGA-based) rung of dispersion-corrected DFTs [33]. This is because vdW-DF-cx, while a consistent implementation of the vdW-DF method, is still limited to a semi-local choice of the exchange formulation [59]. According to the full GMTKN55 comparison [35] of dispersion-corrected DFTs [33], one of two best performing second-rung DFT-D3 formulations is revPBE-D3 [33, 34]. This functional is also available to us (with the very same pesudopotentials) in QUANTUM ESPRESSO [36, 219]. It is an interesting comparison for revPBE-D3 has the very same exchange choice as do both vdW-DF1 [4, 7] and vdW-DF0 [3]. We therefore include a full QUANTUM ESPRESSO [36, 219] survey of the revPBE-D3 as well as of PBE [28].

that arise because we use a periodic unit-cell representation for

We note that a comparison of vdW-DF1, revPBE-D3 and vdW-DF-cx allows us to highlight the role of balancing exchange and correlation contributions, as already discussed in the introduction in connection with figure 1.

We have not made our own comparison of vdW-DF-cx molecular performance versus the best-performing version on the third (meta-GGA) rung of the DFT-D3s, namely SCAN-D3 [33, 35, 258]. Expanding the survey is beyond the present scope and it would distract from our present focus. Here the purpose is to explore if use of Lindhard/Dyson screening logic to balance exchange and correlation significantly enhances the robustness and transferability of XC functional designs in the vdW-DF method. A comparison with SCAN-D3 [33, 35, 258] and SCAN + rVV10 [69] is relevant for a possible, future meta-vdW-DF, a version that may perhaps use a meta-GGA exchange as the internal functional.

We do, however, find that the vdW-DF-cx performance (as we assert in QUANTUM ESPRESSO) generally approaches literature reports for SCAN + rVV10 [69] performance, for here-investigated systems. This holds for SCAN-D3 studies for molecules, computed in TURBOMOLE [259, 260] and reported in the GMTKN55 functional survey [35]. It also holds for SCAN + rVV10 studies of surface properties [223] computed in VASP [217, 218]. We mention also these SCAN-D3/+rVV10 results in a vdW-DF-cx comparison when relevant.

At the same time, we note that care must be taken in comparing our plane-wave/pseudopotential benchmarks with those of quantum-chemistry DFT codes (that are typically used for the surveys in referecne [35]). There are advantages and disadvantages of both approaches [37]. However, it is clear that converging negatively charged systems in a plane-wave code is generally difficult, at least with the presently used pseudopotential choice. Furthermore, our QUANTUM ESPRESSO testing is complicated by the fact that we do not have access to a non-collinear spin formulation [44] for vdW-DF-cx.

To provide a check on our molecular GMTKN55 survey we therefore first discuss differences and similarities between our revPBE-D3 benchmarking in QUANTUM ESPRESSO and that reported in reference [35]. We find that sometimes our plane-wave revPBE-D3 results are closer to reference values (available in reference [35]) and sometimes the quantumchemistry results [35] are closer. However, in almost all cases the agreement is good in terms of percentages and in terms of absolute MAD values.

For all of the benchmark sets involving noncovalent interactions, the agreement between the here-reported revPBE-D3 benchmarking and that of reference [35] is typically within one tenth of a kcal mol⁻¹ and within two tenths of a kcal mol⁻¹ for benchmarks containing charged systems, marked with an asterisk in figure 1. The alignment is also strong with regards to RMSD values and in terms of percentage errors. This robustness is important for our analysis: outliers indicate that something is not covered by the logic and our aim is to draw general conclusions about the logic in consistent vdW-DF versions [9].

The results for the WATER27 benchmark set of the GMTKN55 [35] are, however, an exception with over one kcal mol⁻¹ difference on a total 3.5 kcal mol⁻¹ MAD value reported in reference [35]. We expect that we can improve our QUANTUM ESPRESSO benchmarking for water with a better pseudopotential choice, and that may hold for plane-wave benchmarking in both revPBE-D3 and vdW-DF-cx. The net result on the performance statistics of any one such benchmark is very low.

The GMTKN55 suite also has other benchmarks. There are group 1, 2, and 3 of benchmarks for small-molecule properties, for large-molecule properties, and for transition barriers, respectively [35]. Here the differences between our plane-wave benchmarking and the quantum-chemistry approach of reference [35] are larger for revPBE-D3, but not out of proportion to the overall deviations from the reference energies that exist here. The differences in MAD values for the revPBE-D3 benchmarking are generally well within 1 kcal mol⁻¹. The differences are, however, about 2 kcal mol⁻¹ for the MB16 mindless-benchmarking set and for the DC13 benchmark set of difficult-for-DFT tests [35].

Interestingly, we find that we are 2 kcal mol⁻¹ closer to reference values for the W4-11 set of atomizations energies than what is reported for revPBE-D3 in reference [35]. However, for vdW-DF-cx we find a MAD value of almost 11 kcal mol⁻¹, one of the few benchmarks where vdW-DF-cx fares substantially worse than revPBE-D3. Other benchmarks were vdW-DF-cx generally fares worse than revPBE-D3 are in the GMTKN55 group 3. This will be analyzed in a separate paper.

We observe that bad plane-wave performance (of vdW-DF-cx/revPBE-D3) does result in some charged-system benchmarking problems. It also results when QUANTUM ESPRESSO finds a spin configuration that is frustrated, for example, in triangular molecules like ClO_2 and even in the

linear CCH system. Some molecules are known to require a non-collinear treatment of spin [44, 243], but we lack access for vdW-DF and vdW-DF-cx. In our molecular benchmarking we do make sure that we provide the right value of the total molecular spin M_{tot} , as given in the reference data [35]. That is, denoting the spin-up(-down) density components as $n_{\uparrow(\downarrow)}(\mathbf{r})$, we enforce

$$M_{\text{tot}} = \int \mathrm{d} \mathbf{r} [n_{\uparrow}(\mathbf{r}) - n_{\downarrow}(\mathbf{r})], \qquad (123)$$

while also converging the orbital occupation. However, in some molecular problems we find non-integer values of the so-called absolute magnetic moment

$$M_{\rm abs} = \int \,\mathrm{d}\,\mathbf{r} |n_{\uparrow}(\mathbf{r}) - n_{\downarrow}(\mathbf{r})|. \tag{124}$$

A non-integer value of M_{abs} arises, in our quantum-espresso characterization of vdW-DF-cx performance, for ClO₂ and CCH (part of the W4-11 atomization benchmark set), for C₄H₄ and aC₇H₈ case (in the RC21 benchmark set), for C₃H₂N, C₃H₅, C₂H₂N, C₃H₃, and C₇H₇ (in the RSE43 benchmark set), as well as in aC₃H₇ transition state. A non-integer M_{abs} value also arises in the charged CO⁺, CS⁺, NH⁻, O⁻, and O₂⁻ states, affecting our plane-wave assessment of performance for the electron-affinity (G21EA) and electron-ionization (G21IP) benchmark sets, themselves part of group 1 of the GMTKN55 benchmark suit [35].

Nevertheless, we report our QUANTUM ESPRESSO survey of vdW-DF-cx performance not only for all the noncovalent-interaction benchmarks, figure 1, but across all of the GMTKN55 groups, below. We do that noting that the present GMTKN55-type [35, 267] survey reflects what is presently available to a user that aims to pursue (plane-wave) vdW-DF-cx calculations.

Also, since we aim to check the overall robustness, accuracy and the general-purpose capability, we are in any case interested in statistically averaged values. We worry about outliers that we do not understand but we can accept specific issues. This is so because the percentage differences between MAD values in the plane-wave [36, 219] versus quantum-chemistry [35] benchmarking remain small in essentially all cases, the W4-11 atomization energies, MB13 and DC13 being exceptions. It is nice to see that there is perhaps a correlation between performance hits and known limitations in the calculation platform that is available to us. However, we live with the fact that we could have put a different, and perhaps even a better, foot forward had vdW-DF-cx already been implemented in codes that focus on descriptions of molecules.

5.2. Nature of interactions in low-dimensional systems

All of the following studies are provided using the GBRV ultrasoft pseudopotentials [268] with a 50 Ry wavefunction cutoff.

For our study of the binding in the graphene bilayer, figure 8, we first relaxed the atomic coordinates using a slab setup and a $20 \times 20 \times 1k$ -point sampling. The slab height is 50 Å to ensure amble vacuum and to eliminate coupling between periodic images. From the fully relaxed electron density

variation we subsequently extract the nonlocal-correlation contributions, relying on our PPACF code [14, 15]. This code is designed to split the total system energy $E_{vdW-DF-cx}$ into the range of XC-energy (and other) components, using also a coupling-constant scaling of vdW-DF-cx [14]. In anticipation of this benchmarking and analysis effort, we designed PPACF with an ability to also read in new universal-kernel components, for which appendix D provides details of the evaluation.

Our PPACF code is released into the QUANTUM ESPRESSO post-processing tool set [36, 219].

The study of the nature of binding among carbyne wires, figure 9, is performed with vdW-DF-cx for a set of unit cells of dimensions $l_{C-C} \times 50$ Å $\times 50$ Å using an $80 \times 1 \times 1k$ -point sampling. The value of l_{C-C} is varied for convergence and we find no dimerization for vdW-DF-cx at any pressure. Consistent-exchange vdW-DF-cx describes the carbyne wires as metallic.

For a study of the nature and non-additivity of interactions in the fullerene dimer & trimer, we use Gamma-point calculations. For a similar characterization of the nanotube-trimer bundle we use a *k*-point mesh grids of size $16 \times 1 \times 1$. Again we use PPACF to extract the spatial variation in the nonlocalcorrelation energies for each specific system. We then compute the relevant binding-energy differences with the QUANTUM ESPRESSO post-processing tool set [36, 219] following the procedure introduced in reference [14].

5.3. Benchmarking bulk and surface properties

Our surface-energy results are extracted by computing and comparing the energies of slabs with thickness varying from 4 to 12 layers, extracting an average as in reference [223]. The work functions were computed from slabs of 8 layers by tracking the electrostatic potential outside the surfaces. All slab systems were first allowed full relaxations as described by the XC functional in question.

For tests of bulk-system properties we sample the Brillouin zone using gamma-centered k-mesh grids of size $16 \times 16 \times$ 16, whereas for surface-slab systems we use $16 \times 16 \times 1$. We use the set of GBRV ultrasoft pseudopotentials [268] with a 50 Ry wavefunction cutoff.

For the metal and MgO slab geometries we add 15 Å of vacuum to compensate for possible Coulombic interaction between the actual surface and its periodic image. We also use a dipole corrections [269] to compensate for polar effects. The 15 Å vacuum region allows us to track the electrostatic potential and thus extract work-function values [181, 222, 223, 225].

We compute and extract the surface-energy and work function values for vdW-DF-cx, PBE, and PBEsol in QUANTUM ESPRESSO. Some SIESTA- [270] and VASP-based [217, 218] vdW-DF-cx results for metal-surface work functions and surface energies have already appeared [224–226]. However, here we provide a broader vdW-DF-cx (and PBE/PBEsol) survey. This gives us data to report a performance characterization based on the metal-surface benchmarking that is implicitly suggested in reference [223]. We find essentially identical results using the plane-wave code QUANTUM ESPRESSO [36, 219] for the PBE and PBEsol studies that we repeated here (checking for, but finding no code dependences). We include also these calculations to highlight that we provide a consistent comparison in our assessment of the vdW-DF-cx performance.

5.4. Understanding binding

To understand how vdW-DF-cx works for dense (and sparse) matter, we compare vdW-DF-cx and PBE descriptions of XCenergy contributions to binding in bulk. We also provide this comparison for two fullerene molecules, while we refer to reference [14] for a corresponding comparison and analysis for smaller molecules.

We base this comparison and analysis on calculations of the energy-per-particle variation [14, 55, 57, 58]. The standard energy-per-particle representations of semilocal XC functionals is given by equation (61), while the generalization equation (62) applies to all investigated functionals, including our truly nonlocal vdW-DF-cx.

The general energy-per-particle energy variation $\epsilon_{xc}^0[n](\mathbf{r})$ can be further split into exchange and correlation components. The spatial variation in the LDA/PBE correlation energy densities is simply $e_c^{\text{LDA/PBE}}(\mathbf{r}) = n(\mathbf{r})\epsilon_c[n](\mathbf{r})$. For vdW-DF-cx it is natural to look at the nonlocal-correlation energy density [14, 55, 57, 58]:

$$e_{\rm c}^{\rm nl}(\mathbf{r}) = \frac{1}{2} \int d\mathbf{r}' \, n(\mathbf{r}) \, \phi(\mathbf{r}, \mathbf{r}') \, n(\mathbf{r}')$$
$$= e_{\rm c,\alpha}^{\rm nl}(\mathbf{r}) + e_{\rm c, vdW}^{\rm nl}(\mathbf{r}).$$
(125)

This vdW-DF-cx [9] characterization should be compared with the nonlocal, that is gradient-corrected, PBE correlation

$$e_{\rm c}^{\rm PBE,nl}(\mathbf{r}) \equiv e_{\rm c}^{\rm PBE}(\mathbf{r}) - e_{\rm c}^{\rm LDA}(\mathbf{r}).$$
 (126)

It is also natural to track the spatial variation in the total nonlocal parts of the vdW-DF-cx and PBE XC energies, exploring differences in the nonlocal XC functional components

$$e_{\rm xc}^{\rm cx,\,nl}(\mathbf{r}) \equiv e_{\rm xc}^{\rm cx}(\mathbf{r}) - e_{\rm xc}^{\rm LDA}(\mathbf{r}),\tag{127}$$

$$e_{\mathrm{xc}}^{\mathrm{PBE,nl}}(\mathbf{r}) \equiv e_{\mathrm{xc}}^{\mathrm{PBE}}(\mathbf{r}) - e_{\mathrm{xc}}^{\mathrm{LDA}}(\mathbf{r}).$$
 (128)

The relevant quantity for our discussion is the spatial variation in the binding contributions, for example,

$$\Delta e_{\rm c}^{\rm nl}(\mathbf{r}) = e_{\rm c}^{\rm nl,A}(\mathbf{r}) + e_{\rm c}^{\rm nl,B}(\mathbf{r}) - e_{\rm c}^{\rm nl,AB}(\mathbf{r}).$$
(129)

The integrals of these spatially resolved contributions yield binding-energy contributions like equation (122).

As we use the vdW-DF-cx version, we can also track the binding contributions arising from the (nonlocal-correlation part of) vertex corrections,

$$\Delta e_{c,\alpha}^{nl}(\mathbf{r}) = e_{c,\alpha}^{nl,A}(\mathbf{r}) + e_{c,\alpha}^{nl,B}(\mathbf{r}) - e_{c,\alpha}^{nl,AB}(\mathbf{r}).$$
(130)

A similar definition also allows us to track binding contributions arising from pure vdW interactions, i.e., given by $e_{c,vdW}^{nl}(\mathbf{r})$.

This set of analysis tools is available through our PPACF code [14, 15] and through the post-processing components of QUANTUM ESPRESSO [36, 219]. We used these to map and contrast the nature binding of PBE and vdW-DF-cx for bulk Si and in metals as well as for a fullerene dimer.

For an analysis of the nature of binding in bulk, we first re-compute the electron density variation using Troullier-Martins norm-conserving pseudo potentials of the ABINIT package [38] with a 80 Ry wavefunction cutoff; for W bulk we also increase the k-point mesh to $24 \times 24 \times 24$. Furthermore, W is a heavy element with large gradients in the electron distribution around the individual atoms and we compute the electron-density variation of the atoms using an 800 Ry density cutoff. We apply a cubic-spline fit to the spatial variations of the XC energy components that we extract for each of the atoms. The high-accuracy characterizations of both bulk and atoms are finally used to determine binding contributions, using equations (129) and (130) or using corresponding expressions for LDA and PBE characterizations. This set of steps reduces the numerical noise in the resulting mapping of vdW and other screening effects in bulk-crystal cohesion, below.

6. Validation checks on vdW-DF-cx

The vdW-DF-cx compliance with the underlying screening logic of the ACF and with current and charge conservation is promising. We can expect that the consistent vdW-DF versions will be robust and likely transferable. However, usefulness as a general-purpose functional will only follow if vdW-DF-cx (and by extension consistent vdW-DF versions) continue to perform well in tests. Below we summarize some of those tests and we extend the testing to a broad set of molecular benchmarkings [35] and to metal surfaces [223].

As the vdW-DF-cx passes validation checks, we also propose to use the vdW-DF-cx Lindhard-screening logic to map interaction details, as illustrated below.

6.1. All-round performance of consistent vdW-DF

There are by now many studies for bulk, layered systems, and organics that document that vdW-DF-cx is robust and transferable, for example, references [68, 95, 98, 99, 103, 107, 113]. Below we provide a brief overview of the vdW-DF-cx performance. We limit ourselves to a summary because we also want to provide specific validation checks on vdW-DF-cx ability to capture semilocal-correlation that plays an important role in the description of metal surfaces [29, 223]. Moreover, we want to illustrate the unique advantage that vdW-DF-cx calculations have in terms of mapping cumulant (screening and vertex-correction) effects in material binding.

We begin with a discussion of the asymptotic vdW interactions among atoms and molecules. The original vdW-DF1 [4] (and thus vdW-DF-cx [9]) nonlocal-correlation term leads to a good description of asymptotic C₆ vdW interaction coefficients for many atoms [9, 271]. The same is true for some molecular problems [210, 211], but not for hollow structures [105, 252], where screening causes dramatic changes in the lowest-energy collective mode [252]. A similar effect exists for metallic systems (including graphene sheets and conducting nanotubes) where the extended collective response causes fundamental changes in the nature of the interaction, again at asymptotic separations [202, 246, 251]. However, the asymptotic description is not the focus of the present implementa-

tions of the vdW-DF method [10]. Next we return to the broad survey of the vdW-DF-cx performance for molecular interactions, using the GMTKN55 suite of 55 benchmark sets on molecular properties [35]. The reference data for structure (including the molecular spin state) and energies (as well as per-functional/per-benchmark performance results for DFT-D3 versions) are available [267] and allow for an effective way of testing (spin) vdW-DF-cx relative to dispersion-corrected DFT. For individual benchmark sets, like G21IP, G2RC, S22, Al2X6, DARC, and Al2X6 of the GMTKN55 collection, we have previously compared the vdW-DF-cx performance relative to that of hybrids, references [13, 15]. In those studies we checked if the good vdW-DF-cx performance arose from a lucky cancellation by also tracking the effects of relaxations (computed in vdW-DF-cx). Here instead, we safeguard the testing by going for the full benchmark suite [35].

To accomplish this we have written python scripts to download the data and submit QUANTUM ESPRESSO calculations for the molecular problems, automatically implementing the computational strategy described in section 5. There are in total 6 GMTKN55 benchmark groups. Group 1 probes smallmolecule properties (half of which involve charged systems), group 2 probes large-molecule properties, group 3 concerns barriers for molecular reactions, while group 4 and group 5 test intra- and inter-molecular noncovalent interactions, respectively. Group 6 is just the union of group 4 and 5. For each of the GMTKN55 benchmark groups, the scripts also compute the performance in the individual sets and extract values for the MAD, RMSD, and the mean absolute percentage deviation (MAPD) relative to the CCSD(T) reference data on process energies [35, 267]. We extract a weighted MAD average [35]

$$WTMAD \equiv \frac{1}{55} \sum_{i=1}^{55} w_i \times MAD_i.$$
(131)

The summation runs over the 55 different benchmark sets that make up the GMTKN55, with MAD_i denoting the individualbenchmark MAD value. The weighting factor w_i are set to align benchmark with similar values of the reference energy values for the process [35]. The weighting is such that having 1 kcal mol⁻¹ deviation on the weaker noncovalent interactions can cause as much as 100 times the WTMAD change that arises in benchmarks with large characteristic process energies. We furthermore extract and report a similar WTMAD value for each of the GMTKN55 benchmark groups [35].

In figure 1 we have already reported details of this benchmarking for all of the group-6 benchmarks on noncovalent interactions in molecular systems. There we show the perbenchmark set MAD and RMSD values and contrast the vdW-DF-cx, vdW-DF, and revPBE-D3 performance. The vdW-DFcx outperforms the original vdW-DF version and performs



Figure 10. Full survey of vdW-DF-cx performance across the GMTKN55 suite [35] of benchmarks sets, as asserted in the plane-wave code QUANTUM ESPRESSO [36, 219] and using a weighted total mean absolute deviation measure 'WTMAD' defined in reference [35] and in the text. For comparison, the figure also includes corresponding results for PBE and revPBE-D3 (also in QUANTUM ESPRESSO calculations) and those obtained with TURBOMOLE [259, 260] for SCAN-D3 in reference [35]. Benchmark groups 4, 5 correspond to a set of benchmarks on interand intra-molecular noncovalent interactions. Group 6 is the union of group 4 and 5, that is, the set of general noncovalent interaction benchmarks that was detailed in figure 1. Group 1, 2, and 3 are collections of benchmark sets reflecting small-molecule properties, large-molecule properties, and transition barriers, respectively [35].

slightly better than revPBE-D3 on average. As discussed in the introduction, this fact is an argument for balancing the exchange and correlation terms using the vdW-DF screening logic. It is a motivation for using (spin) vdW-DF-cx.

Figure 10 reports our full GMTKN55 survey, contrasting WTMAD values obtained from a full QUANTUM ESPRESSO characterization in PBE, revPBE-D3, and vdW-DF-cx. We find that vdW-DF-cx performs significantly better than PBE, and at least matches revPBE-D3 performance on all but the group 3 class of barrier problems, when compared on our plane-wave/pseudopotential platform. The comparison with revPBE-D3 is relevant because it is one of the two bestperforming DFT-D3s that have a GGA-level exchange. We also note that on average vdW-DF-cx performs almost as good as SCAN-D3 [33, 258] even if vdW-DF-cx lacks a meta-GGA formulation of exchange.

As discussed in section 5, care should be taken in making the vdW-DF-cx comparison with these literature values for SCAN-D3, obtained in the orbital-based code, TURBOMOLE [259, 260]. We are then comparing with results obtained in a different, non-plane-wave DFT framework. Some care must also be made when making comparison between our vdW-DF-cx and revPBE-D3 characterizations for group 3, for group 2, and, perhaps especially, for group 1. This is because there are some small residual differences between what we obtain as the revPBE-D3 performance characteristics and what was reported in reference [35]. The reason could in part be our representation of the molecular spin, as discussed in section 5. On the other hand, figure 10 does provide a present-day planewave (QUANTUM ESPRESSO) user with an impression of what performance level is already available. We find that vdW-DF-cx is a realistic option even for a pure molecular problem.

We also find (not shown) a small but finite performance advantage of vdW-DF2-b86r [43] over vdW-DF-cx [9] for the broad molecular benchmarking that the GMTKN55 suite offers [35, 267]. The vdW-DF2-b86r (or rev-vdW-DF2) is a variant that Hamada launched to make vdW-DF2 applicable also for solids and general materials. It also has a GGAlevel exchange and it is also highly successful. Completing a vdW-DF2-b86r survey analogous to figure 10, we provide the following vdW-DF-cx/vdW-DF2-b86r contrast of WTMAD performances (listed in kcal mol⁻¹): 5.15/5.18 (group 1), 4.93/4.81 (group 2), 7.25/6.68 (group 3), 3.14/3.22 (group 4), 3.86/3.61 (group 5), 3.47/3.39 (group 6), and 4.74/4.62 (total GMTKN55).

We interpret this vdW-DF2-b86r advantage as a sign that the class of consistent vdW-DF versions has room to grow also beyond vdW-DF-cx. This is true even before going to a possible, future meta-vdW-DF formulation or a hybrid vdW-DF [13, 15]. We recall that for many problems the materials properties are often set by the low-to-medium values of the scaled density gradient *s*. In such cases, the vdW-DF-cx exchange description effectively resembles that of the Langreth–Vosko analysis [91], which is not ideal for molecules [92]. The vdW-DF-cx remains a relevant choice because it stands out in being consistent, in having a regular screening logic for balancing exchange and correlation. However, we can hope to eventually obtain a consistent vdW-DF design that has both a good exchange (as vdW-DF2-b86r has) and a good XC balance.

Table 1 summarizes performance comparisons in a broader context for the description of noncovalent interactions both within and between molecules. The table reflects systematic studies reported in references [9, 13, 15, 98, 124, 266]. It includes a more detailed listing of the GMTKN55 comparison concerning both the S22 and the IDISP [35], and adds an overview of a benchmarking for the L7, A24, and so-called blind test cases [262–264]. Overall, the vdW-DF-cx is found to compete well with the class of dispersion-corrected DFT descriptions [33, 76, 78, 80, 81].

For the G1-set of molecular atomization energies [237, 238] the vdW-DF-cx version does perform worse than PBE (giving a MAD value of 9.7 kcal mol⁻¹ as opposed to the PBE 7.5 kcal mol⁻¹, for fully relaxed structures) [13]. We ascribe this issue, in part, to the fact that the vdW-DF-cx elimination of the vdW-DF cross-over term δE_x^0 is not perfect for atomic density profiles [9, 10]. We also note that the performance picture for the G1 set is reversed once we move to corresponding hybrid formulations, PBE0 and vdW-DF-cx0/-cx0p [13, 15].

Table 1 furthermore provides both direct and indirect evidence that vdW-DF-cx is accurate in its characterization of both the inter- and intra-molecular structure. Direct evidence is available from the excellent agreement between vdW-DFcx results for lattice parameters (and to a lesser extent, for **Table 1.** Comparison of PBE [28], PBE0 [261], PBE-XDM [77, 78], PBE-D3 [33], PBE-TS [80], PBE-TS-SCS [81], and vdW-DF-cx [9] (here abbreviated CX) performance for dispersion-dominated inter- and intra-molecular interactions. The table summarizes performance that is characterized both in terms of MAD and (when available) the mean absolute percentage deviation (MAPD) values. The comparisons of functional performance are obtained, mostly in plane-wave DFT, both at reference geometries (as denoted with a '-refG' hyphen) and, when available, with relaxations native to the functional (as denoted with a '-minG' hyphen). We include MAD and MAPD listings for the so-called new-S22 and intramolecular dispersion (IDISP) subsets of GMTKN55 [35], as well as for the L7 [262], A24 [263], and the so-called 'blind set' [264]. We have also included a comparison defined by theory accuracy in matching low-temperature measurements for structure and cohesion in oligoacene crystals [98]. Cohesion-energy measurements for the oligoacenes are limited to the benzene, naphthalene, and anthracene crystals; we report MAD values that are averaged over these. Structure characterizations, extracted in the low-temperature limit, are available for oligoacenes up to hexacenes [98]; here we report MAD and MAPD values that are obtained by averaging deviations between computed results and measurements of the *a*, *b*, and *c* lattice constants [98]. In the benchmarking summary for the IDISP set at relaxed coordinates, we have omitted the $C_{22}H_{46}$ case, as a more plausible atomic configuration were obtained with full vdW-DF-cx relaxations. This alkane-unfolding problem deserves a separate discussion, see text.

Test Set	Measure	PBE	PBE0	PBE-XDM	PBE-D3	PBE-TS	PBE-TS-SCS	СХ
S22-refG E _b	MAD (kcal mol^{-1})	2.55 ^a /2.72 ^c	2.37 ^a /2.54 ^c	0.59 ^b	0.48 ^a /0.54 ^b	0.34 ^b	_	0.47 ^b
	MAPD (%)	57°	55°	9.8 ^b	11.6 ^b	10.0 ^b /9.2 ^d	5.4 ^d	9.0 ^b
S22-minG $E_{\rm b}$	MAD (kcal mol^{-1})		_	0.54 ^b	0.57 ^b	0.49 ^b	—	0.68 ^b
	MAPD (%)		_	8.4 ^b	11.1 ^b	12.8 ^b	_	9.4 ^b
S22-minG d	MAD (Å)		_	0.075 ^b	0.073 ^b	0.040 ^b	_	0.068 ^b
IDISP-refG Eb	MAD (kcal mol^{-1})	10.78 ^a	9.51 ^a /9.8 ^b		2.76 ^a			2.4 ^b
	MAPD (%)		242 ^b				_	32 ^b
IDISP-minG E _b	MAD (kcal mol^{-1})		_				_	2.7 ^b
	MAPD (%)		_				_	30 ^b
L7-refG E _b	MAD (kcal mol^{-1})	18.6 ^e	_			2.92 ^{e,f}	3.79 ^{e,f}	1.58 ^{e,f}
	MAPD (%)	120 ^e	_			17 ^e	21 ^e	14 ^e
A24-refG E _b	MAD (kcal mol^{-1})	0.40 ^e				0.35 ^e	0.29 ^e	0.14 ^e
	MAPD (%)	42 ^e				28 ^e	22 ^e	11 ^e
'Blind set' $E_{\rm b}$	MAD (kcal mol^{-1})	3.00 ^e	_			0.65 ^e	0.79 ^e	0.54 ^e
	MAPD (%)	52 ^e	_			11 ^e	13 ^e	10 ^e
Acenes-minG $E_{\rm b}$	MAD (kcal mol^{-1})	15.2 ^g	_			5.1 ^{g,h}	_	2.3 ^g
Acenes-minG $\langle a, b, c \rangle$	MAD (Å)	0.76 ^g	_			0.09 ^{g,h}	_	0.06 ^g
	MAPD (%)	9 ^g	_	—	—	1 ^{g,h}	—	1 ^g

^aReference [35].

^bReference [15].

^cExtracted from the computational data that we also used for reference [15].

^dReference [81].

^eReference [124].

^fWith DLPNO-CCSD(T) reference data [265], these calculations [124] yield a 1.39/2.25/1.51 MAD for TS/TS-SCS/CX.

^{g.}Reference [98].

h.Reference [266].

the fully relaxed binding energies) for the set of oligoacene molecular crystals [98]. The accuracy also extends to the characterization of the electron-density variation, as evident by the vdW-DF-cx ability to predict the soft intermolecular libration and rocking modes in the naphthalene [99], figure 5, and in the polyethylene [107] crystals.

Indirect evidence for vdW-DF-cx accuracy is also evident in table 1 by looking at the effects of structural relaxations on the vdW-DF-cx performance on molecular binding/reaction energies. These results, at minimized geometries, are listed by the 'minG' entries (where available). For the tests on noncovalent and covalent binding properties (in the AL2X6, DARC, G2RC, G21IP subsets of GMTKN55 [35],) we find no significant changes in performance from that asserted at reference geometries (listed under 'refG'). There are small effects (improvements) on the binding-energy (and binding-separation) accuracy for the soft S22 [15].

One of these benchmarks, namely the IDISP subset of the GMTKN55 [35], deserves a special discussion concerning the

vdW-DF-cx structural characterization. Interestingly, there are no discernible relaxations for all but the $C_{22}H_{46}$ case of alkane unfolding in the IDISP set [15]. The consistent-exchange vdW-DF-cx version performs well also for describing the $C_{22}H_{46}$ unfolding energy at the reference geometry [15, 35]. However, at the reference geometry, the CCSD(T) result suggests an exothermal unfolding process, in contrast to the results of a fully relaxed vdW-DF-cx study [15]. Since an energy release for unfolding is in contrast with expectations from past investigations of alkane interactions [278], we have excluded the $C_{22}H_{46}$ case in our summary of relaxed-coordinate performances for the IDISP set in table 1.

The soundness and robustness of vdW-DF-cx for molecular relaxations can be further tested by analyzing the $C_{22}H_{46}$ unfolding case in detail, by instead comparing CCSD(T) and vdW-DF-cx energy results for fully relaxed vdW-DF-cx structures. As will be detailed elsewhere, this is done using the GMTKN55 procedure, described in reference [35], for the CCSD(T) calculations. The new CCSD(T) results reflect the expected endothermal behavior of $C_{22}H_{46}$ unfolding. In fact, at the vdW-DF-cx structure, the CCSD(T) result, 1.08 kcal mol⁻¹, for the unfolding energy cost aligns perfectly with the self-consistent, fully relaxed, vdW-DF-cx description, 1.06 kcal mol⁻¹.

A recent study provides a comparison of performance for extended system for many of nonlocal-correlation functionals [230], based on all-electron evaluations [231]. That study includes the full vdW-DF family of versions and variants as well as SCAN + rVV10 [69] and rVV10 [66, 67]. They find that vdW-DF-cx [9] and vdW-DF-C09 [41] comes out best statistically in the vdW-DF family for strongly bound bulk problems (close in performance to SCAN-rVV10), vdW-DF2-b86r [43] and vdW-DF-optb86r [42] are just ahead of vdW-DF-cx for layered solids, while vdW-DF2-b86r [42] is ahead of vdW-DF-cx for investigations of the set of molecular solids. The last point is consistent with the observation above that vdW-DF2b86r has a small but finite performance advantage over vdW-DF-cx for molecules, and that it may be possible to further develop the class of consistent vdW-DF versions.

6.2. Vertex corrections and surface properties

The real challenge for a nonlocal-correlation functional of the vdW-DF method, however, lies in the description of materials that have dense electron distributions. This follows because in these problems it is essential to describe subtle materials interactions as they occur in competition with one another [45]. For such systems we must balance both (a) gradient-corrected exchange effects against truly nonlocal correlation effects and (b) local as well as nonlocal vertex corrections in the response description.

Fortunately, the constraint-based PBE [28] and PBEsol [29] designs are believed (and proven) to be highly robust for materials with a dense electron distribution [30, 31]. As such, they open the door for another, computationally-efficient validation check on the robustness of vdW-DF-cx (and by extension, of the consistent vdW-DF formulations). These test is only available for dense-matter systems, like transition-metal systems [95]. However, for such traditional systems, it is clear that vdW-DF-cx must keep up with the PBE/PBEsol performance to be trusted as a systematic extension. Most noticeable, we must check if vdW-DF-cx is as good for dense matter as PBE/PBEsol in avoiding the outliers in performance.

We mention that the meta-GGA SCAN functional [258] is also constraint-based and typically performs even better than PBE and PBEsol for dense matter, while the SCAN + rVV10 [69] fares well in comparison of broad extendedsystem performance [230]. A comparison of vdW-DF-cx with the SCAN/SCAN + rVV10 is certainly interesting. It is indirectly available in the following reporting but it is not highlighted in the discussion. This is because such a comparison is not a true assessment of soundness of present implementations of the consistent vdW-DF design logic: we repeat that with (spin) vdW-DF-cx we are starting with a GGA-level exchange description and have just enforced consequences of conservation laws and Lindhard screening in an electrodynamics formulation of the ACF. We seek comparisons with constraintbased GGA descriptions that have a GGA-level of exchange to test this idea of building from just one type of plasmons [9]. In summary, we do not here discuss a comparison of vdW-DF-cx and SCAN/SCAN + rVV10, but we are motivated to benchmark the vdW-DF-cx performance against experimental data while checking whether it is as good as PBE/PBEsol at avoiding truly bad characterizations.

In reference [95] one of us helped begin such validation work. The results are partly summarized in the lower panel of figure 5. This survey of thermo-physical properties has already been briefly discussed in sections 4.1 and 4.4 in the context of the accuracy of phonons, figure 5, and of the proper spin vdW-DF-cx extension. The thermo-physical properties study contrasts PBE, PBEsol, and vdW-DF-cx performance for the set of nonmagnetic transition metals. As mentioned above, the comparison stands out by treating vibrational ZPE and thermal effects through phonon calculations that are native to each of the semilocal and nonlocal functionals. The vdW-DF-cx is found to be highly accurate in comparison with experimental data. Moreover, it is found to have at least the same level of limited variation in the individual-system accuracy as do PBE and PBEsol [95].

We have already discussed spin vdW-DF-cx results for weak-chemisortion of graphene on Ni(111), figure 6. There, vdW-DF-cx provides results for the optimal adsorption distance that are closer to experimental observations [234] than what is obtained in RPA [235, 236]. Reference [224] provides an additional comparison of vdW-DF-cx and of rVV10corrected semilocal functionals, including PBEsol + rVV10 [72], with the corresponding RPA results for graphene adsorption on the (111) surfaces of Cu, Pt, Pd, and Ag [235, 236]. The vdW-DF-cx descriptions of the adsorption height are generally found in agreement with the RPA results for these surfaces [224]. The vdW-DF-cx results for the adsorption energy per carbon atom also follows the RPA trend, but the values for vdW-DF-cx (and for the set of investigated rVV10-corrected functionals) are found systematically smaller than the RPA results. Interesting, we have documented elsewhere that vdW-DF-cx, if anything, tends to overestimate the binding energy between two graphene sheets, as well as among other carbon nanostructures [12, 105].

The case of weak chemisorption on Ag(111) merits a separate discussion, since the RPA results [235, 236, 279] for the adsorption height do not line up with recent measurements [280]. Reference [224] reports an adsorption height for vdW-DF-cx of 3.26 Å; adapting our reference [11] study, we find here a vdW-DF-cx value of 3.24 Å for graphene on Ag(111). These vdW-DF-cx results are in fair agreement with the RPA value [235, 236], 3.31 Å, but not with the value 2.5 Å reported in reference [280].

Table 2 reports a comparison between the PBE, PBEsol, and vdW-DF-cx results for structure, cohesion, and elastic response in bulk in a set of ionic crystals, insulators, and alkali metals. The table also lists experimental results, back-corrected for ZPE and temperature effects [29, 272], to help us assert the accuracy and robustness of the functionals. This

Table 2. Comparison of PBE [28], PBE0 [261], PBEsol [29], and vdW-DF-cx [9] performance (lattice constants, bulk moduli, and atomization energies) for ionic crystals and a few alkali metals. The reference values, denoted 'Exp.', are zero-point corrected experimental values, references [29, 272]. References [15, 45, 95, 105] detail the vdW-DF-cx ability to characterize bulk-structure properties for other systems (graphene bilayers, C₆₀ crystals, Si and W bulk, and other metals) that are also further analyzed in this review.

Bulk	Property	PBE	PBE0	vdW-DF-cx	Exp.
LiF	a_0 (Å)	4.005	3.952	3.976	3.972
	$E_{\rm a}$ (eV/atom)	4.289	4.224	4.417	4.46
	B_0 (GPa)	66.79	75.02	69.09	76.3
NaCl	a_0 (Å)	5.658	5.619	5.605	5.569
	$E_{\rm a}$ (eV/atom)	3.146	3.106	3.270	3.34
	B_0 (GPa)	22.92	24.60	23.94	27.6
MgO	a_0 (Å)	4.227	4.186	4.202	4.189
	$E_{\rm a}$ (eV/atom)	4.928	4.913	5.177	5.203
	B_0 (GPa)	147.8	164.6	155.5	169.8
Li	a_0 (Å)	3.405	3.429	3.407	3.453
	$E_{\rm a}$ (eV/atom)	1.555	1.492	1.610	1.658
	B_0 (GPa)	13.6	13.7	13.3	13.9
Na	a_0 (Å)	4.196	4.237	4.183	4.214
	$E_{\rm a}$ (eV/atom)	1.040	0.983	1.078	1.119
	B_0 (GPa)	7.6	7.3	8.3	7.9

testing is included because we will provide a more detailed discussion of binding in some of these systems, later. The table supplements the performance comparison for semiconductors that is included in reference [15] and extends the transitionmetal benchmarking [95].

Overall, the vdW-DF-cx seems to be extending the PBE [28] and PBEsol [29] strengths. It is useful for materials with a dense electron distribution [45, 95]. It brings this advantage to the much larger class of sparse problems [32], that is, systems that have important low-density regions.

The vdW-DF starting point, namely E_{xc}^{in} , retains all LDAcorrelations but there are also relevant vertex corrections when the electron gas is weakly perturbed. An Occam's Razor argument led us to exclude these in E_{xc}^{in} and let them emerge in the exponential re-summation that underpins the screening description. The approach is motivated, appendix B, but the question remains: does the cluster expansion retain enough nonlocal vertex corrections that vdW-DF-cx can work for traditional, dense matter (and not only for thermo-physical properties [95])?

We see the description of metal surface energies [273–275] and work functions, as summarized in reference [223], as an important test case. Also, a comparison of vdW-DF-cx and PBEsol performance is here a highly relevant assessment of the screening and vertex logic in the vdW-DF method. This follows because the gradient-corrected correlation in PBEsol itself is defined by a constraint-based fit to surface energies computed for a trusted model of metals [29].

Figure 11 compares our PBE, PBEsol, and vdW-DF-cx calculations of the surface energies for the (111), (100), and (110) surfaces of a range of metals. For comparison, we also include the computed variation of SCAN [258] and 'SCAN

+ rVV10' [69] (that is SCAN [258] combined with rVV10 [66, 67]) results, as obtained in reference [223] (with another plane-wave code but similar setup). For Ru surface energies, we focus on the fcc structure, denoted Ru*, as it constitutes the high-temperature face and therefore is relevant for the comparison with measurements [223]. All of the here-characterized functionals have the same general variation, although a consistent inclusion of vdW binding is expected to improve the description of the surface energies (and of surface-specific work functions) [223].

Table 3 reports a comparison of the performance relative to experimental observations, for the aforementioned metal set and for MgO. The experimental results are obtained by measuring the contact angle of molten droplets and have, as indicated, about a 10 percent uncertainty [273–275]. Except for MgO (where we focus on the nonpolar 110 surface), the computed surface energy is set by averaging over the facets identified in figure 11, as in reference [223]. The performance of vdW-DF-cx is on par with that of SCAN + rVV10, both landing well within the error bars on the experimental results.

Table 4 reports our PBE, PBEsol, and vdW-DF-cx calculations of facet-specific metal surface work functions. The table also includes comparison with both experimental results and SCAN + rVV10 results, as summarized and obtained in reference [223]. Here we include results both for the facets of the high-temperature Ru* phase and for the low-temperature hcp Ru phase. Overall, the inclusion of vdW interactions (in SCAN + rVV10 and in vdW-DF-cx) leads to improvements in the work-function description relative to PBEsol, and certainly relative to PBE. We note that the error bars are small for the surface-specific work functions of such metals and that this is a strong test of the functional performance and robustness.

Figure 12 reports a summary of the performance comparison that we have here provided for PBE, PBEsol and vdW-DFcx. The top (bottom) panels concern surface energies (work function values), with the left column reporting signed errors; the middle and right column report corresponding MAPD and RMSD values, respectively. We note that this comparison of vdW-DF-cx, PBE, and PBEsol robustness is made from within the same DFT code and setup.

The central message of figure 12 and of reference [95] is that vdW-DF-cx delivers a robust and accurate account of traditional materials. The vdW-DF-cx performance matches or exceeds that of the best constraint-based GGAs, i.e., PBE and PBEsol. This is true for thermo-physical properties of nonmagnetic transition metals [95] and for both surface energies and surface-specific work functions. The vdW-DF-cx errors are generally smaller and the standard deviation is comparable to those of PBEsol (and clearly better than those of PBE), figure 12. The vdW-DF-cx performance is better than that of PBEsol (and better than that of SCAN + rVV10) in the case of surface work functions where there is a smaller uncertainty on the experimental reference data, table 4.

For a discussion of method promise, it is noteworthy that in vdW-DF-cx, we build all nonlocal vertex corrections (and vdW interactions) from an exponential re-summation on



Figure 11. Surface energies ($\bar{\sigma}$) of (111), (100), and (110) metal surfaces, as obtained in different functionals. We also compare our calculations for PBE [28], PBEsol [29], and vdW-DF-cx [9] with the SCAN [258] and SCAN + rVV10 [69] (that is, rVV10-corrected [66–68] SCAN) results reported in reference [223]. As in that SCAN + rVV10 study, the high-temperature fcc (and not hcp) structure, labeled Ru*, is used for a computational study of Ru surface energies.

Table 3. Surface energies (in Jm^{-2}) of the selected metals and MgO. For metals, the calculated surface energies are averaged over (111), (100), and (110) surfaces. Four- to twelve-layer slabs were used in the linear fit for the surface energy, as in reference [223]. The headings 'SCVV' and 'CX' are abbreviations for SCAN + rVV10 and vdW-DF-cx, respectively. We focus on the high-temperature fcc phase of Ru (as marked by an asterix), but also list results for the low-temperature hcp phase. We also report MAD and MAPD values obtained for the set of metal surfaces, for comparison with reference [223]. For MgO, we concentrate on the nonpolar (110) surface.

	PBE	PBEsol	SCVV ^a	CX	Exp.
Al	0.88	1.05	1.16	1.13	$1.14\pm0.2^{\rm b}$
Cu	1.52	1.87	1.89	2.03	1.79 ± 0.19^{b}
Ru*(fcc)	2.74	3.02	2.99	3.32	3.04 ± 0.33^{b}
Ru(hcp)	2.98 ^c	3.44 ^c		3.61 ^c	_
Rh	2.42	2.84	2.81	3.08	2.66 ± 0.29^{b}
Pd	1.54	1.90	2.04	2.08	2.00 ± 0.22^{b}
Ag	0.82	1.11	1.22	1.26	1.25 ± 0.13^{b}
Pt	1.79	2.17	2.15	2.38	2.49 ± 0.26^{b}
Au	0.86	1.17	1.29	1.36	1.51 ± 0.16^{b}
MAD	0.41	0.16	0.12	0.16	
MAPD (%)	23	9	6	7	
MgO	0.89 ^d	1.01 ^d	_	1.18 ^d	1.04 ^e

^aReference [223].

^bFrom liquid metal surface tensions, references [273, 274].

^cSurface energies are here averaged over the (0001), $(11\overline{2}1)$, $(10\overline{1}0)$, and $(11\overline{2}0)$ surfaces of hcp Ru.

^dNonpolar MgO(110) surface only.

^eReference [275].

conservation laws. In contrast, the nonlocal-correlation part of PBEsol is extracted by a constraint-based fit to trusted surface energies [29]. Nevertheless, vdW-DF-cx still performs as good as PBEsol and it avoids outliers in both surface tests, figure 12, as well as for the thermo-physical properties of the nonmagnetic transition metals [95]. This vdW-DF-cx robustness is promising: it suggests a high degree of transferability for indirect (vdW-DF) functional designs that leverage the rules of screening in the electron gas.

6.3. Non-additivity in vdW interactions

A recent carbyne-wire study, reference [246], finds that the TS-MBD method [80, 81] produces an additional enhancement of the interactions at shorter distances (out to about 1-2 nm,) even for cases where the d^{-5} power law applies for the asymptotic vdW attraction. It is interesting to compare the TS-MBD enhancement near binding to that which we find among molecules, parallel wires, layered structures and at surfaces due to multipole and image-plane effects, as illustrated in figure 8 and as discussed in references [6, 10, 14, 53, 56, 181, 210, 211, 281].

The second-order expansion result equation (114) (used in the popular vdW-DF versions) is only the start of the formal vdW-DF method description for two disjunct fragments [10]. This cannot provide a non-additive description (nor in general, a complete description) of vdW interactions in the asymptotic limit [53, 105, 246, 247, 249, 250, 252–254].

At the same time, we trust even the expanded vdW-DF-cx version near binding separations. This is because of the foundation in an underlying Lindhard screening logic and because we are expanding in the plasmon propagator $S_{\rm xc}$, which (by definition) already reflects screening. This is true even if we typically approximate that by asserting the plasmon dispersion to a GGA-level internal functional [4, 8, 10]. Also, the vdW-DF method is electron-density based, not atom centered [53, 210, 211] and it has a natural inclusion of multipole enhancements at and around binding separations [3, 53, 181, 182, 210, 257]. The multipole enhancement effects are, for example, evident in figure 8: the vdW-attraction is dominated by regions that are between the graphene layers and not on the carbon atoms. This is exactly as expected when considering the importance of retaining image-plane effects on the vdW attraction [6, 10, 157, 179, 184–187].

Taken together, the observations of limitations and advantages of vdW-DF-cx merit a further discussion. We focus on the near-binding enhancements that are present both in TS-MBD [81, 246] and in vdW-DF-cx calculations, figure 9.

Table 4. Work functions ϕ (eV) for the selected metals. These values were extracted by tracking the electrostatic field outside fully relaxed slabs having 8 atomic layers. The headings 'SCVV' and 'CX' are abreviations for SCAN + rVV10 and vdW-DF-cx, respectively.

Surface	PBE	PBEsol	SCVV ^a	СХ	Exp. ^b
Al(111)	4.10	4.18	4.23	4.26	4.32 ± 0.06
Al(100)	4.19	4.25	4.42	4.29	4.32 ± 0.06
Al(110)	4.15	4.18	4.00	4.24	4.23 ± 0.13
Cu(111)	4.79	5.01	5.09	5.07	4.90 ± 0.02
Cu(100)	4.51	4.63	4.54	4.69	4.73 ± 0.1
Cu(110)	4.48	4.60	4.53	4.66	4.56 ± 0.1
Ru(11-21)	4.48	4.61	4.65	4.66	4.71 ^c
Ru(10-10)	4.54	4.64	4.97	4.71	4.60 ± 0.28
Ru(11-20)	4.36	4.46	4.72	4.54	
Ru(0001)	4.95	5.10	_	5.17	
Ru*(111)	5.08	5.22	_	5.29	
Ru*(100)	4.89	5.03	_	5.09	5.40 ± 0.11
Ru*(110)	4.46	4.56	_	4.64	
Rh(111)	5.16	5.30	5.20	5.36	5.46 ± 0.09
Rh(100)	5.08	5.22	5.37	5.29	5.30 ± 0.15
Rh(110)	4.57	4.67	4.83	4.75	4.86 ± 0.21
Pd(111)	5.26	5.38	5.47	5.45	5.67 ± 0.12
Pd(100)	5.09	5.23	5.26	5.29	5.48 ± 0.23
Pd(110)	4.72	4.84	5.09	4.91	5.07 ± 0.2
Ag(111)	4.39	4.53	4.63	4.57	4.53 ± 0.07
Ag(100)	4.22	4.38	4.37	4.43	4.36 ± 0.05
Ag(110)	4.22	4.36	4.26	4.40	4.10 ± 0.15
Pt(111)	5.68	5.80	5.97	5.87	5.91 ± 0.08
Pt(100)	5.68	5.81	6.01	5.87	5.75 ± 0.13
Pt(110)	5.31	5.42	5.36	5.51	5.53 ± 0.13
Au(111)	5.05	5.20	5.41	5.24	5.33 ± 0.06
Au(100)	5.09	5.18	5.28	5.26	5.22 ± 0.31
Au(110)	5.03	5.11	5.30	5.20	5.16 ± 0.22

^aReference [223].

^bReference [276].

^cReference [277] for polycrystalline Ru is used as reference.

First, we illustrate the implicit vdW-DF screening effect that is discussed immediately after equation (95) and in references [10, 56, 59, 105]. In figure 9, we report the vdW-DF-cx results for both the total nonlocal-correlation binding energy (given by E_c^{nl}) and for the pure-vdW binding component (given by E_c^{nl}) for the carbyne-wire interaction problem. This extraction permits us to analyze the interaction problem as if we had treated it in some would-be effective atom-centered pair-interaction form. That is, we adapt an analysis (in terms of standard vdW-corrected DFT [33, 80]) that was previously made for the TS-MBD method in references [246, 254].

Specifically, we note that all (carbon) atoms are equivalent by symmetry of the carbyne problem and define separationdependent vdW-interaction coefficients C_6^{eff} and C_6^{damped} from our full vdW-DF-cx characterization of the mutual binding:

$$\Delta E_{\rm c}^{\rm nl}(d) = -C_6^{\rm scr}(d) \sum_{i,j} \frac{1}{|\mathbf{R}_i - \mathbf{R}_j|^6}, \qquad (132)$$

$$\Delta E_{\rm c, vdW}^{\rm nl}(d) = -C_6^{\rm eff}(d) \sum_{i,j} \frac{1}{|\mathbf{R}_i - \mathbf{R}_j|^6}$$
(133)

$$\frac{\Delta E_{\rm c, vdW}^{\rm nl}(d)}{l_{\rm C-C}} = -\frac{B_5^{\rm eff}(d)}{d^5}.$$
(134)

Here, \mathbf{R}_i denotes a carbon-atom position and the summations over '*i*' and '*j*' are restricted to one or the other wire. In equation (134), we consider the interaction per unit-cell length l_{C-C} of the wires [53, 246, 250].

The top right panel of figure 9 shows the wire-separation dependence of both $C_6^{\text{scr}}(d)$ (black) and $C_6^{\text{eff}}(d)$ (red) versus $\Delta d = d - d_0$. The insert gives the dependence for negative Δd values. The $C_6^{\text{scr}}(\Delta d)$ representation of the full E_c^{nl} binding contribution can be compared to what emerges after the application of a damping function when using a atom-centered pair-wise summation description [33, 73, 74, 76–80] to a dispersion-corrected GGA; the line-up is not ideal, however, as there are also other screening effects in $E_{c\alpha}^{\text{nl}}$.

The $C_6^{\text{eff}}(d)$ representation of the pure vdW interaction, red curve in the upper right panel of figure 9, is interesting in light of recent discussions of MBD scaling effects in noncovalent-interaction problems [254]. This $C_6^{\text{eff}}(d)$ representation has itself part of the multipole effects that are expected [53, 157, 184, 185, 257] and for which there are a scaling argument in a related C₆₀ interaction problem [105]. Some dispersion-corrected DFTs seek to include such effects in higher-order atom-centered vdW coefficients [33, 78], but in the vdW-DF method we get them from looking directly at the electron-density variation itself [10, 14].

The bottom panel of figure 9 shows the separation dependence of the mutual attraction, as reflected directly by the effective wire-vdW-interaction coefficients B_5 , defined in equation (134).

There is a clear suppression of B_5 (corresponding to the saturation of the vdW attraction) when distances become smaller than the equilibrium separation value, $d_0 = 3.81$ Å, indicated by the vertical line in the bottom panel. This suppression is expected from the fact that the vdW-DF method ensures a seamless integration with LDA [4, 7, 10]. There are no divergent vdW contributions originating from density points at small distances.

There is also a clear enhancement in the effective B_5 value (relative to the asymptotic value) near binding. As shown in the bottom panel of figure 9, it does not stretch as far out as found in the TS-MBD study [246] for a metallic wire, but it is there. In other words, calculations in vdW-DF and in vdW-DF-cx reflect more than a simple pair-wise summation of atom-centered contributions and they reflect more than a traditional B_5 vdW-coefficient for wire interactions. This finding is consistent with what one of us observed and documented for parallel, semi-conducting nanotubes in 2008, in reference [53].

The here-documented enhancement of binding from purevdW contributions in vdW-DF-cx is part of what can be captured in MBD formulations [2, 3, 75, 81]. These avoid the second-order truncation in the response description [4, 7, 17, 60]. We do not here correctly reflect the dependence on the wire dimerization and hence, nature of conduction [246, 247, 249, 250]. However, our vdW-DF-cx results, figure 9, show a qualitative agreement with MBD-type calculations



Figure 12. Deviations of computed results from measurement values for mean surface energy ($\bar{\sigma}$) (top panels) and mean work functions ($\bar{\phi}$) (bottom panels) of the (111), (100), and (110) metal surfaces. The leftmost panels show the deviation from measured values (averaged over surface facets) while the middle and right panels show mean absolute percentage deviation (MAPD) and RMSD values for each of the here-characterized functionals. The error bars on the measured metal surface energies are listed in table 3. The average experimental uncertainty for surface energies is 0.22 J m⁻² or 12% for surface energies and 0.09 eV or 2.0% for work functions [223].

presented in references [10, 105, 246, 251, 254], concerning the near-binding interaction enhancements. These enhancements are caused by screening and image-plane effects [53, 105] which vdW-DF-cx certainly mimics at a GGA level through the internal functional [10]. Our vdW-DF-cx calculations reflect additional screening and image-plane effects through the cummulant-type nonlocal-correlation component.

This brings us back to the questions on the non-additivity in the vdW-DF-cx descriptions of the vdW interactions. We inquire if the vdW-DF-cx description

- Has sufficient screening mechanisms that it yields a nonadditive description at binding separations?
- Can accurately handle binding in molecular crystals, where fragments will be sitting not only at the nearest-neighbor distances but also further apart?

Both questions are relevant since the design of the vdW-DF-cx is implicit: we focus on near-binding separations, assert the plasmon dispersion from an internal GGA-level functional, and can at most reflect a GGA-level of screening, when used in the typical truncated form [4, 7, 17]. The limitations and the ameliorating factors near the optimal binding separations were discussed in reference [10]. Both questions are here discussed instead on the basis of computational results.

Figure 13 helps us address the first question. It shows that the attraction among three fullerenes (top panels) and among

three carbon nanotubes is indeed non-additive at binding separations, that is, for fully relaxed geometries. The E_c^{nl} binding for three objects exceeds that which results (middle panel) when considering a set of pair interactions. This demonstration of non-additivity for carbon structures supplements the affirmative answer we have previously obtained for a noble-gas atom [14].

We note that while there is symmetry in the binding of three fullerenes, the symmetry is broken for a triple-cluster of carbon nanotubes. The difference reflects a variation in the alignment of structural motifs in the carbon nanotube contact regions, as further discussed in reference [53] (which investigated a similar nanotube system, but with the original vdW-DF version).

Also, it is interesting that the non-additivity effects are, in fact, small for such large objects at binding separations. Here it is a 2% effect, whereas the non-additivity is a 10% effect in the case of a noble-gas trimer [14]. The smaller non-additivity is a consequence of a simple geometry effect: there is not enough room for the third structure to affect the binding of the other two when at or near binding separation.

Observe, however, that finding a small non-additivity effect for nanotubes at binding separations does not imply that screening effects are irrelevant in the present vdW-DF-cx characterization, figure 13. In fact, with the original vdW-DF version (which has the same E_c^{nl} formulation as vdW-DF-cx), one



Figure 13. Non-additivity of the nonlocal-correlation-energy binding contributions in the fullerene trimer (top panels) and a nanotube bundle (bottom panels). The left panels show the nonlocal-correlation contribution $\Delta e_c^{ll,trimer}$ to the binding energy in the trimers, directly. The middle panels show the results of simply making a superposition of three nonlocal-correlation binding contributions, $\Delta e_c^{ll,dimer'}$, each defined by a dimer. The right panels show the spatial variation in the difference between the actual trimer description and that of the dimer-based superposition, that is, $\Delta (\Delta e_c^{nl}) = \Delta e_c^{nl,timer} - \sum \Delta e_c^{nl,dimer'}$.

of us has documented that the nanotube attraction is strongly affected by multipole (or image-plane effects,) as discussed in reference [53]. The multipole effects are also directly evident in figure 9.

The observation that screening significantly affects the vdW attraction within nanotube bundles [53] and among carbyne wire, figure 9, is corroborated by the recent study of cohesion in fullerene crystals [105]. There we helped document and explain that multipole enhancements of the asymptotic intermolecular attraction are inescapable and essential, leading to modifications that have qualitative consequences [105]. For example, while the vdW-DF method underestimates the molecular C₆ interaction coefficient describing the asymptotic case, it still accurately reproduces known motifs in the structure of C₆₀ crystals [105].

For the second question, whether vdW-DF-cx is able to accurately describe molecular crystals, we make a number of observations. First is the fact that vdW-DF-cx produces the exact same asymptotic atom-C6 coefficients as does the original vdW-DF version [4]. These atomic C₆ coefficients are not bad (unlike those of vdW-DF2) for small to medium size molecules [271]. Second, in a very recent paper, other members of the vdW-DF team have launched and investigated a vdW-DF version in which the atomic C₆ coefficients are further improved from the vdW-DF/vdW-DF-cx level [16]. It is observed that limited quality of the atomic C₆ coefficients is not an intrinsic feature of the vdW-DF method, and that the description of molecules does improve when optimizing the link between the plasmon description and the internalfunctional energy-per-particle variation for better atomic C₆ coefficients. However, it is also observed that a different focus may be the key to improving the overall performance within the vdW-DF design strategy [16].

The vdW-DF-cx characterization can, under special conditions, fail to correctly describe the molecular asymptoticvdW C₆ coefficients, for example, for C₆₀ molecules [105, 252, 282] and for carbyne wires [246, 250]. However, there is also a set of arguments that the asymptotic performance is not a good discriminator for what happens at binding separations [10, 258]. The C_{60} case study makes these observations more clear: since there necessarily must exist large multipole enhancement effects it is not a priory certain that a poor asymptotic description must make vdW-DF-cx fail for the C₆₀ crystal [105]. In fact, we observe that vdW-DF-cx performance is good for both the C_{60} and the polyethylene crystals [103, 105, 107]. The vdW-DF-cx performance is excellent for the oligoacene molecular crystals where there exists a full experimental mapping across temperatures and measurements at low temperatures [98, 99].

We have already observed in section 4.1, that perhaps the strongest test of functional robustness is whether it makes accurate predictions for structure and vibrations (and surface work functions) that probe the quality of the density variation. We find that vdW-DF-cx correctly describes the known motifs of the C_{60} molecular crystals [105]. For the polyethylene crystals the experimental results vary to some extent. Nevertheless, the vdW-DF-cx characterizations of structure and defects are accurate [103, 107].

For the oligoacenes there exists accurate low-temperature characterizations both of the low-temperature structure [98] and of the highly soft intermolecular rocking and libration modes [99], figure 5. With full stress and atomic-position relaxations, the vdW-DF-cx is able to predict lattice constants and angles in close agreement with measured structure data, and therefore useful for subsequent MBPT calculation of the optical response [98]. Moreover, as shown in figure 5, there is also (for the vdW-DF-cx predictions of the naphthalene structure) an almost complete agreement between neutron data and vdW-DF-cx phonons.

Finally, for the problem of molecular-crystal binding energies there are limitations of vdW-DF-cx being a truncated implementation of the vdW-DF method. The vdW-DF-cx predictions for the C₆₀ molecular-crystal cohesive energies differ about 4 meV per carbon atoms [105]. The vdW-DF-cx predictions for oligoacene cohesive energies [98] differ 13% MAPD from low-temperature measurements, table 1. We are not aware of any MBD-based results for the oligoacenes but we mention, for comparison, that PBE-TS [80] was found to differ 28% MAPD. It is possible that the vdW-DF-cx description for molecular-crystals cohesive energies is indeed affected by the very weak (but also numerous) off-equilibrium binding contributions in molecular crystals. To capture those we must indeed go beyond the second-order truncation that secures its universal-kernel formulation [4, 7, 17].

An MBD extension of vdW-DF-cx, perhaps adapting the ideas of the first (non-truncated) nonlocal-correlation functional for materials calculations [3], vdW-DF0, is desirable. It is a possible future direction for the vdW-DF development work.



Figure 14. Exchange and correlation contributions to the binding in C_{60} dimers. Counting from the left, the first panel shows a mapping of binding contributions from the LDA XC energy. The last two panels contrast the corrections arising in the vdW-DF-cx and PBE XC functionals. Finally, the second and third panel contrast the variation in binding-energy contribution emerging from the vdW-DF(-cx) nonlocal-correlation term and from the gradient-corrected correlation part of PBE, respectively.

Going foreward, however, there is a lesson to be drawn from the present exploration of consistent vdW-DF implementations. The existing MBD formulations rely on the Dyson equation to balance the lowest-order and higher-order vdW terms [3, 75, 81], but the lower-order contributions to XC designs are picked from other considerations. Meanwhile, the vdW-DF-cx stands out as the first consistent implementation of the vdW-DF method. It has clear accuracy advantages over vdW-DF1 [4] and we see these advantages as coming from using the Lindhard or Dyson logic to balance the exchange and correlation choices, as explained already in the introduction. To fully leverage the screening logic and the current-conservation criterion on all orders of relevant XC contributions seems important. There is thus a rationale for focusing on consistent implementations of the vdW-DF method, whatever other specific XC design choices we may make in the future.

6.4. Nature of materials binding

We propose to use a spatial mapping of binding contributions [10, 14, 55–58] to explore physics implications and to detail differences between the direct and indirect XC functional design approaches.

It is natural to begin with an exploration of binding in sparse matter where some important interactions will be dominated by the vdW attraction. It is clear that the indirect-design vdW-DF-cx (compared with the GGAs) will here provide the more meaningful materials account. However, it is still interesting to track the differences to better understand the origin of the PBE shortcomings for this large class of materials.

Figure 14 presents such a comparison for the binding in a C_{60} dimer. The leftmost or first panel shows the binding contribution that arises in LDA. A set of previous studies documents that this LDA binding in such weakly-interacting,

sparse-matter systems is spurious and arises from an incorrect description of exchange [56, 59, 255, 283–285]. In fact, the PBE XC-functional correction, shown in the right-most fifth panel, compensates for and effectively offsets the spurious LDA binding.

The problem for the set of direct-design, gradient-corrected functionals is that they can only mimic sparse-matter binding at longer atom separation [10, 59, 91]. Moreover, by the nature of the approximation, they must do so in regions with a pronounced density overlap [3], that is, in the mid-region area between the fullerenes. PBE delivers a small enhancement of binding from PBE gradient-corrected correlation (central panel) in this mid-region. However, there is still a need to off set the spurious exchange binding, and that too must occur in the very same region. PBE likely strikes an optimal balance for the more strongly bonded systems, but the balance is delicate [59].

The vdW attraction is more widely distributed in space, references [3, 10, 14, 59] and figure 8, and the relative weights of exchange and correlation effects change as we vary the binding distances [12, 56, 59, 182, 210, 211]. A direct functional design (like PBE) is simply overloaded when put to the task of describing the general type of sparse-matter systems [6, 32].

On the other hand, the second panel shows that the vdW-DF-cx delivers something new, overcoming the limitations of such semilocal functionals. Having an indirect functional design logic, the vdW-DF-cx employs a truly nonlocal correlation XC functional component E_c^{nl} and can thus reflect the nature of weak binding contributions. The vdW-DF functionals have a proper mechanism to represent image-plane effects [6, 12, 53, 181, 182, 184–186] and provide transferability over a range of binding separations. As such, they correctly describe the nonlocal-correlation binding enhancement as emerging



Figure 15. Exchange and correlation contributions to the binding in bulk Si (top row), Na (middle row), and W (bottom row). The first column show mappings of the binding contribution from the LDA XC energy. The last two columns contrast the corrections arising in the vdW-DF-cx and PBE XC functionals. Finally, the second and third columns contrast the variation in binding-energy contribution emerging from the vdW-DF(-cx) nonlocal-correlation term and from the gradient-corrected correlation part of PBE, respectively.

in the density tails rather than just in the overlap region [3, 10, 14].

Contrasting the fourth and fifth panel we see that there are fundamental differences in the resulting binding description between the indirect-design vdW-DF-cx and the direct-design PBE functionals. Simply put, the vdW-DF logic allows us to put the nonlocal-correlation binding effect (second panel) in different spatial locations than the beyond-LDA-exchange correction that both functionals provide.

It is interesting that the E_c^{nl} binding in the fullerene dimer is itself defined by the competition between pure vdW attraction effects and truly nonlocal vertex effects. The fullerene dimer and graphene bilayer systems are sufficiently similar that we can simply port the observations from figure 8 and thus conclude our analysis of binding in this sparse-matter problem. The nonlocal vertex corrections occur in the density overlap regions. The vdW attraction is widely distributed but the end result is that the total nonlocal-correlation binding is dominated by the near-fullerene regions. In short, the nonlocal vertex effects help concentrate the net attraction to regions that can be thought of as molecular image planes [10, 53, 56, 59].

Additional understanding of the nature of material interactions and of vdW-DF-cx/PBE differences can be gained by mapping the spatial variation in binding contributions in (more) strongly bound bulk materials. Figures 15 and 16 contrast such mappings for Si, Na, and W. Since the vdW-DF-cx performance matches or exceeds the PBE performance here, it is relevant to ask if vdW-DF-cx has a similar or different balance between exchange and correlation than does PBE in those cases. It is also interesting to check how the balance between cumulant/vertex effects and purevdW-binding effects vary across regions with different density profiles. As in the fullerene-dimer case, the set of panels in the first column of figure 15 shows the LDA binding contributions, while the fourth and fifth column detail the binding corrections as they emerge in vdW-DF-cx and PBE, respectively. Note that the figures present calculations of the spatial variation in binding contributions for a diagonal cut through the unit cells. The Si system has one extra atom in its bulk unit cell, located in this cut.

It is clear that the LDA description in these cases provides the dominant features although there is some adjustments. Unlike in weakly bonded matter [283–285], this is the expected behavior. The indirect vdW-DF-cx and the direct PBE functionals both deliver accurate accounts of the basic properties, table 2 and references [15, 45, 95]. It is interesting that they still differ in important details of the binding description.

Specifically, there is more spatial variation (or structure) in the vdW-DF-cx than in the PBE binding descriptions. The qualitative differences are most clear in the Na case, which has the lowest cohesive energy. Quantitatively the relative differences seem largest in the Si and W cases. Overall, the PBE corrections to LDA are essentially uniformly distributed and the vdW-DF-cx corrections have a similar moderating effect in the low-density regions. However, the vdW-DF-cx descriptions also include both repulsion and an interaction-enhancement region near the atoms.

Contrasting the nonlocal-correlation binding contributions of vdW-DF-cx and PBE (shown in the second and third column, respectively) details the origin of such differences. For example, the PBE nonlocal correlation does provide a binding contribution in the inter-atomic regions, where it is actually stronger than what arises in vdW-DF-cx. However, while the nonlocal correlation binding enhances near the atoms for vdW-DF-cx, it weakens in PBE. Meanwhile, the nonlocalcorrelation binding in the middle of the cut (furthest away



Figure 16. Correlation contributions to the binding in solid Si, Na and W. The first and second columns contrast the contribution arising from the gradient-corrected correlation part of PBE and from the total nonlocal correlation in vdW-DF-cx. The third and fourth columns depict binding energy contributions from pure vdW interactions and from cumulant effects. The last column shows the binding contribution originating from the vdW-DF-cx description of the model susceptibility $\alpha(\omega)$.

from the atoms) tends to be offset by the repulsion, set by gradient corrections. In the case of the vdW-DF-cx description, however, there is enough spatial variation in the nonlocalcorrelation binding, second column, that the structure also survives in the nonlocal binding contribution, fourth column.

Figure 15 shows that the above-discussed usefulness of vdW-DF-cx (for traditional matter) does not arise simply because it mimics the PBE description in the balancing of exchange and correlation effects everywhere. Given the strong vdW-DF-cx performance, it is plausible that the vdW-DF-cx-based mapping of the nature and spatial variation of binding is trustworthy.

Diving in deeper, we analyze the nonlocal binding contributions also in terms of cumulant and pure vdW binding effects, figure 16. Again, we consider the binding contributions of Si, Na, and W and, for the reader's convenience, we repeat the PBE and vdW-DF-cx characterizations of nonlocal-correlation binding effects in the first and second columns.

We find that for these dense-matter systems there is a pronounced cancellation between vdW attraction and nonlocalvertex corrections. We tend to ignore the relevance of dispersive interactions in traditional bulk. However, the Rapcewicz and Ashcroft observation [65] that vdW forces exist also in the itinerant electron gas, is clearly important [10, 12]. On the other hand, such pure vdW forces are also dramatically compensated by a repulsion defined by nonlocal vertex corrections (shown in the fourth panel). As in the case of the fullerene dimer, these vertex effects grow in the regions where there is, relatively speaking, a large density increase by the density overlap. For dense metals, the compensation ensures that the net E_c^{nl} binding effect remains bounded at the PBE level, possibly with the exception of the description of noble metals [94, 95].

The last column of figure 16 shows our computed results for the spatial variations (for Si, Na, and W) of this local-field susceptibility term, $\Delta e_{c,\alpha}(\mathbf{r})$. Interestingly, the cumulant correction, or nonlocal vertex-correction, part (fourth column) does, to a large extend, offset the spatial variation in the LDA correlation binding contribution. For example, in Si we have pronounced signatures of LDA correlations but these are compensated by the vdW-DF-cx description of nonlocal-vertex corrections. The net effect is that there is only a moderate variation for $\Delta e_{c,\alpha}(\mathbf{r})$ in these dense-matter cases.

7. Summary and conclusions

To set the stage, we mention again that the consistent vdW-DF versions, like (spin) vdW-DF-cx, presently have no input beyond QMC data entering in the LDA definition [23, 24] and a formal MBPT result on gradient-corrected exchange [4, 7, 12, 174]. As discussed in this paper, vdW-DF-cx must build all accounts of beyond-LDA correlation from an exponential re-summation. That is, it must rely on a connection with a cumulant-type expansion to get at more general vertex corrections (and other screening effects) that are already in a GGA. It is encouraging for the indirect-functional design strategy that vdW-DF-cx still manages to perform as good as PBE, even on the (dense-matter) home turf of this versatile gradient-corrected XC functional.

There are several key observations in this review paper concerning the screening nature of consistent vdW-DF functionals.

First, we argue that by asserting the shape of the model propagator, $S_{xc} \equiv \ln(\epsilon)$ from the energy-per-particle variation of E_{xc}^{in} the vdW-DF formulation, equation (3), we incorporate some of the vertex corrections and screening that are relevant to a GGA-type description. This observation supplements the previously published interpretation [10] that the vdW-DF-cx XC energy functional E_{xc}^{DF} (also) represents a succinct

formulation of pure vdW forces, in a ZPE-coupling picture of dispersion interactions [1, 6, 12, 64, 65, 158].

Second, we document the vdW-DF-cx usefulness for descriptions of bulk and molecular cases with covalent and noncovalent binding. We find that the indirect-XC-functional design of vdW-DF-cx provides a performance that is as good as or better than the trusted, constraint-based semilocal GGAs, even for surface properties.

Third, we demonstrate that there are similarities but also important differences between PBE [28] and vdW-DF-cx [9] characterizations of binding in a bulk semiconductor and metals. It is encouraging that the vdW-DF-cx goes beyond simply mimicking PBE for dense matter (as well as for molecular matter) but builds an equally accurate description from binding contributions with different spatial variations.

We think that the vdW-DF-cx characterization of materials binding can be trusted, and we therefore proceed to deliver tools for a deeper analysis. Specifically, we have developed code for universal-kernel evaluations that can separately map binding contributions arising from (a) cumulant effects and (b) pure-vdW interactions, given by the electron–XC-hole mechanism that is illustrated in figure 2.

We have shown that screening underpins not only the response description behind pure, long-range vdW forces but also the here-identified inclusion of a cumulant-type expansion. We have furthermore discussed how these terms provide consistent-exchange vdW-DF-cx with an account of beyond-LDA vertex corrections. As such, it provides vdW-DF-cx with a seamless integration [2, 12, 166] to a GGA-type description. This is true not only in the HEG limit (as discussed in refereces [4, 7]) but also in the case of the weakly perturbed electron gas, i.e., conditions that are important for the description of bonding inside molecules and in bulk.

As for the vdW-DF-cx robustness, we expect that having better compliance with the Dyson equation provides vdW-DFcx [9, 10, 45] with an advantage over the original vdW-DF version [3, 4]. This advantage is relevant when we seek to describe traditional materials [95] and surface problems [134, 257], i.e., cases where the nonlocal correlations can be expected to play a greater relative role than in intra-molecular binding. We note that both versions have identical nonlocal correlation energy E_c^{nl} (for any given density). However, vdW-DF-cx stands out in also effectively achieving compliance with the Dyson equation. This implies in turn that vdW-DF-cx is closer in nature both to the formal ACF recast, equation (18) and to the cumulant-expansion logic [86].

In practical terms, this means for vdW-DF-cx (for vdW-DF1) that E_c^{n1} reflects an exponential re-summation and XC-hole coupling that are more relevant (less relevant). In the consistent vdW-DF versions, like vdW-DF-cx, the total correlation provides a better balance to the exchange component that is contained in the semilocal functional component $E_{xc}^0 \approx E_{xc}^{in}$.

Finally, we mention an implication regarding handling of semilocal correlation effects in indirect functional designs (like vdW-DF-cx). In picking the plasmon propagator model S_{xc} , there is a formal choice concerning retaining or discarding gradient-corrected correlations. In the vdW-DF-cx design we

avoid this explicit inclusion to avoid a double counting because we are also trying to tap into the vertex-correction handling of the cumulant-expansion logic. It is encouraging that this simple approach works, for it simplifies construction of new versions in the class of consistent vdW-DFs.

Overall, this paper documents the central role that screening plays in specifying the various terms that enter in consistent formulations of the vdW-DF design. It seems that the Occam's-razor approach of the vdW-DF method and of vdW-DF-cx is a good starting point.

Finally, it is worth discussing our calculation of the binding contribution that arises from the term, equation (109), reflecting the local-field susceptibility behavior. This term is defined by the LDA-correlation input but corrected for additional screening that we represent in an exponential re-summation. A corresponding local energy density, termed $e_{c,\alpha}(\mathbf{r})$ is given by the sum of equation (119) and the corresponding energy density term for LDA correlation.

Acknowledgment

This review of consistent implementations of the vdW-DF method is dedicated to John W Wilkins.² The analysis of the screening nature of the vdW-DF method was seeded many years ago by discussions by the first author with David C Langreth, shortly before his passing in 2011. We thank Florian Brown-Altvater for adapting the upper panel of figure 5 from figure 3 of his naphthalene phonon paper, reference [99]. We also thank Dr Andreas Hansen for computing a new CCSD(T) reference-energy value for the C22H46 alkane unfolding energy (now obtained for the structure computed with vdW-DF-cx relaxations, rather than at the reference geometry that is listed in IDISP benchmark set of GMTKN55 [35]). We are grateful for discussions over many years with Kristian Berland, Tonatiuh Rangel, Paul Erhart, Henrik Grönbeck, Jung-Hoon Lee, Jeffrey B Neaton, Gerald D Mahan, Elsebeth Schröder, and Bengt I Lundqvist. Figures 3 and 4 were made with the TikZ-Feynman package [286]. The panels of Fig 8 and Figs. 13 through 16 were made using VESTA [287]. Work supported by the Swedish Research Council (VR), the Chalmers Area-of-Advance-Materials theory activity, and the Swedish Foundation for Strategic Research (SSF) under contracts SE13-0016 and ITM17-0324. The authors also acknowledge computer allocations from the Swedish National Infrastructure for Computing (SNIC), under contracts 2016-10-12, 2018-5-138, and 2019-1-36, and from the Chalmers Center for Computing, Science and Engineering (C3SE).

Appendix A. Screening nature of electron dynamics

Consider the Dyson equation for the electron Green function

$$g(1,1') - g_0(1,1') = \int d2 \int d3 g_0(1,2)\sigma(2,3)g(3,1'),$$
(A1)

² John Wilkins passed away December 6, 2019. John helped make the electrongas tradition thrive in DFT. He impacted the lifes of the vdW-DF seniors.

and the effective Dyson equation that exists for the screened Coulomb interaction

$$W(1,1') - V(1-1') = \int d2 \int d3 V(1-2)P(2,3)W(3,1').$$
(A2)

We use Hedin notation [188] where interaction points 1, 2, 3, and 4 are normally thought of as time and space (plus spin) coordinates, but we can work with any complete representation of space.

Figure 4 explains an equivalence between the exact descriptions of the screened interaction and that of the quasi-particle electron dynamics, i.e., the Green function. The descriptions are related because they both fully retain the electron–electron interacting vertex (triangle) and are, as such, both defined by the Feynman diagram for the linked-cluster-expansion evaluation of the exact total energy (shown in panel a).

As used in the Hedin equations [188], we can express both the exact electron self energy σ and the local-field response function $\tilde{\chi}$ in terms of one diagram structure but using an electron-Coulomb vertex function $\Gamma(1'2'; 3')$. Here, 1' and 2' denote the coordinates of connected electron Green functions while 3' denotes the coordinate of a connected screenedinteraction line W. This leads to formal results [188]:

$$\sigma(1,2) = i \int_{3,4} g(1,4) \Gamma(42;3) W(3,1), \qquad (A3)$$

$$P(2,1) = -i \int_{3,4} g(1,3) \Gamma(34;2) g(4,1).$$
 (A4)

A diagram expansion of the second term in the square bracket of equation (A1) can thus be expressed

$$\int_{2} \sigma(1,2)g(2,1) = i \int_{2,3,4} g(1,4) \Gamma(42;3) g(2,1) W(3,1).$$
(A5)

Meanwhile, for the description of the screened interaction, we have

$$\int_{2} W(1,2)P(2,1) = -i \int_{2,3,4} g(1,3) \Gamma(34;2) g(4,1) W(2,1),$$
(A6)

where we have used W(2, 1) = W(1, 2).

Taking equations (A5) and (A6) together, and applying a cyclic permutation of internal integration variables 2, 3, and 4, we conclude

$$\int_{2} W(1,2)P(2,1) = -\int_{2} \sigma(1,2)g(2,1).$$
 (A7)

The result, equation (A7), can also be formulated

$$\int \frac{\mathrm{d}\omega}{2\pi} \langle \mathbf{r}_1 | \sigma(\omega) g(\omega) | \mathbf{r}_1 \rangle = -\sum_{\mu} \int \frac{\mathrm{d}\omega}{2\pi} \langle \mathbf{r}_1 | W(\omega) P^{\mu,\nu}(\omega) | \mathbf{r}_1 \rangle,$$
(A8)

where the quasi-particle dynamics is described for spin ν . This follows by Fourier transformation because we work with a time-translationally invariant problem. The result equation (A8) is used, for example, in discussion of the nature of the electron-phonon interaction problem in reference [167]. It is also consistent with the two ways reference [169] provides for the evaluation of the ground-state expectation value of the electron–electron interaction energy.

Moreover, since equation (A8) holds for any complete (coordinate) representation, we have the formal operator relation

$$\int \frac{d\omega}{2\pi} \sigma(\omega) g(\omega) = -\int \frac{d\omega}{2\pi} W(\omega) P(\omega)$$
$$\rightarrow -\int \frac{d\omega}{2\pi} V \chi(\omega)/2, \qquad (A9)$$

where $P(\omega) \equiv \sum_{\mu} P^{\mu,\nu}(\omega)$. The last line holds when $P(\omega) = \tilde{\chi}(\omega)/2$, which we shall assume to simplify the discussion.

Finally, equation (72) relates the frequency variations of the exponential-re-summation factors of the Green function and of the dielectric-function descriptions. The exact result equation (A9) suggests the connection equation (72) as the simplest assumption on the per-frequency variations in these re-summation factors.

Appendix B. Plasmon-propagator approximation for the indirect vdW-DF XC functional design

In vdW-DF we work with a double-pole model for the plasmon propagator, termed $S_{xc}(\omega)$. We assume a local dispersion of the plasmon dispersion $\omega_q(\mathbf{r})$ in a weak-perturbed electron gas, as detailed below. We note that $S_{xc}(\omega)$ must reflect a semilocal XC hole and we rely on the ideas of the formal MBPT gradient expansion [87]. In effect, in the vdW-DF method, we employ a folding of plasmon-pole contributions from two momenta [4, 47]:

$$S_{\rm xc}(\omega) = \bar{S}_{\mathbf{q},\mathbf{q}'}^{\rm RLL}(\omega) \equiv \frac{1}{2} \left[S_{\mathbf{q},\mathbf{q}'}(\omega) + S_{\mathbf{q},\mathbf{q}'}(-\omega) \right], \quad (B1)$$

$$S_{\mathbf{q},\mathbf{q}'}(\omega) = \int_{\mathbf{r}} e^{i(\mathbf{q}-\mathbf{q}')\cdot\mathbf{r}} \frac{\omega_p^2(\mathbf{r})}{[\omega + \omega_{\mathbf{q}}(\mathbf{r})][-\omega + \omega_{\mathbf{q}'}(\mathbf{r})]}.$$
 (B2)

This plasmon propagator is dominated by the diagonal components [83, 85, 86] used in the early specification of LDA [21, 22], but the off-diagonal terms reflect the effects of density gradients on the screening, i.e., the dielectric function, and hence on the resulting XC hole, equation (20).

In the popular general-geometry versions and variants [4, 8, 9, 40–42], this Rydberg, Lundqvist, Langreth (RLL) plasmonpropagator approximation is used directly with one of two specifications of $\omega_q(\mathbf{r})$, references [4, 8]. Earlier nonlocal functional generations [2, 3, 6] (for example the layered-geometry version) used a slightly modified response form that is more true to the Lundqvist single-pole plasmon-propagator form [21, 22].

The RLL form, equations (B1) and (B2), explicitly complies with time-reversal symmetry [4]. By properly specifying the dispersion form $\omega_q(\mathbf{r})$ this form complies with all known constraints on the plasmon-response behavior [4, 56, 82]. The idea is to set the assumed dispersion

$$\omega_q = q^2 / [2h(q/q_0(\mathbf{r}))], \tag{B3}$$

$$h(y) = 1 - \exp[-\gamma y^2],$$
 (B4)

in terms of an inverse length scale $q_0(\mathbf{r})$ that reflects the energyper-particle density of the internal functional [4, 7]:

$$\varepsilon_{\rm xc}^{\rm in}(\mathbf{r}) = \pi \int \frac{\mathrm{d}^3 \mathbf{q}}{(2\pi)^3} \left[\frac{1}{\omega_q(\mathbf{r})} - \frac{1}{\omega_{\rm self}(q)} \right] = -\frac{3}{4\pi} q_0. \quad (B5)$$

Here $\omega_{self}(q) = q^2/2$ is just the free-electron energy. It defines a contribution which corresponds to the E_{self} term in equation (1). The choice of γ is truly arbitrary, set to $\gamma = (4\pi/9)$, but any other choice would simply adjust the appearence—not the physics content—of the internal-functional description. In making the connection from plasmon-dispersion to the energy-per-particle variation, equation (B5), the vdW-DF method adapts the logic that Langreth and Perdew used to improve computations of surface energies beyond an underlying LDA account [25, 46, 89].

Finally, it is interesting to contrast this RLL model for a GGA-based double-pole plasmon propagator with the generalized plasmon pole (GPP) model [88] that is used, for example, in discussing GW calculations. To that end it is relevant to first consider how the RLL form should be described in a periodic system, using a gradient expansion in the arguments. Thus we express **q** in terms of a major reciprocal-lattice vector **G** component and the residual wavevector component **k** (reflecting the component inside the first Brillouin zone), in a uniquely defined separation $\mathbf{q} = \mathbf{G} + \mathbf{k}$. We note that $\mathbf{q} - \mathbf{q}' = \mathbf{G} - \mathbf{G}'$ and recast the RLL double-pole plasmon propagator

$$\bar{S}_{\mathbf{G},\mathbf{G}'}^{\mathrm{RLL}}(\mathbf{k};\omega) = \int_{\mathbf{r}} \mathrm{e}^{\mathrm{i}(\mathbf{G}-\mathbf{G}')\cdot\mathbf{r}} \frac{\omega_p^2(\mathbf{r})(\omega_{\mathbf{G}+\mathbf{k}}(\mathbf{r})\omega_{\mathbf{G}'+\mathbf{k}}(\mathbf{r})-\omega^2)}{[\omega_{\mathbf{G}+\mathbf{k}}^2(\mathbf{r})-\omega^2][\omega_{\mathbf{G}'+\mathbf{k}}^2(\mathbf{r})-\omega^2]}.$$
(B6)

This RLL form, equation (B6), should be compared to the GPP formulation:

$$S_{\mathbf{G},\mathbf{G}'}^{\mathrm{GPP}}(\mathbf{k}) = \frac{\Omega_{\mathbf{G},\mathbf{G}'}^{2}(\mathbf{k})}{\omega_{\mathbf{G},\mathbf{G}'}^{2}(\mathbf{k}) - \omega^{2}}$$
(B7)

$$\Omega^2_{\mathbf{G},\mathbf{G}'}(\mathbf{k}) = \omega^2_{pl}(\mathbf{G} - \mathbf{G}') \frac{(\mathbf{G} + \mathbf{k}) \cdot (\mathbf{G}' + \mathbf{k})}{|\mathbf{G}' + \mathbf{k}|^2}, \quad (B8)$$

where $\omega_{\rm pl}^2(\mathbf{q})$ is the Fourier transform of $\omega_{\rm pl}^2(\mathbf{r}) \propto n(\mathbf{r})$. This GPP form is set by asserting the ratio of bare and screened plasmon frequencies

$$S_{\mathbf{G},\mathbf{G}'}^{\mathrm{GPP}}(\mathbf{k},\omega=0)\omega_{\mathbf{G},\mathbf{G}'}^{2}(\mathbf{k}) = \Omega_{\mathbf{G},\mathbf{G}'}^{2}(\mathbf{k}). \tag{B9}$$

For semi-conductors and insulators, this is often done in an initial RPA calculation for $S_{G,G'}^{GPP}(\mathbf{k}, \omega = 0)$. The GPP is believed to be broadly applicable for a characterization of response. The RLL plasmon propagator is similar to this GPP form with one important difference: the absence of an explicit longitudinal projection $(\mathbf{G} + \mathbf{k}) \cdot (\mathbf{G}' + \mathbf{k})$ in the RLL form, equation (B6).

The vdW-DF difference from the GPP is deliberate. The longitudinal projection is an absolute must for any proper description for the electron gas response. In the GPP it is hard-wired from the start by the need to comply with the *f*-sum

rule. In the vdW-DF, we delay the enforcement of the longitudinal projection in $S_{xc} \equiv \overline{S}^{RLL}$, and put it instead in the κ_{long} criterion. The projection is no less important for the vdW-DFcx version. However, by lifting it to an explicit constraint on the electrodynamics description, equation (34), we position the vdW-DF framework and the vdW-DF-cx version for an automatic inclusion of vdW forces. This is done in the presence of screening by the electron gas, that is, in the setting first analyzed by the Ashcroft group [1, 10, 12, 64, 65].

Appendix C. Nature of a full vdW-DF method implementation

The analysis of the optical-response description, equations (102) and (105), suggests a path to go beyond the second-order expansion that is reflected in the popular general-geometry vdW-DF version [4]. While the existing functionals are given by equations (2) and (3), a full implementation of the vdW-DF method is formally given

$$E_{\rm xc}^{\rm DF} = \sum_{n=1}^{\infty} \sum_{m \ge n}^{\infty} \sum_{i} E_{{\rm xc},m,n}^{{\rm DF},(i)},\tag{C1}$$

$$E_{\mathrm{xc},m,n}^{\mathrm{DF},(i)} = -\int_0^\infty \frac{\mathrm{d}u}{2\pi} \frac{1}{m} \operatorname{Tr}\{\nabla \cdot \boldsymbol{\alpha}_{m,n}^{(i)}(\mathrm{i}u) \cdot \nabla V\}.$$
 (C2)

We may, for example, add some or all triples (identified as reflecting n = 3 factors of S_{xc}), since each of the terms equation (C2) is, in itself, charge conserving.

In seeking such generalizations, however, we recommend always keeping all terms '(*i*)' containing *n* powers of S_{xc} (up to a given maximum exponent value $n \leq l$) if truncating an expansion like equation (C2). This is done in vdW-DF1, vdW-DF2, and (s)vdW-DF-cx [4, 8, 9, 11]. We see the plasmon propagator S_{xc} as the natural physics property to use as a building block in a vdW-DF-based XC density functional [2–4, 10, 64, 65, 85, 86, 163, 164]. It is designed to reflect the simpler physics of the weakly perturbed electron gas. That is, the choice of S_{xc} as our expansion variable facilitates seamless integration with a near-HEG description [4, 47].

In any case, it is straightforward to sort XC functional contributions subject to the Dyson logic (even when components are expanded in S_{xc} factors). Each and every susceptibility term $\alpha_{m,n}^{(i)}$ in equation (102) corresponds to a given order (*m*) in equation (30)—but that is also the order it takes in the expansion of the logarithm in the full vdW-DF method specification.

Appendix D. Nonlocal-correlation contributions

Finally, we explore the nature of the partial nonlocalcorrelation energy contributions, equations (110) and (113), when restricted to exactly a second-order expansion in $S_{\rm xc}$. Together they define the standard $E_{\rm c}^{\rm nl}$ approximation, equation (2). Using a Fourier-transformed representation [4] of the plasmon-propagator model $S_{\rm xc}(\omega, \mathbf{q}, \mathbf{q}')$ these components can be expressed

$$E_{c,\alpha}^{nl} \approx \frac{1}{2} \int \frac{d\mathbf{q}}{(2\pi)^3} \int \frac{d\mathbf{q}'}{(2\pi)^3}$$
(D1)

$$\times \int_{0} \frac{1}{2\pi} S_{\rm xc}(u, \mathbf{q}, \mathbf{q}') S_{\rm xc}(u, \mathbf{q}', \mathbf{q}),$$
(D2)

$$E_{c,vdW}^{nl} \approx -\frac{1}{2} \int \frac{d\mathbf{q}}{(2\pi)^3} \int \frac{d\mathbf{q}'}{(2\pi)^3} (\hat{\mathbf{q}} \cdot \hat{\mathbf{q}'})^2 \\ \times \int_0^\infty \frac{du}{2\pi} S_{xc}(iu, \mathbf{q}, \mathbf{q}') S_{xc}(iu, \mathbf{q}', \mathbf{q}), \qquad (D3)$$

where $\hat{\mathbf{q}} = \mathbf{q}/|\mathbf{q}|$. The factor $(\hat{\mathbf{q}} \cdot \hat{\mathbf{q}'})^2$ reflects the longitudinal projection in the vdW term.

Each of the terms in the second order expansion for E_c^{nl} can be expressed in terms of corresponding partial kernels,

$$E_{\mathbf{c},\alpha}^{\mathrm{nl}}[n] = \frac{1}{2} \int d\mathbf{r} \, d\mathbf{r}' n(\mathbf{r}) \, \phi_{\alpha}(\mathbf{r},\mathbf{r}') \, n(\mathbf{r}'), \qquad (\mathrm{D4})$$

$$E_{\mathrm{c,\,vdW}}^{\mathrm{nl}}[n] = \frac{1}{2} \int \mathrm{d}\mathbf{r} \,\mathrm{d}\mathbf{r}' n(\mathbf{r}) \phi_{\mathrm{vdW}}(\mathbf{r},\mathbf{r}') n(\mathbf{r}'), \qquad (\mathrm{D5})$$

as in the original vdW-DF determination [4, 47, 56, 171]. As discussed in the main text and elsewhere [4, 5, 10], the evaluation of E_c^{nl} , and of the here-sought components, is naturally cast by tracking the variation in a set of inverse length scales $q_0(\mathbf{r})$ and $q_0(\mathbf{r}')$. Meanwhile, the partial and full kernels are universal and can be tabulated in advance in terms of an effective separation $D = [(q_0 + q_0')/2]|\mathbf{r} - \mathbf{r}'|$ and in terms of an asymmetry parameter $\delta = (q_0 - q_0')/(q_0 + q_0')$, reference [4].

To specify the form of the universal (partial) kernels, we adapt the derivation presented in references [56, 60]. We first extract a density component from the classical-plasmon numerator $\omega_{\rm pl}^2(\mathbf{r}) = 4\pi n(\mathbf{r})/m$ from the plasmon-propagator model [4, 56]

$$S_{\rm xc}(\omega, \mathbf{q}, \mathbf{q}') = \int_{\mathbf{r}} e^{-i(\mathbf{q} - \mathbf{q}') \cdot \mathbf{r}} n(\mathbf{r}) \\ \times \frac{2\pi}{m} \left[f(\omega, \mathbf{q}, \mathbf{q}', \mathbf{r}) + f(\omega, \mathbf{q}', \mathbf{q}, \mathbf{r}) \right],$$
(D6)

$$f(\omega, \mathbf{q}, \mathbf{q}', \mathbf{r}) = \frac{1}{[\omega + \omega_{\mathbf{q}}][-\omega + \omega_{\mathbf{q}'}]},$$
(D7)

where $\omega_{\mathbf{q}}$ denotes the assumed plasmon dispersion [4, 47, 171]. This allows us to split the special kernels into a spatial and a frequency-dependent part [4, 56, 171]

$$\phi_{\alpha}(\mathbf{r},\mathbf{r}') = \int_{0}^{\infty} \frac{\mathrm{d}q}{(2\pi)^{3}} q^{2} \int_{0}^{\infty} \frac{\mathrm{d}q'}{(2\pi)^{3}} (q')^{2}$$
$$\times (4\pi)^{2} W_{\alpha}(a = q |\mathbf{r} - \mathbf{r}'|, b = q |\mathbf{r} - \mathbf{r}'|)$$
$$\times I_{\text{freq}}(q, q', \mathbf{r}, \mathbf{r}'), \qquad (D8)$$

in the same way as can be done for the total E_c^{nl} energy [47, 56, 171], equation (2). The frequency-integration term is by construction unchanged from the original kernel evaluation:

$$I_{\text{freq}}(q,q';\mathbf{r},\mathbf{r}') = \frac{(2\pi)^2}{m^2} T[\omega_q(\mathbf{r}), \omega_{q'}(\mathbf{r}), \omega_q(\mathbf{r}'), \omega_{q'}(\mathbf{r}')] \quad (D9)$$

with T[w, x, y, z] > 0 as defined in reference [4], equation (15).

The kernels from the two partial contributions differ in how we handle the spatial integrations in equation (D8). For the $(S_{xc})^2$ contribution, we simply obtain [56]

$$(4\pi)^{2} W_{\alpha}(a = q |\mathbf{r} - \mathbf{r}'|, b = q |\mathbf{r} - \mathbf{r}'|)$$

$$\equiv \int d\Omega_{1} \int d\Omega_{2} e^{-i\mathbf{q}(\mathbf{r} - \mathbf{r}')} e^{i\mathbf{q}'(\mathbf{r} - \mathbf{r}')}.$$

$$= \frac{\sin(a)\sin(b)}{ab}.$$
(D10)

For the pure vdW-attraction term we need

$$(4\pi)^2 W_{\rm vdW}(a = q |\mathbf{r} - \mathbf{r}'|, b = q |\mathbf{r} - \mathbf{r}'|)$$

$$\equiv -\int d\Omega_1 \int d\Omega_2 (\hat{\mathbf{q}} \cdot \hat{\mathbf{q}}')^2 e^{-i\mathbf{q}(\mathbf{r} - \mathbf{r}')} e^{i\mathbf{q}'(\mathbf{r} - \mathbf{r}')}. \quad (D11)$$

This evaluation is also indirectly detailed in reference [56]. The resulting specification is

$$(4\pi)^2 W_{\rm vdW}(a = q | \mathbf{r} - \mathbf{r}'|, b = q | \mathbf{r} - \mathbf{r}'|)$$

= $-\frac{1}{(ab)^3} \left(2a \cos(a) \left[3b \cos(b) + (b^2 - 3) \sin(b) \right] + 2 \sin(a) \cos(b)(a^2 - 3)b + \sin(a) \sin(b) \left[a^2(b^2 - 2) - 2(b^2 - 3) \right] \right).$ (D12)

This determination follows simply from performing the wavevector integrations in polar coordinates [47, 56, 60, 171].

Figure 7 presents our numerical evaluation of both the full vdW-DF kernel ϕ and of the kernel components, ϕ_{α} and ϕ_{vdW} . The results are given in terms of universal kernel forms, $\Phi_{\alpha}(D, \delta)$, $\Phi_{vdW}(D, \delta)$ and $\Phi(D, \delta) = \Phi_{\alpha} + \Phi_{vdW}$, that permit evaluations under various density conditions [4, 7, 17]. For example, $\phi_{vdW}(\mathbf{r}, \mathbf{r}') = \Phi_{vdW}(D, \delta)$ where the functional forms of $D(\mathbf{r}, \mathbf{r}')$ and $\delta(\mathbf{r}, \mathbf{r}')$ are given above.

The set of blue and red curves compares the variation of Φ_{α} and of Φ_{vdW} as a function of D for a selection of indicated δ values, respectively. Finally, the set of the black curves tracks the corresponding variation in the full vdW-DF kernel, Φ . It will, in the typical case, be relevant to use such kernels also when $q_0(\mathbf{r})$ differs from $q_0(\mathbf{r}')$, and δ will seldom be zero.

The kernel Φ_{vdW} is always negative, and $E_{c, vdW}^{nl}[n]$ reflects a pure-vdW attraction. This is expected as it is formed from the term that has an explicit longitudinal projection (that cannot be removed by a partial integration as in equation (92)). The full vdW-DF behavior is offset by the other kernel Φ_{α} which is always positive. The $E_{c,\alpha}^{nl}[n]$ term does not comply with the logic of equation (55) and does not secure a seamless integration, by itself.

One of the advantages of the vdW-DF method, and of vdW-DF-cx in particular, is that seamless integration is provided when combining the terms, as E_c^{nl} does. The solid black curve in figure 7 shows the total kernel at $\delta = 0$, i.e., the condition that $q_0(\mathbf{r}) = q_0(\mathbf{r}')$, which is relevant for discussing the

homogeneous-gas limit. For this curve, the short-range repulsion and longer-ranged attraction components fully compensate each other, the curve for E_c^{nl} (given by equation (2)) integrates to zero in HEG, as it was designed to do [4, 5, 171].

ORCID iDs

Per Hyldgaard b https://orcid.org/0000-0001-5810-8119 Yang Jiao b https://orcid.org/0000-0003-4771-0088 Vivekanand Shukla b https://orcid.org/0000-0001-7724-6357

References

- Andersson Y, Langreth D C and Lundqvist B I 1996 van der Waals interactions in density-functional theory *Phys. Rev. Lett.* 76 102
- [2] Rydberg H, Lundqvist B I, Langreth D C and Dion M 2000 Tractable nonlocal correlation density functionals for flat surfaces and slabs *Phys. Rev.* B 62 6997
- [3] Rydberg H, Dion M, Jacobson N, Schröder E, Hyldgaard P, Simak S I, Langreth D C and Lundqvist B I 2003 van der Waals density functional for layered structures *Phys. Rev. Lett.* 91 126402
- [4] Dion M, Rydberg H, Schröder E, Langreth D C and Lundqvist B I 2004 van der Waals density functional for general geometries *Phys. Rev. Lett.* **92** 246401
- [5] Dion M, Rydberg H, Schröder E, Langreth D C and Lundqvist B I 2005 Erratum: van der Waals density functional for general geometries [Phys. Rev. Lett. 92, 246401 (2004)] *Phys. Rev. Lett.* 95 109902(E)
- [6] Langreth D C, Dion M, Rydberg H, Schröder E, Hyldgaard P and Lundqvist B I 2005 van der Waals density functional theory with applications *Int. J. Quantum Chem.* 101 599
- [7] Thonhauser T, Cooper V R, Li S, Puzder A, Hyldgaard P and Langreth D C 2007 van der Waals density functional: selfconsistent potential and the nature of the van der Waals bond *Phys. Rev. B.* 76 125112
- [8] Lee K, Murray È D, Kong L, Lundqvist B I and Langreth D C 2010 Higher-accuracy van der Waals density functional *Phys. Rev. B* 82 081101(R)
- [9] Berland K and Hyldgaard P 2014 Exchange functional that tests the robustness of the plasmon description of the van der Waals density functional *Phys. Rev. B* 89 035412
- [10] Hyldgaard P, Berland K and Schröder E 2014 Interpretation of van der Waals density functionals *Phys. Rev.* B 90 075148
- [11] Thonhauser T, Zuluaga S, Arter C A, Berland K, Schröder E and Hyldgaard P 2015 Spin signature of nonlocal correlation binding in metal-organic frameworks *Phys. Rev. Lett.* 115 136402
- [12] Berland K, Cooper V R, Lee K, Schröder E, Thonhauser T, Hyldgaard P and Lundqvist B I 2015 van der Waals forces in density functional theory: a review of the vdW-DF method *Rep. Prog. Phys.* **78** 066501
- [13] Berland K, Jiao Y, Lee J-H, Rangel T, Neaton J B and Hyldgaard P 2017 Assessment of two hybrid van der Waals density functionals for covalent and noncovalent binding of molecules J. Chem. Phys. 146 234106
- [14] Jiao Y, Schröder E and Hyldgaard P 2018 Signatures of van der Waals binding: a coupling-constant scaling analysis *Phys. Rev.* B 97 085115
- [15] Jiao Y, Schröder E and Hyldgaard P 2018 Extent of Fockexchange mixing for a hybrid van der Waals density functional? J. Chem. Phys. 148 194115

- [16] Berland K, Chakraborty D and Thonhauser T T 2019 van der Waals density functional with corrected C₆ coefficients *Phys. Rev.* B **99** 195418
- [17] Román-Pérez G and Soler J M 2009 Efficient implementation of a van der Waals density functional: application to double-wall carbon nanotubes *Phys. Rev. Lett.* 103 096102
- [18] Hjort Larsen A, Kuisma M, Löfgren J, Pouillon Y, Erhart P and Hyldgaard P 2017 libvdwxc: a library for exchangecorrelation functionals in the vdW-DF family *Modelling Simul. Mater. Sci. Eng.* 25 065004
- [19] Singwi K S, Tosi M P, Land R H and Sjölander A 1968 Electron correlations at metallic densities *Phys. Rev.* 176 589
- [20] Singwi K S, Sjölander A, Tosi M P and Land R H 1969 Electron correlations at metallic densities—III Solid State Commun. 7 1503
- [21] Hedin L and Lundqvist B I 1971 Explicit local exchangecorrelation potentials J. Phys. C: Solid State Phys. 4 2064
- [22] Gunnarsson O and Lundqvist B I 1976 Exchange and correlation in atoms, molecules, and solids by the spin-densityfunctional formalism *Phys. Rev.* B 13 4274
- [23] Perdew J P and Wang Y 1992 Accurate and simple analytic representation of the electron-gas correlation energy *Phys. Rev.* B 45 13244
- [24] Perdew J P and Wang Y 1992 Pair-distribution function and its coupling-constant average for the spin-polarized electron gas *Phys. Rev.* B 46 12947
- [25] Langreth D C and Perdew J P 1977 Exchange-correlation energy of a metallic surface: wave-vector analysis *Phys. Rev.* B 15 2884
- [26] Langreth D C and Mehl M J 1981 Easily implementable nonlocal exchange-correlation energy functional *Phys. Rev. Lett.* 47 446
- [27] Perdew J P and Wang Y 1986 Accurate and simple density functional for the electronic exchange energy: generalized gradient approximation *Phys. Rev. B* 33 8800(R)
- [28] Perdew J P, Burke K and Ernzerhof M 1996 Generalized gradient approximation made simple *Phys. Rev. Lett.* 77 3865
- [29] Perdew J P, Ruzsinszky A, Csonka G I, Vydrov O A, Scuseria G E, Constantin L A, Zhou X and Burke K 2008 Restoring the density-gradient expansion for exchange in solids and surfaces *Phys. Rev. Lett.* **100** 136406
- [30] Burke K 2012 Perspective on density functional theory J. Chem. Phys. **136** 150901
- [31] Becke A D 2014 Perspective: fifty years of density-functional theory in chemical physics J. Chem. Phys. **140** 18A301
- [32] Langreth D C, Lundqvist B I, Chakarova-Käck S D, Cooper V R, Dion M, Hyldgaard P, Kelkkanen A, Kleis J, Kong L, Li S, Moses P G, Murray E, Puzder A, Rydberg H, Schröder E and Thonhauser T 2009 A density functional for sparse matter J. Phys.: Condens. Matter. 21 084203
- [33] Grimme S, Antony J, Ehrlich S and Krieg H 2010 A consistent and accurate ab initio parametrization of density functional dispersion correction (DFT-D) for the 94 elements H–Pu J. Chem. Phys. 132 154104
- [34] Zhang Y and Yang W 1998 Comment on 'generalized gradient approximation made simple' *Phys. Rev. Lett.* 80 890
- [35] Goerigk L, Hansen A, Bauer C, Ehrlich S, Najibi A and Grimme S 2017 A look at the density functional theory zoo with the advanced GMTKN55 database for general main group thermochemistry, kinetics and noncovalent interactions *Phys. Chem. Chem. Phys.* **19** 32184
- [36] Giannozzi P et al 2017 Advanced capabilities for materials modelling with quantum espresso J. Phys.: Condens. Matter. 29 465901

- [37] Witte J, Neaton J B and Head-Gordon M 2018 Push it to the limit: comparing periodic and local approaches to density functional theory for intermolecular interactions *Mol. Phys.* **117** 1298
- [38] Gonze X et al 2005 A brief introduction to the ABINIT software package Z. Kristallogr. 220 558
- [39] Makov G and Payne M C 1995 Periodic boundary conditions in ab initio calculations *Phys. Rev.* B 51 4014
- [40] Klimeš J, Bowler D R and Michaelides A 2010 Chemical accuracy for the van der Waals density functional J. Phys.: Condens. Matter. 22 022201
- [41] Cooper V R 2010 van der Waals density functional: an appropriate exchange functional *Phys. Rev. B* 81 161104(R)
- [42] Klimeš J, Bowler D R and Michaelides A 2011 van der Waals density functionals applied to solids *Phys. Rev. B* 83 195131
- [43] Hamada I 2014 van der Waals density functional made accurate *Phys. Rev. B* 89 121103(R)
- [44] Christivana M, Obata M and Oda T 2019 Noncollinear magnetic structure with spin-dependent van der Waals density functional approach: β-phase solid oxygen J. Phys. Soc. Japan 88 104601
- [45] Berland K, Arter C A, Cooper V R, Lee K, Lundqvist B I, Schröder E, Thonhauser T and Hyldgaard P 2014 van der Waals density functionals built upon the electron-gas tradition: facing the challenge of competing interactions J. Chem. Phys. 140 18A539
- [46] Langreth D C and Perdew J P 1975 The exchange-correlation energy of a metallic surface Solid State Commun. 17 1425
- [47] Rydberg H 2001 Nonlocal correlations in density functional theory *Ph.D. Thesis* Department of Applied Physics, Chalmers University of Technology (Göteborg, Sweden) http://bitmath.se/rydberg/Thesis
- [48] Levy M 1991 Density-functional exchange correlation through coordinate scaling in adiabatic connection and correlation hole *Phys. Rev.* A 43 4637
- [49] Levy M, March N H and Handy N C 1996 On the adiabatic connection method, and scaling of electron–electron interactions in the Thomas–Fermi limit J. Chem. Phys. 104 1989
- [50] Görling A and Levy M 1993 Correlation-energy functional and its high-density limit obtained from a coupling-constant perturbation expansion *Phys. Rev.* B 47 13105
- [51] Ernzerhof M, Perdew J P and Burke K 1997 Coupling-constant dependence of atomization energies *Int. J. Quantum Chem.* 64 285
- [52] Barker D, Fors A, Lindgren E, Olesund A and Schröder E 2020 Filter function of graphene oxide: trapping perfluorinates molecules J. Chem. Phys. 152 024704
- [53] Kleis J, Schröder E and Hyldgaard P 2008 Nature and strength of bonding in a crystal of semiconducting nanotubes: van der Waals density functional calculations and analytical results *Phys. Rev. B.* 77 205422
- [54] Chakarova-Käck S D, Vojvodic A, Kleis J, Hyldgaard P and Schröder E 2010 Binding of polycyclic aromatic hydrocarbons and graphene dimers in density functional theory *New J. Phys.* **12** 013017
- [55] Gulans A, Puska M J and Nieminen R M 2009 Linear-scaling self-consistent implementation of the van der Waals density functional *Phys. Rev.* B **79** 201105
- [56] Berland K 2012 Connected by voids: interactions and screening in sparse matter *PhD Thesis* Department of Microtechnology and Nanoscience—MC2, Chalmers University of Technology (Göteborg, Sweden)
- [57] Lazić P, Atodiresei N, Caciuc V, Brako R, Gumhalter B and Blügel S 2012 Rationale for switching to nonlocal functionals in density functional theory J. Phys.: Condens. Matter. 24 424215

- [58] Callsen M, Atodiresei N, Caciuc V and Blügel S 2012 Semiempirical van der Waals interactions versus ab initio nonlocal correlation effects in the thiophene-Cu(111) system Phys. Rev. B 86 085439
- [59] Berland K and Hyldgaard P 2013 Analysis of van der Waals density functional components: binding and corrugation of benzene and C₆₀ on boron nitride and graphene *Phys. Rev.* B 87 205421
- [60] Schröder E, Cooper V R, Berland K, Lundqvist B I, Hyldgaard P and Thonhauser T 2017 The vdW-DF family of non-local exchange-correlation functionals *Non-Covalent Interactions in Quantum Chemistry and Physics: Theory and Applications* ed A O de la Roza and G Di Labio (Amsterdam: Elsevier) pp 241–274
- [61] Bohr N and Lindhard J 1954 Electron capture and loss by heavy ions penetrating through matter Mat.-Fys. Medd. K. Dan. Vidensk. Selsk. 28 1–30
- [62] Lindhard J 1954 On the properties of a gas of charged particles Mat.-Fys. Medd. K. Dan. Vidensk. Selsk. 28 1–57
- [63] Pines D and Nozières P 1989 The Theory of Quantum Liquids: Normal Fermi Liquids (Redwood City, CA: Addison-Wesley)
- [64] Maggs A C and Ashcroft N W 1987 Electronic fluctuation and cohesion in metals *Phys. Rev. Lett.* 59 113
- [65] Rapcewicz K and Ashcroft N W 1991 Fluctuation attraction in condensed matter: a nonlocal functional approach *Phys. Rev.* B 44 4032
- [66] Vydrov O A and Van Voorhis T 2010 Nonlocal van der Waals density functional: the simpler the better *J. Chem. Phys.* 133 244103
- [67] Sabatini R, Gorni T and de Gironcoli S 2013 Nonlocal van der Waals density functional made simple and efficient *Phys. Rev.* B 87 041108(R)
- [68] Björkman T 2014 Testing several recent van der Waals density functionals for layered structures J. Chem. Phys. 141 074708
- [69] Peng H, Yang Z-H, Perdew J P and Sun J 2016 Versatile van der Waals density functional based on a meta-generalized gradient approximation *Phys. Rev.* X 6 041005
- [70] Peng H and Perdew J P 2017 Rehabilitation of the Perdew-Burke-Ernzerhof generalized gradient approximation for layered materials *Phys. Rev.* B 95 081105(R)
- [71] Terentjev A V, Cortona P, Constantin L A, Pitarke J M, Della Salla F and Fabiano E 2018 Solid-state testing of a van-der-Waals-corrected exchange-correlation functional based on the semiclassical atom theory *Computation* 6 7
- [72] Terentjev A V, Constantin L A and Pitarke J M 2018 Dispersion-corrected PBEsol exchange-correlation functional *Phys. Rev.* B 98 214108
- [73] Grimme S 2004 Accurate description of van der Waals complexes by density functional theory including empirical corrections J. Comput. Chem. 25 1463
- [74] Grimme S 2006 Semiempirical hybrid density functional with perturbative second-order correlation J. Chem. Phys. 124 034108
- [75] Caldeweyher E, Bannwarth C and Grimme S 2017 Extension of the D3 dispersion coefficient model J. Chem. Phys. 147 034112
- [76] Becke A D and Johnson E R 2005 A density-functional model of the dispersion interaction J. Chem. Phys. 123 154101
- [77] Becke A D and Johnson E R 2007 Exchange-hole dipole moment and the dispersion interaction revisited J. Chem. Phys. 127 154108
- [78] Otero-de-la-Roza A and Johnson E R 2012 Van der Waals interactions in solids using the exchange-hole dipole moment model J. Chem. Phys. 136 174109
- [79] Silvestrelli P L 2008 van der Waals interactions in DFT made easy by Wannier functions *Phys. Rev. Lett.* **100** 053002

- [80] Tkatchenko A and Scheffler M 2009 Accurate molecular van der Waals interactions from ground-state electron density and free-atom reference data *Phys. Rev. Lett.* **102** 073005
- [81] Tkatchenko A, DiStasio R A Jr, Car R and Scheffler M 2012 Accurate and efficient method for many-body van der Waals interactions *Phys. Rev. Lett.* **108** 236402
- [82] Nozières P and Pines D 1958 Correlation energy of a free electron gas Phys. Rev. 111 442
- [83] Lundqvist B I 1967 Single-particle spectrum of degenerate electron gas. I. Structure of spectral weight function *Phys. Kondens. Mater.* 6 193
- [84] Singwe K S, Sjölander A, Tosi M P and Land R H 1970 Electron correlations at metallic densities. IV Phys. Rev. B 1 1044
- [85] Lundqvist B I 1969 Characteristic structure in core electron spectra of metals due to the electron-plasmon coupling *Phys. Kondens. Mater.* 9 236
- [86] Langreth D C 1970 Singularities in the x-ray spectra of metals *Phys. Rev.* B 1 471
- [87] Langreth D C 1976 Linear and nonlinear response theory with applications 1975 NATO Advanced Study Institute on Linear and Nonlinear Transport in Solids, volume B17 (Antwerpen 1975) (New York, NY: Plenum) pp 3–32
- [88] Hybertsen M S and Louie S G 1986 Electron correlation in semiconductors and insulators: band gabs and quasiparticle energies *Phys. Rev.* B 34 5390
- [89] Langreth D C and Perdew J P 1980 Theory of nonuniform electronic systems. I. Analysis of the gradient approximation and a generalization that works *Phys. Rev.* B 21 5469
- [90] Levy M and Perdew J P 1985 Hellmann-Feynman Virial, and scaling requisites for the exact universal density functionals. Shape of the correlation potential and diamagnetic susceptibility for atoms *Phys. Rev.* A **32** 2010
- [91] Langreth D C and Vosko S H 1987 Exact electron-gas response functions at high density *Phys. Rev. Lett.* 59 497
- [92] Langreth D C 2010 private communication.
- [93] Perdew J P, Burke K and Wang Y 1996 Generalized gradient approximation for the exchange-correlation hole of a many-electron system *Phys. Rev.* B 54 16533
- [94] Ambrosetti A and Silvestrelli P L 2016 Cohesive properties of noble metals by van der Waals-corrected density functional theory: Au, Ag, and Cu as case studies *Phys. Rev.* B 94 045124
- [95] Gharaee L, Erhart P and Hyldgaard P 2017 Finite-temperature properties of nonmagnetic transition metals: comparison of the performance of constraint-based semilocal and nonlocal functionals *Phys. Rev.* B **95** 085147
- [96] Linderälv C, Lindman A and Erhart P 2018 A unifying perspective on oxygen vacancies in wide band gab oxides J. Phys. Chem. Lett. 9 222
- [97] Wang Y-P, Li X-G, Fry J N and Cheng H-P 2016 Firstprinciples studies of electric effects on the electronic structure of trilayer graphene *Phys. Rev.* B 94 165428
- [98] Rangel T, Berland K, Sharifzadeh S, Brown-Altvater F, Lee K, Hyldgaard P, Kronik L and Neaton J B 2016 Structural and excited-state properties of oligoacene crystals from first principles *Phys. Rev.* B 93 115206
- [99] Brown-Altvater F, Rangel T and Neaton J B 2016 Ab initio phonon dispersion in crystalline naphthalene using van der Waals density functionals *Phys. Rev.* B 93 195206
- [100] Mehboudi M, Fregoso B M, Yang Y, Zhu W, van der Zande A, Ferrer J, Bellaiche L, Kumar P and Barraza-Lopez S 2016 Structural phase transition and material properties of few-layer monochalcogenides *Phys. Rev. Lett.* **117** 246802
- [101] Mehbourdi M, Dorio A M, Zhu W, van der Zande A, Churchill H O H, Pacheco-Sanjuan A A, Harriss E O, Kumar P and Barraza-Lopez S 2016 Two-dimensional disorder in black

phosporus and monochalcogenide monolayers *Nano Lett.* **16** 1704

- [102] Loncaric I, Popovic J, Despoja V, Burazer S, Grgicevic I, Popovic D and Skoko Z 2017 Reversible thermosalient effect of N/-2-propylidene-4-hydroxybenzohydrazide accompanied by an immense negative compressibility: structural and theoretical arguments aiming toward the elucidation of jumping phenomenon *Cryst. Growth Des.* 17 4445
- [103] Olsson P A T, Schröder E, Hyldgaard P, Kroon M, Andreasson E and Bergvall E 2017 Ab initio and classical atomistic modelling of structure and defects in crystaline orthorhombic polyethylene: twin boundaries, slip interfaces, and nature of barriers *Polymer* 121 234
- [104] Wang X, Esfarjani K and Zebarjadi M 2017 First-principles calculation of charge transfer at the silicon–organic interface J. Phys. Chem. C 121 15529
- [105] Tao J, Jiao Y, Mo Y, Yang Z-H, Zhu J-X, Hyldgaard P and Perdew J P 2018 First-principles study of the binding energy between nanostructures and its scaling with system size *Phys. Rev.* B **97** 155143
- [106] Tao J, Perdew J P, Tang H and Shahi C 2018 Origin of the size-dependence of the equilibrium van der Waals binding between nanostructures J. Chem. Phys. 148 074110
- [107] Olsson P A T, Hyldgaard P, Schröder E, Jutemar E P, Andreasson E and Kroon M 2018 Ab initio investigations of monoclinic phase stability and martensitic transformation in crystalline polyethylene *Phys. Rev. Mater.* 2 075602
- [108] Barraza-Lopes S, Kaloni T P, Pourdel S P and Kumar P 2018 Tuning the ferroelectric-to-paraelectric transition temperature and dipole orientation of group-IV monochalcogenide monolayers *Phys. Rev.* B **97** 024110
- [109] Choudhary K, Cheon G, Reed E and Tavazza F 2018 Elastic properties in bulk and low-dimensional materials using van der Waals density functional *Phys. Rev.* B 98 014107
- [110] Ishibashi S, Horiuchi S and Kumai R 2018 Computational findings of metastable ferroelectric phases of squaric acid *Phys. Rev. B* 97 184102
- [111] Chang K, Kaloni T P, Lin H, Bedoya-Pinto A, Pandeya A K, Kostanovskiy I, Zhao K, Zhong Y, Hu X, Xue Q-K, Chen X, Ji S-H, Barraza-Lopez S and Parkin S S P 2018 Enhanced spontaneous polarization in ultrathin SnTe films with layered antipolar structure Adv. Mater. **30** 1804428
- [112] Neu J, Nemes C T, Regan K P, Williams M R C and Schmuttenmaur C A 2018 Exploring the solid state phase transition in dl-norvaline with terahertz spectroscopy *Phys. Chem. Chem. Phys.* 20 276
- [113] Erhart P, Hyldgaard P and Lindroth D O 2015 Microscopic origin of thermal conductivity reduction in disordered van der Waals solids *Chem. Mater.* 27 5511
- [114] Sadeghi H, Santarash S and Lambert L C 2015 Enhanced thermoelectric efficiency of porous silicene nanoribbons Sci. Rep. 5 9514
- [115] Javaid S and Akhtar M J 2016 Adsorption and electronic properties of fullerene/Zn-phthalocyanine (C60/ZnPc) interface with face-on orientation: a van der Waals corrected density functional theory investigation *Chem. Phys. Lett.* 649 73
- [116] Yanagisawa S and Hamada I 2017 Determination of geometric and electronic structures of organic crystals from first principles: role of the molecular configuration on the electronic structure J. Appl. Phys. 121 045501
- [117] Chen L, Ma C, Li X, Lin L, Yang S and Li G 2017 First principles design of anthraquinone derivatives in redox flow batteries *Int. J. Electrochem. Sci.* **12** 10433
- [118] Kuusik I, Berholts M, Kruusma J, Kisand V, Tõnisso A, Lust E and Nõrmiste E 2018 Valence electronic structure

of [EMIM][BF₄] ionic liquid: photoemission and DFT-D study *RSC Adv.* **8** 30298

- [119] Azadi S and Cohen R E 2016 Low-pressure phase diagram of crystalline benzene from quantum Monte Carlo J. Chem. Phys. 145 064501
- [120] Kadioglu Y, Santana J A, Özaydin H D, Ersan F, Aktürk O Ü, Aktürk E and Reboredo F A 2018 Diffusion quantum Monte Carlo and density functional calculations of the structural stability of bilayer arsenene J. Chem. Phys. 148 214706
- [121] Lindroth D O and Erhart P 2016 Thermal transport in van der Waals solids from first-principles calculations *Phys. Rev.* B 94 115205
- [122] Clay R CIII, Holzmann M, Ceperley D M and Morales M A 2016 Benchmarking density functionals for hydrogenhelium mixtures with quantum Monte Carlo: energetics, presures, and forces *Phys. Rev. B* **93** 035121
- [123] Fritz M, Fernandez-Serra M and Soler J M 2016 Optimization of an exchange-correlation density functional for water J. *Chem. Phys.* 144 224101
- [124] Claudot J, Kim W J, Dexit A, Kim H, Gould T, Rocca D and Lebegue S 2018 Benchmarking several van der Waals dispersion approaches for the description of intermolecular interactions J. Chem. Phys. 148 064112
- [125] Klinkla R, Sakulsupich V, Pakornchote T, Pinsook U and Bovornratanaraks T 2018 The crucial role of density functional nonlocality and on-axis CH₃NH₃ rotation induced I₂ formation in hybrid organic-inorganic CH₃NH₃PbI₃ cubic perovskite *Nat. Sci. Rep.* 8 13161
- [126] Cechal J, Kley C S, Petuya R, Schramm F, Ruben M, Stepanow S, Arnau A and Kern K 2016 CO₂ binding and induced structural collapse of a surface-supported metal-organic network J. Phys. Chem. C 120 18622
- [127] Zhong J-Q, Wang M, Akter N, Kestell J D, Boscoboinik A M, Kim T, Stacchiola D J, Lu D and Boscoboinik J A 2017 Immobilization of single argon atoms in nano-cages of two-dimensional zeolite model systems *Nat. Commun.* 8 16118
- [128] Braunecker W A, Hurst K E, Ray K G, Owczarczyk Z R, Martinez M B, Leick N, Keuhlen A, Sellinger A and Johnson J C 2018 Phenyl/perfluorophenyl stacking interactions enhance structural order in two-dimensional covalent organic frameworks *Cryst. Growth Des.* 18 4160
- [129] Hellström M, Beinik I, Broqvist P, Lauritsen J V and Hermansson K 2016 Subsurface hydrogen bonds at the polar Zn-terminated ZnO(0001) surface *Phys. Rev.* B 94 245433
- [130] Morisset S, Rougeau N and Teillet-Billy D 2017 Influence of graphene surface on the first steps of the hydrogenation of a coronene molecule *Chem. Phys. Lett.* 679 225
- [131] Borca B, Michnowicz T, Petuya R, Pristl M, Schendel V, Pentegov I, Kraft U, Klauk H, Wahl P, Gutzler R, Arnau A, Schlickum U and Kern K 2017 Electric-field-driven direct desulfurization ACS Nano 11 4703
- [132] Lukin S, Loncaric I, Tireli M, Stolar T, Blanco M V, Lazic P, Uzarevic K and Halasz I 2018 Experimental and theoretical study of selectivity in mechanochemical cocrystallization of nicotinamide with anthranilic and salicylic acid *Cryst. Growth Des.* 18 1539
- [133] Löfgren J, Grönbeck H, Moth-Poulsen K and Erhart P 2016 Understanding the phase diagram of self-assembled monolayers of alkanethiolates on gold J. Phys. Chem. C 120 12059
- [134] Borca B, I Schende V, Petuya R, Pentegov I, Michnowicz T, Kraft U, Klauk H, Arnau A, Wahl P, Schlickum U and Kern K 2015 Bipolar conductance switching of single anthradithiophene molecules ACS Nano 9 12506
- [135] Kuisma E, Hansson C F, Lindberg T B, Gillberg C A, Idh S and Schröder E 2016 Grahene oxide and adsorption of chloroform: a density functional study *J. Chem. Phys.* 144 184704

- [137] Kebede G G, Språngberg D, Mitav P D, Broqvist P and Hermansson K 2017 Comparing van der Waals DFT methods for water on NaCl(001) and MgO(001) J. Chem. Phys. 146 064703
- [138] Shayeghi A, Krähling S, Hörtz P, Johnston R L, Heard C J and Schäfer R 2017 Adsorption of acetonitrile, benzene, and benzonitrile on Pt(111): single crystal adsorption calorimetry and density functional theory J. Phys. Chem. C 121 21354
- [139] Brihuega I and Yndurain F 2018 Selective hydrogen adsorption in graphene rotated bilayers J. Phys. Chem. B 122 595
- [140] Juarez-Mosqueda R, Malola S and Häkkinen H 2017 Stability, electronic structure, and optical properties of protected gold-doped silver $Ag_{29-x}Au_x$ (x = 0-5) nanoclusters *Phys. Chem. Chem. Phys.* **19** 13868
- [141] Buimage-Iarinca L and Morari C 2018 Translation of metalphthalocyanines adsorbed on Au(111): from van der Waals interaction to strong electron correlation *Nat. Sci. Rep.* 8 12728
- [142] Tabatabaei S-M, Pourfath M and Fathipour M 2018 Adsorption characteristics of epigenetically modified DNA nucleobases on single-layer MoS₂: a first-principles study J. Appl. Phys. **124** 134501
- [143] Hermann G, Steinkasserer L E M, Paulus B and Tremblay J C 2018 Dipole-induced transition orbitals: a novel tool for investigating optical transitions in extended systems J. *Phys. Chem. Lett.* 9 6624
- [144] Barcaro G, Sementa L, Monti S, Carravetta V, Broqvist P, Kullgren J and Hermansson K 2018 Dynamical and structural characterization of the adsorption of fluorinated alkane chains onto CeO₂ J. Phys. Chem. C 122 23405
- [145] Loncaric I, Rukelj Z, Silkin V M and Despoja V 2018 Strong two-dimensional plasmon in Li-intercalated hexagonal boron-nitride film with low damping Npj. 2D Mater. Appl. 2 33
- [146] Wang M, Zhong J-Q, Kestell J, Waluyo I, Stacchiola D J, Boscoboinik J A and Lu D 2017 Energy level shifts at the silica/Ru(0001) heterojunction driven by surface and interface dipoles *Top. Catal.* 60 481
- [147] Akter N, Wang W, Zhong J-Q, Liu Z, Kim T, Lu D, Boscoboinik J A and Stacchiola D J 2018 Stabilization of oxidized copper nanoclusters in confined spaces *Top. Catal.* 61 419
- [148] Shi W, Wu S and Wang Z 2018 Triggering basal plane active sites of monolayer MoS₂ for the hydrogen evolution reaction by phosphorus doping *J. Nanopart. Res.* 20 271
- [149] Koch M, Pagan M, Persson M, Gawinkowski S, Waluk J and Kumagai T 2017 Direct observation of double hydrogen transfer via quantum tunneling in a single porphycene molecule on a Ag(110) surface J. Am. Chem. Soc. 139 12681
- [150] Petuya R and Arnau A 2017 Magnetic coupling between 3d transition metal adatoms on graphene supported by metallic substrates *Carbon* 116 599
- [151] Capitan M J, Alvarez J and Navio C 2018 Study of the electronic structure of electron accepting cyano-films: TCNQ versus TCNE *Phys. Chem. Chem. Phys.* **20** 10450
- [152] Mehta A N, Mu W, Murugesan M, Jiao Y, Fu Y, Hyldgaard P and Liu J 2018 Understanding noninvasive charge transfer doping of graphene:a comparative study J. Mater. Sci., Mater. Electron. 29 5239
- [153] Buimage-Iarinca L and Morari C 2018 Charge transport pathways in metal porphyrin as interplay between long and short range scattering processes *Nanotechnology* 30 045204

- [154] London F 1930 Zur Theorie und Systematik der Molekularkräfte Z. Phys. 63 245
- [155] Eisenshitz R and London F 1930 Über das Verhältnis der van der Waalsschen Kräfte zu den homöopolaren Bindungskräften Z. Phys. 60 491
- [156] London F 1937 The general theory of molecular forces *Trans. Faraday Soc.* **33** 8
- [157] Dzyaloshinskii I E, Lifshitz E M and Pitaevskii L P 1961 The general theory of van der Waals forces Adv. Phys. 10 165
- [158] Mahan G D 1965 van der Waals forces in solids J. Chem. Phys. 43 1569
- [159] Bohm D and Pines D 1951 A collective description of electron interactions. I. Magnetic interactions *Phys. Rev.* 82 625
- [160] Noziéres P and de Dominicis C T 1969 Singulatities in the x-ray adsorption and emission of metals. III. One-body theory exact solution *Phys. Rev.* **178** 1097
- [161] Mahan G D 1967 Excitons in degenerate semiconductors *Phys. Rev.* 153 882
- [162] Mahan G D 1967 Excitons in metals: infinite hole mass Phys. Rev. 163 612
- [163] Hedin L 1980 Effects of recoil on shake-up spectra in metals *Phys. Scr.* 21 477
- [164] Gunnarsson O, Meden V and Schönhammer K 1994 Corrections to Migdal's theorem for spectral functions: a cumulant treatment of the time-dependent green's function *Phys. Rev.* B 50 10462
- [165] Kleis J, Hyldgaard P and Schröder E 2005 van der Waals interaction of parallel polymers and nanotubes *Comput. Mater*. *Sci.* 33 192
- [166] Dobson J F and Wang J 1999 Successful test of a seamless van der Waals density functional Phys. Rev. Lett. 82 2123
- [167] Mahan G D 1981 Many-Particle Physics (New York, NY: Plenum)
- [168] Ma S and Brueckner K A 1968 Correlation energy of an electron gas with a slowly varying high density *Phys. Rev.* 165 18
- [169] Fetter A L and Walecka J D 1971 Quantum Theory of Many Particle Systems (New York, NY: McGraw-Hill)
- [170] Burke K and Wagner L O 2013 DFT in a nutshell Int. J. Quantum Chem. 113 96
- [171] Dion M and van der 2004 Waals forces in density functional theory *PhD Thesis* Rutgers University (Piscataway, NC, USA)
- [172] von Barth U and Hedin L 1972 A local exchange-correlation potential for the spin polarized case: I J. Phys. C: Solid State Phys. 5 1629
- [173] Gunnarsson O, Jonson M and Lundqvist B I 1979 Descriptions of exchange and correlation effects in inhomogeneous electron systems *Phys. Rev.* B 20 3136
- [174] Langreth D C and Vosko S H 1990 Response functions and non-local functionals Adv. Quantum Chem. 21 175
- [175] Andersson S, Wilzén L and Persson M 1988 Physisorption interaction of H₂ with noble-metal surfaces: a new H₂-Cu potential *Phys. Rev.* B 38 2967
- [176] Andersson S and Persson M 1993 Crystal-face dependence of physisorption potentials *Phys. Rev.* B 48 5685
- [177] Hult E, Andersson Y, Lundqvist B I and Langreth D C 1996 Density functional for van der Waals forces at surfaces *Phys. Rev. Lett.* **77** 2029
- [178] Hult E and Kiejna A 1997 Trends in atom/molecule-surface van der Waals interactions *Surf. Sci.* **383** 88
- [179] Andersson Y, Hult E, Apell P, Langreth D C and Lundqvist B I 1998 Density-functional account of van der Waals forces between parallel surfaces *Solid State Commun.* 106 235
- [180] Hult E, Rydberg H, Lundqvist B I and Langreth D C 1999 Unified treatment of asymptotic van der Waals forces *Phys. Rev.* B 59 4708

- [181] Lee K, Kelkkanen A K, Berland K, Andersson S, Langreth D C, Schröder E, Lundqvist B I and Hyldgaard P 2011 Evaluation of a density functional with account of van der Waals forces using experimental data of H₂ physisorption on Cu(111) *Phys. Rev.* B 84 193408
- [182] Lee K, Berland K, Yoon M, Andersson S, Schröder E, Hyldgaard P and Lundqvist B I 2012 Benchmarking van der Waals density functionals with experimental data: potential-energy curves for H₂ molecules on Cu(111), (100), and (110) surfaces J. Phys.: Condens. Matter. 24 424213
- [183] Baym G and Kadanoff L P 1961 Conservation laws and correlation functions *Phys. Rev.* 124 287
- [184] Zaremba E and Kohn W 1976 Van der Waals interaction between an atom and a solid surface *Phys. Rev.* B **13** 2270
- [185] Zaremba E and Kohn W 1977 Theory of helium adsorption on simple and noble-metal surfaces *Phys. Rev.* B **15** 1769
- [186] Nordlander P and Harris J 1984 The interaction of helium with smooth metal surfaces J. Phys. C: Solid State Phys. 17 1141
- [187] Hult E, Hyldgaard P, Rossmeisl J and Lundqvist B I 2001 Density-functional calculation of van der Waals forces for free-electron-like surfaces *Phys. Rev.* B 64 195414
- [188] Hedin L 1965 New method for calculating the one-particle Green's function with application to the electron-gas problem *Phys. Rev.* 139 A796
- [189] Rasolt M and Geldart D J W 1975 Gradient corrections in the exchange and correlation energy of an inhomogeneous electron gas *Phys. Rev. Lett.* **35** 1234
- [190] Hubbard J and Peierls R E 1957 The description of collective motions in terms of many-body perturbation theory *Proc. R. Soc.* A 240 539
- [191] Yuk S F, Pitike K C, Nakhmanson S, Eisenbach M, Li Y W and Cooper V R 2017 Towards an accurate description of perovskite ferroelectrics: exchange and correlation effects *Nat. Sci. Rep.* 7 43482
- [192] Hubbard J and Peierls R E 1958 The description of collective motions in terms of many-body perturbation theory. II. The correlation energy of a free-electron gas *Proc. R. Soc.* A 243 336
- [193] Andersson Y 1996 van der Waals density functionals PhD Thesis Department of Applied Physics, Chalmers University of Technology (Göteborg, Sweden)
- [194] Jastrow R 1955 Many-body problem with strong forces *Phys. Rev.* 98 1479
- [195] Ceperley D, Chester G V and Kalos M H 1977 Monte Carlo simulation of a many-fermion study *Phys. Rev.* B 16 3081
- [196] Ceperley D M and Alder B J 1980 Ground state of the electron gas by a stochastic method *Phys. Rev. Lett.* 45 566
- [197] Burke K, Ernzerhof M and Perdew J P 1997 The adiabatic connection method: a non-empirical hybrid *Chem. Phys. Lett.* 265 115
- [198] Hyldgaard P 2008 Density-functional theory of nonequilibrium tunneling *Phys. Rev.* B 78 165109
- [199] Hyldgaard P 2012 Nonequilibrium thermodynamics of interacting tunneling transport: variational grand potential, universal density functional description, and nature of forces *J. Phys.: Condens. Matter.* 24 424219
- [200] Runge E and Gross E K U 1984 Density-functional theory for time-dependent systems *Phys. Rev. Lett.* **52** 997
- [201] Vignale G and Rasolt M 1987 Density-functional theory in strong magnetic fields *Phys. Rev. Lett.* 59 2360
- [202] Dobson J F, Gould T and Vignale G 2014 How many-body effects modify the van der Waals interaction between graphene sheets *Phys. Rev.* X 4 021040
- [203] Lundqvist B I, Andersson Y, Shao H, Chan S and Langreth D C 1995 Density functional theory including van der Waals forces Int. J. Quantum Chem. 56 247

- [204] Dobson J F and Dinte B P 1996 Constraint satisfaction in local and gradient susceptibility approximations: application to a van der Waals density functional *Phys. Rev. Lett.* **76** 1780
- [205] Andersson Y, Hult E, Rydberg H, Apell P, Lundqvist B I and Langreth D C 1997 van der Waals interactions in density functional theory *Electronic Density Functional Theory: Recent Progress and New Directions* ed J F Dobson, G Vignale and M P Das (Berlin: Plenum)
- [206] Rydberg H, Jacobson N, Hyldgaard P, Simak S I, Lundqvist B I and Langreth D C 2003 Hard numbers on soft matter Surf. Sci. 532–535 606
- [207] Schröder E and Hyldgaard P 2003 van der Waals interactions of parallel and concentric nanotubes *Mater. Sci. Eng.* C 23 721
- [208] Schröder E and Hyldgaard P 2003 The van der Waals interactions of concentric nanotubes Surf. Sci. 532–535 880
- [209] Kleis J and Schröder E 2005 van der Waals interaction of simple, parallel polymers J. Chem. Phys. 122 164902
- [210] Berland K and Hyldgaard P 2010 Structure and binding in crystals of cagelike molecules: hexamine and platonic hydrocarbons J. Chem. Phys. 132 134705
- [211] Berland K, Borck Ø and Hyldgaard P 2011 van der Waals density functional calculations of binding in molecular crystals *Comput. Phys. Commun.* 182 1800
- [212] Wellendorff J, Lundgaard K T, Møgelhøj A, Petzold V, Landis D D, Nørskov J K, Bligaard T and Jacobsen K W 2012 Density functionals for surface science: exchangecorrelation model development with Bayesian error estimation *Phys. Rev.* B **85** 235149
- [213] Natkaniec I, Bokhenkov E L, Dorner B, Kalus J, Mackenzie G A, Pawley G S, Schmelzer U and Sheka E F 1980 Phonon dispersion in d₈-naphthalene crystal at 6k J. Phys. C: Solid State Phys. 13 4265
- [214] Armiento R and Mattsson A E 2005 Functional designed to include surface effects in self-consistent density functional theory *Phys. Rev.* B 72 085108
- [215] Fultz B 2010 Vibrational thermodynamics of materials Prog. Mater. Sci. 55 247
- [216] Togo A and Tanaka I 2015 First principles phonon calculations in materials science *Scr. Mater.* **108** 1
- [217] Kresse G and Hafner J 1993 Ab initio molecular dynamics for liquid metals *Phys. Rev.* B 47 558(R)
- [218] Kresse G and Furthmüller J 1996 Efficiency of ab-initio total energy calculations for metals and semiconductors using a plane-wave basis set *Comput. Mater. Sci.* 6 15
- [219] Giannozzi P et al 2009 Quantum espresso: a modular and open-source software project for quantum simulations of materials J. Phys.: Condens. Matter. 21 395502
- [220] Medvedev M G, Bushmarinov I S, Sun J, Perdew J P and Lyssenkov K A 2017 Density functional theory is straying from the path toward the exact functional *Science* 355 49
- [221] Hohenberg P and Kohn W 1964 Inhomogeneous electron gas Phys. Rev. 136 B864
- [222] Lee K, Morikawa Y and Langreth D C 2010 Adsorption of n-butane on Cu(100), Cu(111), Au(111), and Pt(111): van der Waals density-functional study *Phys. Rev. B* 82 155461
- [223] Patra A, Bates J E, Sun J and Perdew J P 2017 Properties of real metallic surfaces: effects of density functional semilocality and van der Waals nonlocality *Proc. Natl Acad. Sci. USA* 144 E9188
- [224] Terentjev A V, Constantin L A, Artacho E and Pitarke J M 2019 Comparison of dispersion-corrected exchangecorrelation functionals using atomic orbitals *Phys. Rev.* B 100 235439
- [225] Löfgren J 2019 Modeling colloidal nanoparticles: from growth to deposition *PhD Thesis* Department of Physics, Chalmers University of Technology (Göteborg, Sweden)

- [226] Löfgren J, Rahm J, Brorsson J M and Erhart P 2020 in reference [225] with tentative title 'Quantitative assessment of the efficacy of halides as shape-directing agents in nanoparticle growth.' Submitted
- [227] Ziman J M 1972 Principles of the Theory of Solids (Cambridge: Cambridge University Press)
- [228] Tran F and Hutter J 2013 Erratum: nonlocal van der Waals functionals: the case of rare-gas dimers and solids [J. Chem. Phys. 138, 204103 (2013)] J. Chem. Phys. 139 039903
- [229] Tran F, Stelzl J and Blaha P 2016 Rungs 1 to 4 of DFT Jacob's ladder: extensive test on the lattice constant, bulk modulus, and cohesive energy of solids J. Chem. Phys. 144 204120
- [230] Tran F, Kalantari L, Traore B, Rocquefelte X and Blaha P 2019 Nonlocal van der waals functionals for solids: choosing an appropriate one *Phys. Rev. Mater.* 3 063602
- [231] Tran F, Stelzl J, Koller D, Ruh T and Blaha P 2017 Simple way to apply nonlocal van der Waals functionals within allelectron methods *Phys. Rev. B* 96 054103
- [232] Andermatt S, Cha J, Schiffmann F and VandeVondele J 2016 Combining linear-scaling DFT with subsystem DFT in born-oppenheimer and ehrenfest molecular dynamics simulations: from molecules to a virus in solution J. Chem. Theory Comput. 12 3214
- [233] CP2K version 2.4 CP2K is freely available at http://cp2k. org (accessed January 2020).
- [234] Gamo Y, Nagashima A, Wakabayashi M, Terai M and Oshima C 1997 Atomic structure of monolayer graphite formed on Ni(111) Surf. Sci. 374 61
- [235] Olsen T, Yan Y, Mortensen J J and Thygesen K 2011 Dispersive and covalent interactions between graphene and metal surfaces from the random phase approximation *Phys. Rev. Lett.* **107** 156401
- [236] Olsen T and Thygesen K 2013 Random phase approximation applied to solids, molecules, and graphene-metal interfaces: from van der Waals to covalent bonding *Phys. Rev. B.* 87 075111
- [237] Pople J A, Head-Gordon M, Fox D J, Raghavachari K and Curtiss L A 1989 Gaussian-1 theory: a general procedure for prediction of molecular energies J. Chem. Phys. 90 5622
- [238] Curtiss L A, Jones C, Trucks G W, Raghavachari K and Pople J A 1990 Gaussian-1 theory of molecular energies for second-row compounds J. Chem. Phys. 93 2537
- [239] Curtiss L A, Raghavachari K, Trucks G W and Pople J A 1991 Gaussian-2 theory for molecular energies of first- and second-row compounds J. Chem. Phys. 94 7221
- [240] Curtiss L A, Raghavachari K, Redfern P C and Pople J A 1997 Assessment of Gaussian-2 and density functional theories for the computation of enthalpies of formation J. Chem. Phys. 106 1063
- [241] Goerigk L and Grimme S 2010 A general database for main group thermochemistry, kinetics, and noncovalent interactions – assessment of common and reparameterized (meta) GGA density functionals J. Chem. Theory Comput. 6 107
- [242] Karton A, Daon S, Martin J M L and Ruscic B 2011 W4-11: a high-confidence benchmark dataset for computational thermochemistry derived from first-principles W4 data Chem. Phys. Lett. 510 165
- [243] Ivanov O and Antropov V P 1999 Molecular magnetism: noncollinear ordering and spin dynamics J. Appl. Phys. 85 4821
- [244] Obata M, Nakamura M, Hamada I and Oda T 2013 Implementation of van der Waals density functional approach to the spin-polarized system: interaction potential between oxygen molecules J. Phys. Soc. Japan 82 093701
- [245] Obata M, Nakamura M, Hamada I and Oda T 2015 Improving the description of nonmagnetic and magnetic molecular crystals via the van der Waals density functional J. Phys. Soc. Japan 84 024715

- [246] Ambrosetti A, Ferri N, DiStatio R A Jr and Tkatchenko A 2016 Wavelike charge density fluctuations and van der Waals interactions at the nanoscale *Science* 351 1171
- [247] Barash Y S and Ginzburg V L 1984 Some problems in the theory of van der Waals forces Usp. Fiz. Nauk 143 345
 - Barash Y S and Ginzburg V L 1984 Some problems in the theory of van der Waals forces *Sov. Phys. Usp.* **27** 467 (Engl. transl.)
- [248] Barash Y S 1988 van der Waals interactions between thin conducting layers *Fiz. Tverd. Tela* 30 2738
 - Barash Y S 1988 van der Waals interactions between thin conducting layers *Sov. Phys. Solid State* **30** 1578 (Engl. transl.)
- [249] Sernelius B E and Björk P 1998 Interaction energy for a pair of quantum wells *Phys. Rev.* B 57 6592
- [250] Dobson J F, White A and Rubio A 2006 Asymptotics of the dispersion interaction: analytic benchmarks for van der Waals energy functionals *Phys. Rev. Lett.* **96** 073201
- [251] Lebègue S, Harl J, Gould T, Ángyán J G, Kresse G and Dobson J F 2010 Cohesive properties and asymptotics of the dispersion interaction in graphite by the random phase approximation *Phys. Rev. Lett.* **105** 196401
- [252] Ruzsinszky A, Perdew J P, Tao J, Csonka G I and Pitarke J M 2012 Van der Waals coefficients for nanostructures: fullerenes Defy conventional wisdom *Phys. Rev. Lett.* 109 233203
- [253] Dobson J F and Gould T 2012 Calculation of dispersion energies J. Phys.: Condens. Matter. 24 073201
- [254] Gobre V V and Tkatchenko A 2013 Scaling laws for the Waals interactions in nanostructured materials *Nat. Commun.* 4 2341
- [255] Harris J 1985 Simplified method for calculating the energy of weakly interacting fragments *Phys. Rev.* B **31** 1770
- [256] Berland K, Londero E, Schröder E and Hyldgaard P 2013 Harris-type van der Waals density functional scheme *Phys. Rev.* B 88 045431
- [257] Berland K, Einstein T L and Hyldgaard P 2009 Rings sliding on a honeycomb network: adsorption contours, interactions, and assembly of benzene on Cu(111) *Phys. Rev. B* 80 155431
- [258] Sun J, Ruzsinszky A and Perdew J P 2015 Strongly constrained and appropriately normed semilocal density functional *Phys. Rev. Lett.* **115** 036402
- [259] 2016 TURBOMOLE Version 7.1.1 TURBOMOLE is developed by University Karlsruhe and Forschungcenter Karlsruhe GmbH 1989–2007 and is available at http://turbomole.com (accessed January 2020)
- [260] Ahlrichs R, Bär M, Häser M, Horn H and Kölmel C 1989 Electronic structure calculations on workstation computers: the program system turbomole *Chem. Phys. Lett.* 162 165
- [261] Adamo C and Barone V 1999 Towards reliable density functional methods without adjustable parameters: The PBE0 model J. Chem. Phys. 110 6158
- [262] Sedlak R, Janowski T, Pitonak M, Rezac J, Pulay P and Hobza P 2013 Accuracy of quantum chemical methods for large noncovalent complexes J. Chem. Theory Comput. 9 3364
- [263] Rezac J and Hobza P 2013 Describing noncovalent interactions beyond the common approximations: how accurate is the 'gold standard,' CCSD(T) at the complete basis set limit? *J. Chem. Theory Comput.* **9** 2151
- [264] Taylor D E et al 2016 Blind test of density-functional-based methods on intermolecular interaction energies J. Chem. Phys. 145 124105
- [265] Kruse H, Mladek A, Gkionis K, Hansen A, Grimme S and Sponer J 2015 Quantum chemical benchmark study on 46 RNA backbone families using a dinucleotide unit *J. Chem. Theory Comput.* **11** 4972

- [266] Reilly A M and Tkatchenko A 2013 Understanding the role of vibrations, exact exchange, and many-body van der Waals interactions in the cohesive properties of molecular crystals *J. Chem. Phys.* **139** 024705
- [267] Access to published, trusted reference geometries exists at the GMTKN55 webpage https://www.chemie.uni-bonn. de/pctc/mulliken-center/software/GMTKN/gmtkn55/. The webpage also provides easy access to the GMTKN55 CCSD(T) results for those geometries.
- [268] Garrity K F, Bennett J W, Rabe K M and Vanderbilt D 2014 Pseudopotentials for high-throughput DFT calculations *Comput. Mater. Sci.* 81 446
- [269] Bengtsson L 1999 Dipole correction for surface supercell calculations Phys. Rev. B 59 12301
- [270] Soler J M, Artacho E, Gale J D, Garcia A, Junquera J, Ordejon P and Sanchez-Portal D 2002 The SIESTA method for ab initio order-N materials simulation J. Phys.: Condens. Matter. 14 2745
- [271] Vydrov O A and Van Voorhis T 2010 Dispersion interactions from a local polarizability model *Phys. Rev.* A 81 062708
- [272] Lejaeghere K, Van Speybroeck V, Van Oost G and Cottenier S 2014 Error estimates for solid-state density-functional theory predictions: an overview by means of the ground-state elemental crystals Crit. Rev. Solid State Mater. Sci. 39 1
- [273] Tyson W 1975 Surface energies of solid metals Can. Metall. O. 14 307
- [274] Tyson W and Miller W 1977 Surface free energies of solid metals: estimation from liquid surface tension measurements Surf. Sci. 62 267
- [275] Jura G and Garland C W 1952 The experimental determination of the surface tension of magnesium oxide J. Am. Chem. Soc. 74 6033
- [276] Derry G N, Kern M E and Worth E H 2015 Recommended values of clean metal surface work functions J. Vac. Sci. Technol. A 33 060801
- [277] Michaelson H B 1977 The work function of the elements and its periodicity J. Appl. Phys. 48 4729
- [278] Londero E, Karlson E K, Landahl M, Ostrovskii D, Rydberg J D and Schröder E 2012 Desorption of n-alkanes from graphene: a van der Waals density functional study *J. Phys.: Condens. Matter.* 24 424212
- [279] Loncaric I and Despoja V 2014 Benchmarking van der Waals functionals with noncontact RPA calculations on graphene-Ag(111) Phys. Rev. B 90 075414
- [280] Kiraly B, Iski E V, Mannix A J, Fisher B L, Hersam M C and Guisinger N P 2013 Solid-source growth and atomic-scale characterization of graphene on ag(111) *Nat. Commun.* 4 2804
- [281] Lee K, Kolb B, Thonhauser T, Vanderbilt D and Langreth D C 2012 Structure and energetics of a ferroelectric organic crystal of phenazine and chloranilic acid *Phys. Rev.* B 86 104102
- [282] Perdew J P, Ruzsinszky A, Sun J, Glindmeyer S and Csonka G I 2012 van der Waals interaction as a summable asymptotic series *Phys. Rev.* A 86 062714
- [283] Lacks D J and Gordon R G 1993 Pair interactions of rare-gas atoms as a test of exchange-energy-density functionals in regions of large density gradients *Phys. Rev.* A 47 4681
- [284] Murray É D, Lee K and Langreth D C 2009 Investigation of exchange energy density functional accuracy for interacting molecules J. Chem. Theory Comput. 5 2754
- [285] Kannemann F O and Becke A D 2009 van der Waals interactions in density-functional theory: rare-gas diatomics J. Chem. Theory Comput. 5 719
- [286] Ellis J P 2017 TikZ-Feynman: Feynman diagrams with TikZ J. Comput. Phys. Commun. 210 103
- [287] Momma K and Izumi F 2011 Vesta 3 for three-dimensional visualization of crystal, volumetric and morphology data J. Appl. Crystall 44 1272